The Role of Automatic Processing of Food Cues in Unhealthy Eating Behaviour

Naomi Kakoschke

B. BSc. (Psych Hons)

A thesis submitted to Flinders University, South Australia, in fulfilment of the requirement for the degree of Doctor of Philosophy

School of Psychology

Faculty of Social and Behavioural Sciences

December 2016

Summaryiv
Declarationvi
Acknowledgementsvii
CHAPTER 1: GENERAL INTRODUCTION1
Chapter Overview1
Cognitive Underpinnings of Unhealthy Eating Behaviour1
Dual-Process Models of Behaviour2
Automatic Processing, Self-Regulatory Control and Trait Eating Style7
Modifying Automatic Processing and Self-Regulatory Control8
Aims of the Thesis9
Outline of the Thesis10
References12
CHAPTER 2: STUDY 1
Combined effects of cognitive bias for food cues and poor inhibitory control on unhealthy
food intake
Introduction
Method26
Results
Discussion
References
CHAPTER 3: STUDY 246
Implicit food evaluation and emotional eating interact to predict unhealthy food choice
Introduction
Method

Results	55
Discussion	57
References	61
CHAPTER 4: STUDY 3	68
Differential effects of approach bias and eating style on unhealthy food consumption	on in
overweight and normal weight women	
Introduction	70
Method	73
Results	76
Discussion	
References	87
CHAPTER 5: LITERATURE REVIEW	93
Approach bias modification training and consumption: A review of the literature	
Introduction	95
Method	98
Results	100
Discussion	107
References	112
CHAPTER 6: STUDY 4	118
The effect of combined avoidance and control training on implicit food evaluation	and choice
Introduction	120
Method	124
Results	130
Discussion	133
References	137

CHAPTER 7: STUDY 5	147
Impulsivity moderates the effect of approach bias modification on healthy food co	nsumption
Introduction	149
Method	154
Results	159
Discussion	163
References	167
CHAPTER 8: GENERAL DISCUSSION	172
Chapter Overview	172
Summary of Findings	172
Theoretical Implications	174
Practical Implications	177
Methodological Issues and Future Directions	181
Conclusion	184
References	

Summary

Recent empirical research and theoretical models have acknowledged that automatic processing of appetitive food cues is an important contributor to unhealthy eating behaviour. However, relatively little research has examined when, and for whom, automatic processing of food cues influences such behaviour. Therefore, the first aim of the thesis was to investigate automatic processing in conjunction with self-regulatory control or trait eating styles to gain a better understanding of unhealthy eating behaviour. The second aim was to determine whether interventions designed to modify automatic processes and/or selfregulatory control are effective for reducing unhealthy eating and/or promoting healthier eating. These two aims were addressed in a series of correlational and experimental studies.

The thesis consists of six papers (two published, one under revision, and three under review). Study 1 examined the combined effects of cognitive bias (attentional and approach) and inhibitory control on unhealthy eating behaviour. It was found that a stronger approach bias for unhealthy food combined with lower inhibitory control predicted increased consumption of unhealthy food. Study 2 tested the effects of the affective aspect of automatic processing (implicit evaluation) and an emotional eating style on unhealthy eating behaviour. The findings showed that a positive implicit food evaluation (increased liking) predicted increased increased choice for unhealthy snack food in participants with lower emotional eating.

Study 3 investigated the combined effects of approach bias for food and eating style on unhealthy food consumption in normal weight and overweight individuals using a pooled sample. Among overweight participants, an external and emotional eating style individually moderated the relationship between approach bias for unhealthy food and snack intake, such that approach bias was positively related to consumption in high external or emotional eaters, but negatively related in low external or emotional eaters. These interactions were not observed among normal weight participants. Studies 4 and 5 experimentally manipulated automatic processing and/or inhibitory control. These studies were preceded by a literature review on the effectiveness of approach bias modification as an intervention for reducing the consumption of appetitive substances in general. All of the reviewed studies (with one exception) that reported a positive outcome for consumption also showed a successful reduction of approach bias for appetitive cues. Study 4 demonstrated that the combined re-training of approach bias for unhealthy food together with inhibitory control was more effective than either task alone for reducing implicit liking of unhealthy food; however, no significant effects were found for food consumption. Approach bias re-training on its own did reduce unhealthy snack food choice. Study 5 found that the effect of approach bias re-training on subsequent consumption was moderated by trait impulsivity, such that only highly impulsive participants ate a greater proportion of healthy food following training.

Overall, the findings contribute to emerging evidence for the role of automatic processing of appetitive food cues together with self-regulatory control and trait eating style in unhealthy eating behaviour. The results also contribute to a theoretical understanding of unhealthy eating based on recent dual-process models of behaviour (Strack & Deutsch, 2004), and inform the development of interventions designed to reduce such behaviour.

Declaration

'I certify that this thesis does not incorporate without acknowledgement any material previously submitted for a degree or diploma in any university; and that to the best of my knowledge and belief it does not contain any material previously published or written by another person except where due reference is made in the text.'

.....

Naomi Kakoschke, B. BSc. (Psych Hons)

Acknowledgements

First, I would like to thank my primary supervisor, Professor Eva Kemps, for all of her help, support, and guidance throughout my PhD. I would also like to thank Eva for her invaluable feedback, enthusiasm, and commitment to assisting me to complete my thesis. I would especially like to thank her for motivating me to undertake this PhD. I am very glad to have had such an incredible supervisor and to have gained an inspirational role model from this experience. Thank you to my co-supervisor, Professor Marika Tiggemann, for the words of wisdom, valuable input, and expertise over the last few years. I am grateful to Sarah Hollitt for her practical assistance and to Paul Douglas for developing the software for my research. Thank you to the School of Psychology staff and to everyone who participated in my studies.

To all of my friends and PhD colleagues, thank you for giving me a constant source of encouragement, and for the countless coffee and lunch breaks that we have enjoyed together. A special mention to Sue, who I shared an office with, for the moral support and importantly, for making me endless cups of tea. To my family, thank you for always believing in me. I am especially grateful to my parents and brother. I would not be where I am today if it were not for their unconditional love and support. Last but not least, I would like to thank Ben for his unwavering patience, understanding, and optimism, which kept me going during this journey. I am sincerely thankful to everyone that has been acknowledged as it would not have been possible to complete my PhD without their continued support, advice, and encouragement.

CHAPTER 1: GENERAL INTRODUCTION

Chapter Overview

This introductory chapter aims to provide a brief overview of the research relevant to the cognitive underpinnings of unhealthy eating behaviour from the theoretical perspective of contemporary dual-process models (Strack & Deutsch, 2004). Specifically, the empirical research on automatic and self-regulatory control processes in the context of unhealthy eating is reviewed, followed by research on eating styles that is related to these two processes. This chapter concludes with a summary of the main aims of the thesis and an outline of its content.

Cognitive Underpinnings of Unhealthy Eating Behaviour

Rates of overweight and obesity in modern Western societies have more than doubled during the last few decades (WHO, 2014). It is estimated that worldwide 39% of adults can now be classified as overweight and 13% as obese (WHO, 2014). Being overweight or obese is considered a major health problem as individuals in the unhealthy weight range (body mass index [BMI] >25 kg/m²) are at an increased risk of developing several chronic diseases, such as cardiovascular disease, diabetes, and some types of cancer (WHO, 2014). Excess weight is generally due to an energy imbalance that is largely driven by over-consumption of palatable, energy-dense foods that are high in fat, salt, and/or sugar (Hill, Wyatt, Reed, & Peters, 2003).

One acknowledged contributor to unhealthy eating behaviour is the contemporary 'obesogenic environment' in which we live, where an abundance of high-calorie foods are readily accessible (Polivy, Herman & Coelho, 2008). In this environment, individuals are continually exposed to cues associated with appetitive food through advertisements in various media outlets, such as on billboards, television, the internet, and in magazines (Brunner, van der Horst, Siegrist, 2010). An often overlooked contributor to unhealthy eating behaviour is the automatic processing of appetitive food and associated cues in the environment, which occurs with little awareness (Marteau, Hollands, & Fletcher, 2012). However, not everyone is affected to the same extent by the food-rich environment as some individuals do maintain a healthy weight. Therefore, it is important to develop an understanding of how automatic processing of food cues may interact with other constructs, such as self-regulatory control of responses to such cues and eating styles, in predicting unhealthy eating behaviour. Examining the mechanisms underlying eating behaviour is likely to inform the development of effective interventions to reduce consumption of high-calorie food that can lead to excess weight gain.

Section 1.2 begins with a description of recent dual-process models of behaviour to provide a theoretical account of how food cues in the environment may influence unhealthy eating behaviour. Next, research on automatic processing and self-regulatory control in eating behaviour is outlined. Finally, the potential role of individual differences in trait eating style in the interplay between these two processes is discussed.

Dual-Process Models of Behaviour

The premise that automatic and controlled processing both contribute to behaviour is based on contemporary dual-process models (Strack & Deutsch, 2004). According to these models, behaviour is determined by two different types of information processing: automatic and controlled processing. Automatic processing, which is fast, implicit, and effortless, can involve affective (e.g., attitudes, preferences) and cognitive (e.g., attending to, approaching) responses to rewarding cues in the environment, including unhealthy food. Components of automatic processing can often be biased. The two main types of biased automatic processing that have been identified are attentional bias, an automatic allocation of attention toward rewarding cues (MacLeod & Mathews, 2012), and approach bias, an automatic (action) tendency to reach out toward (approach) rather than move away from (or avoid) such cues (Wiers, Gladwin, Hofmann, Salemink, & Ridderinkoff, 2013). Behaviour can also be guided by an automatic appraisal of the affective properties of a stimulus, which can in turn drive approach toward that specific stimulus if it becomes associated with positive affect. In contrast to automatic processing, controlled processing is slow, explicit, and more effortful, and is thought to be driven by conscious deliberation based on long-term goals and personal standards (e.g., to lose weight and maintain health). Dual-process models are also concerned with how these two types of processes work together to determine behaviour. Specifically, it is suggested that these automatic and controlled processes elicit conflicting signals when confronted with appetitive stimuli in the environment (Wiers et al., 2013). The outcome of this conflict is determined by the relative strength of each type of process. For example, consumption of unhealthy food may be determined by a strong attentional and/or approach bias or a positive evaluation of such food combined with poor self-regulatory control of responses (Hofmann, Friese, & Wiers, 2008; Strack & Deutsch, 2004). Dual-process models posit that self-regulatory control enables individuals to resist the cognitive-motivational drive of automatic processing to consume unhealthy food, so that eating behaviour can be regulated in line with long-term goals (e.g., to lose weight). Both state (e.g., inhibitory control) and trait (e.g., impulsivity) self-regulatory control can regulate the influence of automatic processing on behaviour.

Research on Automatic Processing and Eating Behaviour

Support for the role of automatic processing in eating behaviour has primarily come from research examining attentional and approach biases for food cues. Of these, attentional bias has by far been the most researched. Indeed, a number of studies have shown that both normal weight (e.g., Hollitt, Kemps, Tiggemann, Smeets, & Mills, 2010; Brignell, Griffiths, Bradley, & Mogg, 2009; Nijs, Franken, & Muris, 2009; Hou et al., 2011) and overweight or obese individuals (Castellanos et al., 2009; Mogg, Bradley, Hyare, & Lee, 1998; Njis, Muris, Euser, & Franken, 2010) are faster to respond to food cues relative to neutral (non-food) cues.

Studies have also demonstrated a positive relationship between attentional bias for unhealthy food and subsequent snack food consumption during laboratory taste tests in both normal weight and obese individuals (Nijs, et al., 2010; Werthmann et al., 2011). In addition, Calitri, Pothos, Tapper, Brunstrom, and Rogers (2010) found that an attentional bias for food predicted increased body weight over one year, although Pothos, Tapper, and Calitri (2009) found no such relationship. Nevertheless, some emerging evidence supports the proposition that biased automatic attentional processing of food is related to unhealthy eating behaviour.

More recently, researchers have begun to investigate approach bias for food cues. In particular, studies have shown that normal weight (Brignell et al., 2009; Kemps, Tiggemann, Martin, & Elliott, 2013; Veenstra & de Jong, 2010) and overweight or obese individuals (Havermans, Giesen, Houben, & Jansen, 2011; Kemps & Tiggemann, 2015; Mogg et al., 2012) are faster to approach than avoid food cues. Approach bias is likely to be a more important contributor to unhealthy eating behaviour than attentional bias due to its additional behavioural component. Indeed, two studies have shown that approach bias for unhealthy food is positively related to consumption of such food in normal weight samples (Hofmann, Gschwendner, Wiers, Friese, & Schmitt, 2008; Nederkoorn, Houben, Hofmann, Roefs, & Jansen, 2010). Thus, there is emerging evidence to support the role of approach bias for food in unhealthy eating.

Another aspect of automatic processing is implicit evaluation of food, which refers to an association between food stimuli and positive affect at an automatic level (Czyzewska & Graham, 2008). Unhealthy food is often evaluated positively as it is highly appetitive (Cohen & Farley, 2008). A number of studies have shown that a more positive implicit evaluation of unhealthy food predicts increased choice and consumption of unhealthy food, as well as a higher BMI (Conner, Perugini, O'Gorman, Ayres, & Prestwich, 2007; Friese et al., 2008; Haynes, Kemps, Moffitt, & Mohr, 2015; Hofmann & Friese, 2008; Hofmann et al., 2008; Hofmann, Rauch, & Gawronski, 2007; Perugini, 2005; Richetin, Perugini, Prestwich, & O'Gorman, 2007). Thus, the evidence suggests that positive implicit food evaluation may contribute to unhealthy eating behaviour.

In line with dual-process models, existing empirical research in the eating domain has demonstrated a role for different aspects of automatic processing, including attentional and approach bias for food (cognitive component), as well as implicit food evaluation (affective component), in predicting unhealthy eating behaviour and excess weight gain. These findings emphasise the importance of understanding how the automatic processing of food cues might impact upon unhealthy eating with little conscious awareness.

Research on Self-Regulatory Control and Eating Behaviour

Support for the other component of dual-process models comes from research on selfregulatory control in unhealthy eating behaviour. A growing number of studies have shown that inhibitory control, the 'ability to inhibit a behavioural impulse' (Houben, Nederkoorn, Wiers, & Jansen, 2012, p. 550), is related to unhealthy eating and weight gain. For example, Loeber et al. (2012) found that both normal weight and obese individuals are less effective at inhibiting responses to food cues relative to neutral (non-food) cues. Studies have also shown that obese individuals are less effective than their normal weight counterparts at inhibiting responses to food cues (Nederkoorn, Smulders, Havermans, Roefs & Jansen, 2006). Poor inhibitory control has been associated with increased consumption of unhealthy food in both normal weight (Guerrieri, Nederkoorn, & Jansen, 2007b) and overweight women (Appelhans et al., 2011). Finally, poor inhibitory control predicted an increase in BMI over a one-year period in normal weight women (Nederkoorn, et al., 2010). Together, these studies suggest that poor state self-regulatory control for food cues may undermine the ability to resist unhealthy food cues, which in the long-term, could lead to overeating and weight gain.

While studies have often considered the role of state self-regulatory control in eating behaviour, there are also stable differences. Specifically, trait impulsivity refers to a general

tendency to act or think without regarding the consequences (Hofmann et al., 2008). Studies have shown that impulsivity is positively associated with unhealthy eating behaviour, such as increased food consumption during a laboratory taste test (Appelhans et al., 2011; Guerrieri, Nederkoorn & Jansen, 2007a; Guerrieri et al., 2007) and a higher BMI (Batterink, Yokum, & Stice, 2010; Cohen, Yates, Duong & Convit, 2011). Highly impulsive people are also more likely to be overweight or obese (Nederkoorn, et al., 2010; Ryden et al., 2003; Guerrieri, Nederkoorn & Jansen, 2008). In support of dual-process models, the previous literature has shown a role for different aspects of self-regulatory control, including inhibitory control for food-related cues and trait impulsivity, in predicting unhealthy eating behaviour and excess weight gain. The findings highlight the importance of understanding self-regulatory control as it may determine the ability to resist appetitive food and related cues in the environment.

Research on Trait Eating Style and Eating Behaviour

Trait eating style refers to individual differences in the habitual way of eating due to early learning experiences, which develop over time to become a stable characteristic of that person (Carnell & Wardle, 2007; Larsen, van Strien, Eisinga, & Engels, 2006). Researchers have typically focused on three different types of eating style: restrained, emotional and external eating, most often assessed by the Dutch Eating Behaviour Questionnaire (DEBQ; van Strien, Frijters, Bergers, & Defares, 1986). Restrained eating refers to the tendency to restrict intake of food; the major component of Herman and Polivy's (1980) restraint theory. Emotional eating refers to the tendency to eat in response to negative affect due to a lack of interoceptive awareness and is derived from Bruch's (1964) psychosomatic theory of obesity. External eating is the tendency to eat when exposed to external food-related cues, such as the sight or smell of food, in line with Schachter's (1968) classic externality theory of obesity.

A number of studies (but not all) have shown that the three types of eating style are associated with increased unhealthy food consumption (e.g., Anschutz, van Strien, van De Ven, & Engels, 2011; Elfhag, Tholin, & Rasmussen, 2008; Wardle et al., 2002) and higher BMI in unselected samples (e.g., Burton, Smit, & Lightowler, 2007; Elfhag & Linne, 2005; Jasinska et al., 2012; Snoek, van Strien, Janssens, Rutger, & Engels, 2007). Of these eating styles, the most reliable findings have been demonstrated for external eating. Taken together, the evidence suggests that trait eating styles contribute to eating behaviour and weight gain.

Eating styles primarily comprise factors such as personality traits (Elfhag & Morey, 2008), emotional regulation (Topham et al., 2011), and feeding practices in the early home environment (Kral & Faith, 2009), but also involve aspects of cognitive processing. Research has shown that eating styles are related to automatic processing (Ayres, Prestwich, Connor & Smith, 2011; Brignell et al., 2009; Hollitt et al., 2010; Hou et al., 2011; Kemps, et al., 2013; Nijs et al., 2009; Veenstra & de Jong; 2010) and self-regulatory control (Elfhag & Morey, 2008; Hou et al., 2011). Thus, evidence suggests that eating styles might include both types of processes. Cognitive biases (attentional and approach) may contribute to the development of eating style and in turn be reinforced by them, as both are likely to be determined by trait sensitivity to rewarding food cues (Brignell et al., 2009). Likewise, inhibitory control may play a role in, and be influenced by eating style characteristics that involve self-regulation.

Automatic Processing, Self-Regulatory Control and Trait Eating Style

A growing body of research shows that both automatic processes and self-regulatory control individually determine unhealthy eating behaviour. While most research to date has examined these two types of processes separately, contemporary dual-process models posit that behaviour is guided by their combination. In particular, biased automatic processing and poor self-regulatory control are both likely contributors to unhealthy eating behaviour. At the outset of the current thesis, the interaction between automatic processing and self-regulatory control in predicting behaviour was yet to be examined in the food domain. Examining both types of processes together may contribute to a greater understanding of why people engage in unhealthy eating behaviour.

Furthermore, individual differences in eating style might influence susceptibility to biased automatic processing of unhealthy food and the ability to regulate responses to such food. The studies presented in the current thesis attempted to expand upon previous research by investigating the potential interactions between automatic processing and self-regulatory control or trait eating style in the prediction of unhealthy eating behaviour. Developing an increased understanding of the mechanisms underlying unhealthy eating is important for informing and designing interventions that are aimed at modifying such behaviour.

Modifying Automatic Processing and Self-Regulatory Control

According to dual-process models, it should be possible to change behaviour through training either automatic or controlled processing. For automatic processing, researchers have begun to investigate the effectiveness of training aimed at reducing attentional and approach biases using Cognitive Bias Modification techniques (CBM; Koster, Fox, & MacLeod, 2009). Research on CBM in the eating domain has generally focused on modifying attentional bias. Previous literature reviews indicate that attentional bias modification for appetitive cues tends to produce small effects on attention and behaviour (e.g., Beard, Sawyer, & Hofmann, 2012).

A more recently researched CBM technique is approach bias modification, which aims to reduce automatic approach biases for appetitive cues. The available studies on approach bias modification and consumption behaviour suggest that it may be a useful intervention as it is generally successful at reducing consumption of appetitive substances; however, there is some variability in its effects on consumption. Therefore, it is important to determine under which conditions approach bias modification is most likely to be effective and for whom it is most beneficial. Developing a better understanding of the crucial aspects of approach bias retraining is likely to enhance its effectiveness as a technique for modifying unhealthy eating. For controlled processing, researchers have investigated the effectiveness of training aimed at increasing inhibitory control. A number of studies have shown that training people to inhibit responses to unhealthy food cues (using either a go/no-go task or stop-signal task) reduced subsequent consumption of unhealthy food (Houben, 2011; Houben & Jansen, 2011, 2015; Lawrence, Verbruggen, Morrison, Adams & Chambers, 2015; Veling, Aarts & Papies, 2011), and snack food choice (van Koningsbruggen, Veling, Stroebe & Aarts, 2014; Veling, Aarts & Stroebe, 2013a; Veling, Aarts & Stroebe, 2013b). However, a recent meta-analysis found that the effect size on consumption is small (Allom, Mullan, & Hagger, 2015), which suggests that the effectiveness of inhibitory control training can be improved.

One way of making inhibitory control training, as well as approach bias modification, more effective derives from a theoretical understanding of unhealthy eating behaviour based on contemporary dual-process models (Strack & Deutsch, 2004). According to such models, it should be more useful to modify automatic and controlled processes together. Interventions could target both processes simultaneously to reduce unhealthy eating behaviour.

Aims of the Thesis

The current thesis had two primary aims. The first was to investigate how automatic processing interacts with self-regulatory control or trait eating style in order to obtain a better understanding of these constructs in the context of unhealthy eating behaviour. Chapters 2-4 present the results of Studies 1, 2 and 3, which investigated automatic processing of food cues (cognitive bias or implicit evaluation) combined with self-regulatory control (inhibitory control) or trait eating style as predictors of unhealthy eating behaviour. The second aim of the thesis was to determine whether interventions designed to modify automatic and/or controlled processes are effective for reducing unhealthy eating and/or promoting healthy eating. A literature review was conducted on the effectiveness of approach bias modification as a technique for reducing consumption of unhealthy substances (Chapter 5). Study 4

investigated whether approach bias modification was more effective when combined with inhibitory control training to reduce unhealthy eating (Chapter 6), while Study 5 examined whether approach bias modification was effective for promoting healthy eating (Chapter 7).

Outline of the Thesis

Chapters 2-4 address the first aim of the thesis and contain reports of studies testing various combinations of automatic processing and self-regulatory control or eating styles in predicting unhealthy eating behaviour. Specifically, Chapter 2 presents the results of Study 1, which aimed to address one of the key theoretical propositions of recent dual-process models by investigating the combined effects of attentional and approach bias with inhibitory control on unhealthy food consumption. Chapter 3 presents Study 2, which examined whether the affective aspect of automatic processing and emotional eating interact in predicting unhealthy eating. Finally, Chapter 4 presents the results of Study 3, which tested the combined effect of approach bias and eating style on consumption in normal weight and overweight individuals. To obtain sufficient overweight individuals, participants from Study 1 and the control groups from Studies 4 and 5 were pooled. It was expected that approach bias and trait eating style would better predict unhealthy food consumption in overweight individuals than in normal weight individuals, as both factors have been related to unhealthy eating and excess weight.

The next three chapters (Chapters 5-7) address the second aim, which was to examine whether interventions designed to modify automatic processing and/or self-regulatory control can discourage unhealthy eating behaviour. Chapter 5 presents a review of the literature on the effectiveness of approach bias modification for appetitive substances in general (i.e., in the alcohol, cigarette, and unhealthy food domains). Chapter 6 then reports on the results of Study 4, an experiment that aimed to investigate the effectiveness of reducing approach bias for unhealthy food cues and strengthening inhibitory control to discourage unhealthy eating behaviour. This study builds upon the finding from Study 1 (Chapter 2), which showed that

approach bias and inhibitory control interactively predicted greater consumption of unhealthy food. Chapter 7 presents the results of Study 5, an experiment that aimed to extend the use of approach bias modification from only reducing unhealthy eating behaviour to also increasing healthy eating. An aspect of self-regulatory control, namely, trait impulsivity, was assessed as a potential moderator of the effectiveness of training.

The final chapter (Chapter 8) presents a general discussion of the main findings from all of the studies. It also considers the theoretical and practical implications for the main aims of the thesis and provides recommendations for future research directions.

All of the chapters in the current thesis (except for Chapters 1 and 8) are formatted as manuscripts for publication. Chapter 2 is published in the journal *Appetite* and Chapter 5 is published in the journal *Addictive Behaviors*. Chapter 6 is currently under revision, while Chapters 3, 4, and 7 are under review. The formatting of each chapter varies slightly as the manuscripts were prepared according to the requirements of each particular journal. There is considerable repetition of the background information and methodology in the Introduction and Method sections of each chapter.

References

- Allom, V., Mullan, B., & Hagger, M. (2015). Does inhibitory control training improve health behaviour? A meta-analysis. *Health Psychology Review*, 1-19.
- Anschutz, D. J., van Strien, T., Van De Ven, M. O., & Engels, R. C. (2009). Eating styles and energy intake in young women. *Appetite*, *53*, 119–22.
- Appelhans, B. M., Woolf, K., Pagoto, S. L., Schneider, K. L., Whited, M. C., & Liebman, R. (2011). Inhibiting food reward: delay discounting, food reward sensitivity, and palatable food intake in overweight and obese women. *Obesity*, *19*, 2175-2182.
- Ayres, K., Prestwich, A., Conner, M. T., & Smith, P. (2011). Emotional eating moderates the relationship between implicit measures of attitudes and chocolate consumption. *European Journal of Personality*, 25, 317-325.
- Batterink, L., Yokum, S., & Stice, E. (2010). Body mass correlates inversely with inhibitory control in response to food among adolescent girls: an fMRI study. *Neuroimage, 52*, 1696-1703.
- Beard, C., Sawyer, A. T., & Hofmann, S. G. (2012). Efficacy of attention bias modification using threat and appetitive stimuli: A meta-analytic review. *Behavior Therapy*, 43, 724-740.
- Brignell, C., Griffiths, T., Bradley, B. P., & Mogg, K. (2009). Attentional and approach biases for pictorial food cues. Influence of external eating. *Appetite*, 52, 299–306.
- Brunner, T. A., van der Horst, K., & Siegrist, M. (2010). Convenience food products. Drivers for consumption. *Appetite*, *55*, 498–506.
- Bruch, H. (1964). Psychological aspects in overeating and obesity. *Psychosomatics*, *5*, 269–274.
- Burton, P., Smit, H. J., & Lightowler, H. J. (2007). The influence of restrained and external eating patterns on overeating. *Appetite*, *49*, 191–197.

- Calitri, R., Pothos, E. M., Tapper, K., Brunstrom, J. M., & Rogers, P. J. (2010). Cognitive biases to healthy and unhealthy food words predict change in BMI. *Obesity*, 18, 2282–2287.
- Carnell, S., & Wardle, J. (2007). Measuring behavioural susceptibility to obesity: validation of the child eating behaviour questionnaire. *Appetite*, *48*, 104-113.
- Castellanos, E. H., Charboneau, E., Dietrich, M. S., Park, S., Bradley, B P., Mogg, K., & Cowan, R. L. (2009). Obese adults have visual attention bias for food cue images:
 Evidence for altered reward system function. *International Journal of Obesity*, *33*, 1063–1073.
- Cohen, D.A., & Farley, T.A. (2008). Eating as an automatic behavior. *Preventing Chronic Disease*, *5*, 1-7.
- Cohen, J. I., Yates, K. F., Duong, M., & Convit, A. (2011). Obesity, orbitofrontal structure and function are associated with food choice: A cross-sectional study. *BMJ open*, *1*(2), e000175.
- Conner, M. T., Perugini, M., O'Gorman, R., Ayres, K., & Prestwich, A. (2007). Relations between implicit and explicit measures of attitudes and measures of behavior:
 Evidence of moderation by individual difference variables. *Personality and Social Psychology Bulletin, 33*, 1727-1740.
- Coskunpinar, A., & Cyders, M. A. (2013). Impulsivity and substance-related attentional bias. A meta-analytic review. *Drug and Alcohol Dependence*, *133*, 1–14.
- Czyzewska, M., & Graham, R. (2008). Implicit and explicit attitudes to high-and low-calorie food in females with different BMI status. *Eating Behaviors*, *9*, 303-312.
- Elfhag, K., & Morey, L. C. (2008). Personality traits and eating behavior in the obese: poor self-control in emotional and external eating but personality assets in restrained eating. *Eating Behaviors*, 9, 285-293.

- Elfhag, K., & Linn., Y. (2005). Gender differences in associations of eating pathology between mothers and their adolescent offspring. *Obesity Research, 13*, 1070–1076.
- Elfhag, K., Tholin, S., & Rasmussen, F. (2008). Consumption of fruit, vegetables, sweets and soft drinks are associated with psychological dimensions of eating behaviour in parents and their 12-year-old children. *Public Health Nutrition*, *11*, 914-923.
- Field, M., & Cox, W. M. (2008). Attentional bias in addictive behaviors. A review of its development, causes, and consequences. *Drug and Alcohol Dependence*, 97, 1–20.
- Guerrieri, R., Nederkoorn, C., & Jansen, A. (2007a). How impulsiveness and variety influence food intake in a sample of healthy women. *Appetite*, *48*, 119-122.
- Guerrieri, R., Nederkoorn, C., & Jansen, A. (2007b). The interaction between impulsivity and a varied food environment: its influence on food intake and overweight. *International Journal of Obesity*, *32*, 708-714.
- Guerrieri, R., Nederkoorn, C., Stankiewicz, K., Alberts, H., Geschwind, N., Martijn, C., & Jansen, A. (2007). The influence of trait and induced state impulsivity on food intake in normal-weight healthy women. *Appetite*, *49*, 66-73.
- Haynes, A., Kemps, E., Moffitt, R. and Mohr, P. (2015). Reduce temptation or resist it?
 Experienced temptation mediates the relationship between implicit evaluations of unhealthy snack foods and subsequent intake. *Psychology and Health, 30*, 534-550.
- Havermans, R. C., Giesen, J. C., Houben, K., & Jansen, A. (2011). Weight, gender, and snack appeal. *Eating Behaviors*, *12*, 126-130.
- Herman, C. P., & Polivy, J. (1980). Restrained eating. In A. J. Stunkard (Ed.), *Obesity* (pp. 208–225). Philadelphia: Saunders.
- Hill, J. O., Wyatt, H. R., Reed, G. W., & Peters, J. C. (2003). Obesity and the environment:Where do we go from here? *Science*, 299, 853-855.

- Hofmann, W., & Friese, M. (2008). Impulses got the better of me: Alcohol moderates the influence of implicit attitudes toward food cues on eating behavior. *Journal of Abnormal Psychology*, 117, 420-427.
- Hofmann, W., Friese, M., & Wiers, R. W. (2008). Impulsive versus reflective influences on health behavior: A theoretical framework and empirical review. *Health Psychology Review*, 2, 111-137.
- Hofmann, W., Rauch, W., & Gawronski, B. (2007). And deplete us not into temptation:
 Automatic attitudes, dietary restraint, and self-regulatory resources as determinants of eating behavior. *Journal of Experimental Social Psychology*, 43, 497-504.
- Hofmann, W., Gschwendner, T., Wiers, R., Friese, M., & Schmitt, M. (2008b). Working memory capacity and self-regulation. Towards an individual differences perspective on behavior determination by automatic versus controlled processes. *Journal of Personality and Social Psychology*, 95, 962–977
- Hollit, S., Kemps, E., Tiggemann, M., Smeets, E. & Mills, J. S. (2010). Components of attentional bias for food cues among restrained eaters. *Appetite*, *54*, 309–313.
- Hou, R., Mogg, K., Bradley, B., Moss-Morris, R., Peveler, R., & Roefs, A. (2011). External eating, impulsivity and attentional bias to food cues. *Appetite*, *56*, 424-427.
- Houben, K. (2011). Overcoming the urge to splurge: Influencing eating behavior by manipulating inhibitory control. *Journal of Behavior Therapy and Experimental Psychiatry*, 42, 384-388.
- Houben, K., & Jansen, A. (2011). Training inhibitory control. A recipe for resisting sweet temptations. *Appetite*, *56*, 345-349.
- Houben, K., & Jansen, A. (2015). Chocolate equals stop. Chocolate-specific inhibition training reduces chocolate intake and go associations with chocolate. *Appetite*, 87, 318-323.

- Houben, K., Nederkoorn, C., Wiers, R. W., & Jansen, A. (2011). Resisting temptation:
 Decreasing alcohol-related affect and drinking behavior by training response
 inhibition. *Drug and Alcohol Dependence*, *116*, 132–136.
- Jasinska, A. J., Yasuda, M., Burant, C. F., Gregor, N., Khatri, S., Sweet, M., & Falk, E. B. (2012). Impulsivity and inhibitory control deficits are associated with unhealthy eating in young adults. *Appetite*, *59*, 738-747.
- Kemps, E., & Tiggemann, M. (2015). Approach bias for food cues in obese individuals. *Psychology & Health, 30*, 370-380.
- Kemps, E., Tiggemann, M., Martin, R., & Elliott, M. (2013). Implicit approach–avoidance associations for craved food cues. *Journal of Experimental Psychology: Applied, 19*, 30-38.
- Kral, T. V., & Faith, M. S. (2009). Influences on child eating and weight development from a behavioral genetics perspective. *Journal of Pediatric Psychology*, 34, 596-605.
- Larsen, J. K., van Strien, T., Eisinga, R., & Engels, R. C. (2006). Gender differences in the association between alexithymia and emotional eating in obese individuals. *Journal of Psychosomatic Research*, 60, 237-243.
- Lawrence, N. S., Verbruggen, F., Morrison, S., Adams, R. C., & Chambers, C. D. (2015a). Stopping to food pictures reduces food intake: Effects of cue specificity, control conditions and individual differences. *Appetite*, 85, 91-103.
- MacLeod, C., & Matthews, A. (2012). Cognitive bias modification approaches to anxiety. *Annual Review Clinical Psychology*, 8, 189–217.
- Marteau, T. M., Hollands, G. J., & Fletcher, P. C. (2012). Changing human behavior to prevent disease: The importance of targeting automatic processes. *Science*, 337, 1492-1495.

- Mogg, K., Bradley, B. P., Hyare, H., & Lee, S. (1998). Selective attention to food-related in hunger: Are attentional biases specific to emotional and pathological states, or are they found in normal drive states? *Behaviour Research and Therapy*, *36*, 227-237.
- Mogg, K., Bradley, B. P., O'Neill, B., Bani, M., Merlo-Pich, E., Koch, A., ... & Nathan, P. J. (2012). Effect of dopamine D3 receptor antagonism on approach responses to food cues in overweight and obese individuals. *Behavioural Pharmacology*, 23, 603-608.
- Nederkoorn, C., Smulders, F. T., Havermans, R. C., Roefs, A., & Jansen, A. (2006). Impulsivity in obese women. *Appetite*, 47, 253-256.
- Nederkoorn, C., Houben, K., Hofmann, W., Roefs, A., & Jansen, A. (2010). Control yourself or just eat what you like? Weight gain over a year is predicted by an interactive effect of response inhibition and implicit preference for snack foods. *Health Psychology*, 29, 389-393.
- Nijs, I. M., Franken, I. H., & Muris, P. (2009). Enhanced processing of food-related pictures in female external eaters. *Appetite*, *53*, 376-383.
- Nijs, I. M., Muris, P., Euser, A. S., Franken, I. H. (2010). Differences in attention to food and food intake between overweight/obese and normal-weight females under conditions of hunger and satiety. *Appetite*, 54, 243–254.
- Perugini, M. (2005). Predictive models of implicit and explicit attitudes. *British Journal of Social Psychology*, 44, 29-45.
- Polivy, J., Herman, C. P., & Coelho, J. S. (2008). Caloric restriction in the presence of attractive food cues: External cues, eating, and weight. *Physiology and Behavior*, 94, 729–733.
- Pothos, E. M., Tapper, K., & Calitri, R. (2009). Cognitive and behavioral correlates of BMI among male and female undergraduate students. *Appetite*, *52*, 797-800.

- Richetin, J., Perugini, M., Prestwich, A., & O'Gorman, R. (2007). The IAT as a predictor of food choice: The case of fruits versus snacks. *International Journal of Psychology*, 42, 166-173.
- Ryden, A., Sullivan, M., Torgerson, J. S., Karlsson, J., Lindroos, A. K., & Taft, C. (2003). Severe obesity and personality: A comparative controlled study of personality traits. *International Journal of Obesity*, 27, 1534–1540.

Schachter, S. (1968). Obesity and eating. Science, 161, 751–756.

- Snoek, H. M., Van Strien, T., Janssens, J. M. A. M., & Engels, R. C. M. E. (2007). Emotional, external, restrained eating and overweight in Dutch adolescents. *Scandinavian Journal of Psychology*, 48, 23–32.
- Strack, F., & Deutsch, R. (2004). Reflective and impulsive determinants of social behavior. Personality and Social Psychology Review, 8, 220–247.
- Topham, G. L., Hubbs-Tait, L., Rutledge, J. M., Page, M. C., Kennedy, T. S., Shriver, L. H., & Harrist, A. W. (2011). Parenting styles, parental response to child emotion, and family emotional responsiveness are related to child emotional eating. *Appetite*, 56, 261-264.
- Van Koningsbruggen, G. M., Veling, H., Stroebe, W., & Aarts, H. (2014). Comparing two psychological interventions in reducing impulsive processes of eating behaviour:
 Effects on self-selected portion size. *British Journal of Health Psychology*, *19*, 767-782.
- Veling, H., Aarts, H., & Papies, E. K. (2011). Using stop signals to inhibit chronic dieters' responses toward palatable foods. *Behaviour Research and Therapy*, 49, 771-780.
- Veling, H., Aarts, H., & Stroebe, W. (2013a). Using stop signals to reduce impulsive choices for palatable unhealthy foods. *British Journal of Health Psychology*, *18*, 354–368.

- Veling, H., Aarts, H., & Stroebe, W. (2013b). Stop signals decrease choices for palatable foods through decreased food evaluation. *Frontiers in Psychology*, 4, 1-7.
- Veenstra, E. M., & de Jong, P. J. (2010). Restrained eaters show enhanced automatic approach tendencies towards food. *Appetite*, *55*, 30-36.
- Wardle, J. (1988). Cognitive control of eating. *Journal of Psychosomatic Research*, *32*, 607-612.
- Wardle, J., Marsland, L., Sheikh, Y., Quinn, M., Fedoroff, I., & Ogden, J. (1992). Eating style and eating behaviour in adolescents. *Appetite*, *18*, 167-183.
- Werthmann, J., Roefs, A., Nederkoorn, C., Mogg, K., Bradley, B. P., & Jansen, A. (2011).Can (not) take my eyes off it: Attention bias for food in overweight participants.*Health Psychology*, *30*, 561-569.
- Wiers, R. W., Gladwin, T. E., Hofmann, W., Salemink, E., & Ridderinkhof, K. R. (2013).
 Cognitive bias modification and cognitive control training in addiction and related psychopathology mechanisms, clinical perspectives, and ways forward. *Clinical Psychological Science*, 1, 192-212.
- World Health Organization [WHO]. (2014). *Obesity and overweight*. Retrieved October 1, 2015 from http://www.who.int/mediacentre/factsheets/fs311/en/

CHAPTER 2: STUDY 1

Combined Effects of Cognitive Bias for Food Cues and Poor Inhibitory Control on Unhealthy Food Intake

Naomi Kakoschke, Eva Kemps, and Marika Tiggemann School of Psychology, Flinders University, South Australia

Corresponding author: Naomi Kakoschke, School of Psychology, Flinders University,

GPO Box 2100, Adelaide, 5001, South Australia. Email: naomi.kakoschke@flinders.edu.au

Funding: This research was supported under the Australian Research Council's Discovery Project funding scheme (project number DP130103092).

Statement of co-authorship: All authors were involved in the formulation of the study concept and design. Naomi Kakoschke collected the data, and completed the data analysis and the initial draft of the manuscript. Eva Kemps and Marika Tiggemann edited multiple revisions of the manuscript.

Acknowledgements: We are grateful to Paul Douglas for developing the software for the computer administered tasks.

This manuscript has been published as:

Kakoschke, N., Kemps, E., & Tiggemann, M. (2015). Combined effects of cognitive bias for food cues and poor inhibitory control on unhealthy food intake. *Appetite*, *87*, 358-364. DOI 10.1016/j.appet.2015.01.004

Abstract

The present study aimed to investigate the combined effects of cognitive bias (attentional and approach biases) and inhibitory control on unhealthy snack food intake. Cognitive biases reflect automatic processing, while inhibitory control is an important component of controlled processing. Participants were 146 undergraduate women who completed a dot probe task to assess attentional bias and an approach-avoidance task to assess approach bias. Inhibitory control was measured with a food-specific go/no-go task. Unhealthy snack food intake was measured using a so-called "taste test". There was a significant interaction between approach bias and inhibitory control on unhealthy snack food intake. Specifically, participants who showed a strong approach bias combined with low inhibitory control consumed the most snack food. Theoretically, the results support contemporary dual-process models which propose that behaviour is guided by both automatic and controlled processing systems. At a practical level, the results offer potential scope for an intervention that combines re-training of both automatic and controlled processing.

Introduction

During the last three decades, the global prevalence of overweight and obesity has doubled, with 35% of adults classified as overweight and 11% as obese (WHO, 2013). One important contributor to chronic health problems such as overweight and obesity is unhealthy eating (NHMRC, 2003). The contemporary Western diet is characterised by unhealthy eating, in particular consuming too much fat, salt and sugar. Given the potential negative health consequences of unhealthy eating, it is important to investigate the cognitive mechanisms that underlie such behaviour. Specifically, recent theoretical perspectives and empirical evidence suggest that automatic and controlled cognitive processing make important contributions to unhealthy behaviour.

Dual-process models (e.g., Strack & Deutsch, 2004) propose that our behaviour is determined by two different information processing systems, i.e., automatic and controlled processing. Automatic processing is fast, implicit and effortless, and includes affective (i.e., attitudes, preferences) and motivational (i.e., attending to, approaching) responses to relevant stimuli, such as unhealthy food cues. In contrast, controlled processing is effortful, slow, and explicit, and involves conscious decisions based on personal goals and standards (e.g., health and weight loss). These two processing systems elicit conflicting signals, and the outcome is determined by the relative strength of each processing system. According to dual-process models, behaviour is guided by automatic processing and regulated by controlled processing (if cognitive resources are available). For example, the presence of unhealthy food cues may elicit a conflict between the two systems, i.e., automatically attending to and approaching such cues while maintaining incompatible goals related to health and weight. Thus, a strong automatic system (an attentional or approach bias for food cues) and a weak controlled system (poor inhibitory control or working memory capacity) may result in unhealthy eating.

Automatic and controlled processing systems have given rise to two separate lines of research. Support for the role of automatic processing in eating behaviour generally comes from research investigating cognitive biases for food cues. A cognitive bias refers to "systematic selectivity in information processing that operates to favour one type of information over another" (MacLeod & Matthews, 2012, p. 191). Most research has focused on attentional bias, which refers to the automatic allocation of attention to food cues in preference to other cues (MacLeod & Matthews, 2012). More recently, researchers have turned their focus toward approach bias, which is the automatic behavioural tendency to move toward rather than avoid food cues (Wiers et al., 2013a). Studies have demonstrated biased attentional processing of high-caloric food cues in relation to neutral (non-food) cues in healthy weight participants (Nijs, Franken, & Muris, 2010; Werthmann et al., 2013). Both attentional and approach biases for food cues have also been documented in populations with eating-related issues. Specifically, restrained (Hollitt, Kemps, Tiggemann, Smeets, & Mills, 2010; Veenstra & de Jong, 2010) and external eaters (Brignell, Griffiths, Bradley, & Mogg, 2009; Hou et al., 2011; Nijs, Franken, & Muris, 2009), as well as overweight and obese individuals (Castellanos et al., 2009; Havermans, Giesen, Houben, & Jansen, 2011; Nijs et al., 2010a; Nijs, Muris, Euser, & Franken, 2010b), are faster to detect and approach highcaloric food cues relative to neutral cues.

Furthermore, research has demonstrated a positive correlation between attentional biases for unhealthy food cues (e.g., cake, salted peanuts) and the subsequent consumption of snack foods during a laboratory taste test in both healthy weight and obese participants (Nijs, Muris, Euser, & Franken, 2010; Werthmann et al., 2011). Research from our laboratory (Kakoschke, Kemps, & Tiggemann, 2014; Kemps, Tiggemann, Martin, & Elliot, 2013; Kemps, Tiggemann, Orr, & Grear, 2014), as well as others (Werthmann, Field, Roefs, Nederkoorn, & Jansen, 2013), has also found that experimentally reducing an attentional bias for unhealthy food cues decreases unhealthy food intake. This evidence is consistent with the idea that cognitive biases for food cues play a causal role in consumption (Berridge, 2009). Similar findings have also been shown for alcohol (Field & Eastwood, 2005) and cigarettes (Attwood, O'Sullivan, Leonards, Mackintosh, & Munafo, 2008). Nevertheless, the evidence is mixed as some studies have found no such link between attentional bias and consumption of food (Hardman, Rogers, Etchells, Houstoun, & Munafo, 2013), alcohol (Field et al., 2007; Fadardi & Cox, 2009), and cigarettes (Field, Duka, Tyler, & Schoenmakers, 2009). In contrast, the smaller amount of research on approach bias shows a more consistent link between approach bias and consumption of alcohol (Wiers et al., 2009; Wiers, Rinck, Kordts, Houben, & Strack 2010) and cannabis (Cousijn et al., 2011). One possible explanation for these contradictory findings is that attentional and approach biases behave differently, as has been evidenced by research in the alcohol domain. Specifically, Sharbanee, Stritzke, Wiers, and MacLeod (2013) demonstrated that these two cognitive biases are distinct mechanisms that can make independent contributions to consumption. Another potential explanation for the overall mixed evidence is that the previous research has not taken into account the role of controlled processing in consumption.

Research investigating the role of controlled processing in eating behaviour has primarily focused on inhibitory control (or response inhibition), which has been defined as "the ability to inhibit a behavioural impulse in order to attain higher-order goals, such as weight loss" (Houben, Nederkoorn, & Jansen, 2012, p. 550). A recent study by Loeber et al. (2012) found that both healthy weight and obese participants were less effective at inhibiting behavioural responses to food cues relative to neutral (non-food) cues. Furthermore, studies have shown that obese participants were less effective at inhibiting responses to neutral cues (Nederkoorn, Braet, Van Eijs, Tanghe, & Jansen, 2006) as well as to food-related cues (Nederkoorn, Smulders, Havermans, Roefs & Jansen, 2006) than healthy weight participants. Several studies have also demonstrated that poor inhibitory control is associated with increased food intake during a laboratory taste test in both healthy weight (Guerrieri, Nederkoorn, & Jansen, 2007) and overweight or obese women (Appelhans et al., 2011). In addition, poor inhibitory control predicted an increase in weight (BMI) over a one year period in a sample of healthy weight women (Nederkoorn, Houben, Hofmann, Roefs & Jansen, 2010). Some studies have also shown that experimentally increasing inhibitory control reduces chocolate (Houben & Jansen, 2011) and alcohol (Houben, Nederkoorn, Wiers, & Jansen, 2011) consumption; however, Guerrieri et al. (2007b) found that experimentally increasing behavioural inhibitory control is related to working memory capacity, which is the ability to store and process goal-relevant information (Conway, Kane & Engle, 2003). A recent study found that experimentally increasing working memory capacity reduced alcohol intake in a sample of problem drinkers (Houben, Wiers, & Jansen, 2011).

As indicated above, prior research has largely focused on automatic or controlled processing separately. However, it may be their combination that is most important for consumption. In line with dual-process models, recent meta-analyses suggest that a cognitive bias for appetitive cues combined with poor inhibitory control may result in unhealthy behaviour, such as consuming appetitive substances like drugs and alcohol (Field & Cox, 2008; Coskunpinar & Cyders, 2013). Nederkoorn et al. (2010) investigated this theoretical prediction in the food domain and found that automatic and controlled processing interacted in determining an increase in BMI over a one year period in healthy weight women. Specifically, women with strong implicit preferences for food and low inhibitory control gained the most weight. Other studies have shown that the combination of strong implicit preferences and low inhibitory control predicts candy (Hofmann, Friese, & Roefs, 2009) and alcohol (Houben & Wiers, 2009) intake on a laboratory taste test. The above studies measured automatic processing with the implicit association task, which assesses evaluative attitudes for appetitive cues. However, we chose to focus on the motivational bias component of automatic processing. Similarly, in the alcohol domain, Peeters et al. (2012) recently found that the combination of an approach bias for alcohol and low inhibitory control (measured by the Stroop task) predicted alcohol use in adolescent drinkers. To the best of our knowledge, this finding has not been demonstrated in the food domain. In addition, the above studies have all measured inhibitory control in general, not specifically related to food. Yet specific food-related inhibitory control needs to be examined as a more proximal potential mechanism associated with unhealthy eating (Appelhans et al., 2011; Meule et al., 2014).

The aim of the current study was to investigate the combined effects of automatic and controlled processing on unhealthy food intake. Cognitive biases for food cues were assessed as an indicator of automatic processing, and food-specific inhibitory control was assessed as an important component of controlled processing. Both of the two main forms of cognitive bias, namely attentional bias and approach bias were included. Attentional bias was assessed by the often used dot probe task, developed by Macleod, Matthews and Tata (1986). Approach bias was assessed by the approach-avoidance task of Rinck and Becker (2007). Inhibitory control was assessed using the food-related go/no-go task of Houben and Jansen (2011). A so-called "taste test" was used to measure unhealthy food consumption. It was predicted that a stronger cognitive bias together with lower inhibitory control would lead to increased unhealthy food intake. This was tested for the two different components of cognitive bias (attentional and approach) separately.

Method

Participants

Participants were 146 women recruited from the Flinders University undergraduate student population. They were aged 18-25 years (M = 20.20, SD = 2.64). Most participants

were within the healthy weight range (i.e. $18.5-24.9 \text{ kg/m}^2$) with a mean BMI of 22.9 kg/m², (*SD* = 5.11). Only women were recruited as they have shown a greater tendency to overeat (Burton, Smit, & Lightowler, 2007). Participants were included if they spoke English as their first language, liked most foods, and did not have any food allergies or dietary requirements. As hunger has been linked to cognitive biases for food cues (Mogg, Bradley, Hyare, & Lee, 1998; Seibt, Hafner, & Deutsch, 2007) and poor inhibitory control (Nederkoorn, Guerrieri, Havermans, Roefs, & Jansen, 2009), participants were instructed to eat something two hours before the testing session to ensure they were not hungry. All participants reported having complied with this instruction. Participants also rated their current level of hunger on a 100 mm visual analogue scale ranging from "not hungry at all" to "extremely hungry" (Grand, 1968). Mean hunger ratings were around the mid-point of the scale (M = 49.7, SD = 24), and did not correlate with measures of cognitive biases, inhibitory control, or consumption (.03 < r < .08, ps > .05).

Materials

Stimuli. The stimuli were 80 digital coloured photographs (presented in a resolution of approximately 1024 x 768 pixels) comprising 20 food and 60 animal pictures. The food pictures depicted food items high in sugar, salt and/or fat (e.g., chocolate, potato chips). Animals were chosen for the comparison category of neutral non-food stimuli as they, like food, are overall appealing. Animal pictures depicted well-liked species that are not commonly consumed in Western society (e.g., giraffe, koala.). For the dot probe task, the pictures were divided into 20 critical pairs (food-animal) and 20 control non-food pairs (animal-animal). Each picture pair was individually matched on characteristics such as quality, brightness, and size, as well as ratings of pleasure and arousal. These ratings were obtained from a previous pilot study, in which 21 women aged 17-45 years (M = 23.67, SD =

8.28) rated 590 food and animal pictures on 9-point pleasure and arousal scales (Kemps,Tiggemann, & Hollitt, 2014). The same stimulus set was used for all three computer tasks.

Dot probe task. Following the procedure of MacLeod et al. (1986), each trial began with the display of a fixation cross in the centre of the computer screen presented for 500 ms. This was followed by the presentation of a picture pair for 500 ms. The pictures were displayed on the left and right hand side of the screen and each were an equal distance from the centre. When the pictures disappeared, a small dot probe appeared in the location of one of the pictures and remained there until the participant responded. Participants were asked to press the corresponding key on the keyboard (the key labelled "L" for left or "R" for right) to indicate, as quickly as possible, whether the dot probe replaced the picture on the left or right hand side of the screen. Each picture pair was presented four times so that every combination of the replacement of the dot probe position (left or right picture location and left or right dot probe location) was presented, for a total of 160 trials.. The picture pairs were presented in a different random order for each participant and the dot probes replaced the pictures in each pair with equal frequency (50/50). Participants' reaction times (ms) were recorded on each trial. An attentional bias score was calculated for each participant by subtracting the reaction times to the dot probes replacing unhealthy food pictures from the reaction times to the dot probes replacing animal pictures. Therefore, positive scores indicate an attentional bias towards unhealthy food, while negative scores indicate an attentional bias away from unhealthy food.

Approach-Avoidance task. Following Rinck and Becker (2007), the approachavoidance task was used to measure an approach-avoidance bias. This task, originally developed for anxiety disorders was adapted here for the food domain. The approachavoidance task consisted of 160 trials. On each trial, participants began by pressing the start button on the top of a joystick. A picture of an unhealthy food or an animal then appeared in
the centre of the screen. Participants were instructed to pull (approach) or push (avoid) the joystick according to whether the picture was presented in portrait or landscape format. These instructions were counterbalanced (i.e., half of the participants pulled for portrait and pushed for landscape and half vice versa). Pulling the joystick increased the picture size (as if physically approaching the picture), while pushing the joystick decreased the picture size (as if moving away from the picture). The picture disappeared once the participant had pulled or pushed the joystick. Participants were asked to respond as quickly and accurately as possible. Each picture was presented four times, twice in portrait format and twice in landscape format, and participants were asked to pull and push the food and animal pictures with equal frequency (50/50). The joystick was part of an apparatus that was connected to the table to prevent it from moving around during the task.

Participants' reactions times were recorded on each trial. For each participant, reaction time scores were calculated for pulling and pushing the food and animal pictures. The primary outcome measure was approach bias for food, which was calculated as the difference between median pushing and pulling reaction times for food pictures. Positive scores indicate an approach bias for food (i.e., tendency to pull faster than push an image), whereas negative scores indicate an avoidance bias for food (i.e., faster push than pull) (Wiers et al., 2013).

Go/no-go task. Following Houben and Jansen (2011), a food-related version of the go/no-go task was used to measure inhibitory control. The go/no-go task consisted of two blocks of 160 trials. On each trial, a picture was presented together with a go or a no-go cue (i.e., the letters 'p' or 'f') for 1500 ms. The go/no-go cues were displayed in black font type and were presented randomly in one of four locations near the outside corners of the pictures. Participants were instructed to press the space bar when a go cue was displayed on the picture, and to refrain from responding when a no-go cue was displayed. Instructions

regarding letter type ('p' versus 'f') and response assignment (go versus no-go) were counterbalanced. The food and animal pictures were each presented eight times so that every combination of letter type ('p' or 'f') and letter location (left: top, bottom; right: top, bottom) was presented. Pictures were presented in a different random order for each participant, and food and animal pictures were paired with a go cue with equal frequency (50/50). The number of commission errors (i.e., space bar pressed in response to a no-go cue) and the number of omission errors (i.e., space bar not pressed in response to a go cue) were recorded. Following previous studies (e.g., Bezdjian, Baker, Lozano, & Raine, 2009; Muele & Kubler, 2014), the primary outcome measure was the percentage of commission errors in response to food pictures, with a higher percentage of errors indicative of poorer inhibitory control

Consumption. Consumption was measured using a so-called taste test. Participants were presented with a platter comprising four snacks (two sweet and two savoury): milk-chocolate M&Ms, chocolate-chip biscuits, plain potato chips, and pretzels. The four foods were presented in equally-filled separate bowls and were chosen as they are commonly consumed and are bite-sized to facilitate eating. The presentation order of the bowls was counterbalanced across participants using a 4×4 Latin square. Participants were instructed to taste and rate each snack on several dimensions (e.g., flavour, likelihood of purchase). They were given 10 minutes to complete their ratings and told that they could try as much of the food as they liked. The amount of each food consumed was calculated by subtracting the weight (in grams) of the snacks after the taste test from the weight of the snacks before the taste test. The weight in grams for each food was then converted into the number of calories consumed and summed to obtain a total measure of intake.

Procedure

The experiment took place in the Food Laboratory in the School of Psychology at Flinders University, South Australia. The testing session took approximately one hour. After providing informed consent, participants completed a brief demographics questionnaire, followed by the three computer tasks presented in counterbalanced order, and finally the taste test. The study was approved by the University's Social and Behavioural Research Ethics Committee.

Results

Statistical considerations

Data were examined to ensure assumptions underlying statistical analysis were met. Participants' response times for attentional bias and approach bias, as well as commission errors and consumption that were more than 3 SD from the mean were identified as outliers and were changed to plus or minus one unit from the first non-outlier (Tabachnick & Fidell, 1989). An alpha value of .05 was used to determine significant p values.

Descriptive statistics

As expected, participants showed an attentional bias for food cues as indicated by a positive mean bias score, although the large standard deviation indicated that there was a considerable range in scores (M = 8.26, SD = 24.59). To formally test this, a 2 (target [food] location: left vs. right) × 2 (probe/response location: left vs. right) repeated measures ANOVA was conducted on the mean reaction times of the critical trials. The means, as presented in Table 1, showed no main effects of target location, F(1, 145) = 1.95, p = .165, or probe location, F(1, 145) = .114, p = .736. However, the interaction was significant, F(1, 145) = 16.66, p < .001. The results confirm that, irrespective of probe/response position, participants were faster to respond on compatible trials (when the probe replaced the target [food]) (M = 373.46, SD = 51.61), than on incompatible trials (when the probe replaced the non-target [animal]) (M = 381.79, SD = 54.98), which shows an attentional bias for food cues.

Table 2 presents the approach bias scores for both food and animal cues (although only the approach bias for food is of interest). As expected, participants showed an approach

bias for food cues, as indicated by a positive mean bias score (M = 32.32, SD = 103.09). They also showed an approach bias for animal cues (M = 31.05, SD = 127.11). A 2 (picture type: food vs. animal) × 2 (joystick movement: push vs. pull) repeated measures ANOVA confirmed a significant main effect of joystick movement, F(1, 145) = 12.132, p = .001, whereby participants were faster to pull the joystick ('approach') than to push it. The main effect of picture type, F(1, 145) = .024, p = .876, and the interaction were not significant, F(1, 145) = .045, p = .833. This indicates that participants showed an approach bias for food (and animals).

Table 1

Mean reaction times in ms (with standard deviations) depending on target (food) location and probe/response location.

	Target (food) location	
	Food left/animal right	Food right/animal left
Probe/response location	M (SD)	M (SD)
Left	375.33 (53.44)	380.84 (55.83)
Right	382.73 (62.05)	371.58 (57.10)

Table 2

Mean reaction times in ms (with standard deviations) depending on picture type and joystick movement.

	Picture type	
	Food left/animal right	Food right/animal left
Joystick movement	M (SD)	M (SD)
Push	917.59 (190.09)	916.54 (192.00)
Pull	885.28 (168.06)	885.49 (164.24)
Approach Bias (Push - Pull)	32.32 (103.09)	31.05 (127.11)

For inhibitory control for food, participants produced a relatively low percentage of commission errors (M = 3.04, SD = 4.81), indicating on average good inhibitory control. Overall, there were very few omission errors (M = .015%, SD = .097), indicating good attention.

Effects of cognitive bias and inhibitory control for food on consumption

A series of hierarchical multiple regression analyses was conducted to investigate the combined effects of cognitive bias and inhibitory control for food on consumption. This analysis was done separately for attentional bias and approach bias (see Fig. 1 for graphical representations of the results).

For attentional bias, centred attentional bias scores and commission errors were entered in Step 1, and the product term representing the interaction between these two variables in Step 2. As can be seen in Fig. 1a, the attentional bias results were in the predicted direction. However, they fell short of significance. The main effects of neither attentional bias, $R^2 = .007$, F(1, 143) = 1.02, p = .315, nor inhibitory control, $R^2 = .00$, F(1, 144) = .00, p = .879, explained a significant proportion of the variance in consumption. Nor did the product term explain any additional variance in consumption, $R^2_{Change} = .016$, $F_{Change}(1,$ 141) = 1.23, p = .269.

A similar analysis was conducted for approach bias. In this case, approach bias for food, approach bias for animals, and inhibitory control for food were entered in Step 1, and the two-way interaction terms with inhibitory control (approach bias for food × inhibitory control; approach bias for animals × inhibitory control) were entered in Step 2. Step 1 showed no significant main effects of approach bias for food, $R^2 = .003$, F(1, 144) = .50, p = .481, approach bias for animals, $R^2 = .012$, F(1, 144) = .02, p = .881, or inhibitory control for food, $R^2 = .00$, F(1, 144) = .00, p = .879. However, Step 2 showed that the product terms explained a significant additional 8.1% of the variance in food consumption, $R^2_{Change} = .036$, F_{Change} (1, 142) = 3.597, p = .031. Specifically, the approach bias for food × inhibitory control interaction was significant ($\beta = .355$, p = .009), whereas the approach bias for animals × inhibitory control interaction was not ($\beta = -.055$, p = .567).



Fig 1. Interaction between inhibitory control and (a) attentional bias and (b) approach bias on consumption (calories).

In order to determine the form of the significant interaction, simple slopes analyses were conducted. As shown in Fig. 1b, inhibitory control for food had no effect on food consumption in women with a low approach bias, B = -1.63, t(145) = -1.02, p = .308. In contrast, for women with a high approach bias, those who made a higher percentage of commission errors (indicating lower inhibitory control) consumed significantly more food than those who made a lower percentage of commission errors (indicating higher inhibitory control), B = 6.417, t(145) = 2.04, p = .044. Thus, individuals who consumed the most unhealthy snack food showed a high approach bias for food cues and low inhibitory control (i.e., a high percentage of commission errors) for food cues.

Discussion

The aim of this study was to investigate the combined effects of automatic and controlled processing on unhealthy food intake. Attentional and approach biases were assessed as an indicator of automatic processing, and inhibitory control was assessed as an indicator of controlled processing. Results indicated that neither type of cognitive bias nor inhibitory control alone predicted unhealthy food intake. However, approach bias for food interacted with inhibitory control to predict unhealthy food intake. Specifically, for women with a higher approach bias, those with lower inhibitory control consumed more food than those with higher inhibitory control. In contrast, for women with a lower approach bias, their level of inhibitory control did not predict the amount of food consumed. These findings suggest that consumption of unhealthy food is determined by a combination of automatic and controlled processing. Specifically, participants who showed a high approach bias for food combined with low inhibitory control consumed the most unhealthy snacks.

These findings extend to the food domain previous research which has shown that implicit preferences for alcohol cues (as measured by an IAT with word stimuli) interact with inhibitory control to predict increased alcohol intake (Houben & Wiers, 2009). Further, previous research has shown that tasks which use word stimuli, like the IAT, may be less effective than tasks using pictorial stimuli, like the approach-avoidance task used in the present study, in determining automatic processing of real-world food cues (Roefs, Werrij, Smulders, & Jansen, 2006). The current study has also extended previous research on implicit preferences for food cues (Hofmann et al, 2009 and Nederkoorn et al, 2010) to another component of automatic processing, namely motivational cognitive biases. In addition, food-specific inhibitory control was measured, rather than a general inability to inhibit responses (Houben & Wiers, 2009; Peeters et al., 2012). Food-specific inhibitory control has been argued to be particularly important with regard to achieving a more detailed understanding of the mechanisms which may be associated with unhealthy eating (Appelhans et al., 2011; Meule, 2014). In support, neuroimaging research has shown that brain regions associated with motivation and disinhibition are activated in obese people in response to unhealthy food images (Carnell, Gibson, Benson, Ochner, & Geliebter, 2012).

The current findings are consistent with contemporary dual-process models (Strack & Deutsch, 2004) which propose that behaviour is determined by a combination of automatic and controlled processing. Specifically, the current study has demonstrated that motivational cognitive biases and inhibitory control interact to predict unhealthy eating behaviour. Appetitive food cues elicit automatic approach-action tendencies in some individuals. Individuals with a strong controlled system are able to inhibit responses to such cues. In contrast, those with a weaker controlled system are unable to inhibit this response, which leads to the consumption of unhealthy food. Thus, the present study provides empirical support for the theoretical predictions of dual-process models, as well as recent meta-analyses (Coskunpinar & Cyders, 2013; Field, Cox, 2008).

Neither attentional nor approach bias alone made a significant contribution to food intake. The finding for attentional bias is not surprising as previous research has been inconsistent, and some studies have shown no link with consumption (Fadardi & Cox, 2009; Field et al, 2007, Field et al, 2009 and Hardman et al, 2013). Questions have also been raised regarding the reliability of the dot probe task (Price et al, 2014 and Schmukle, 2005). Approach bias for food also did not make an independent contribution to consumption, but interacted with inhibitory control for food in predicting unhealthy eating. One possible explanation is that a main effect of approach bias will be dependent on the sample. Specifically, the interaction showed relatively less effect of approach bias on consumption for participants with high inhibitory control. In contrast, approach bias did have the predicted effect on consumption for participants with low inhibitory control. Therefore, a main effect of approach bias may be more likely found in samples with low inhibitory control (e.g., overweight or obese individuals) than in the present sample (which had on average good inhibitory control). Taken together, the results confirm that attentional and approach biases are two distinctive types of cognitive bias, which is consistent with previous research in the alcohol domain (Sharbanee et al., 2013). Although both attentional and approach biases are components of automatic processing, the current findings suggest that approach bias may be more pertinent to understanding the association between food cues and eating behaviour. This may be due to the behavioural component of approach bias (i.e., moving towards or away from food cues) in addition to the cognitive one. However, future research needs to determine whether such findings apply to other appetitive substances, such as alcohol and cigarettes.

The present study also has some important practical implications. The results suggest that strong automatic tendencies to approach food cues as well as an inability to inhibit these responses can lead to unhealthy food intake. The current study used a healthy weight sample. It remains to be tested whether these findings apply to overweight and obese populations, who have been shown to have an approach bias for food (Havermans et al., 2011) and low inhibitory control (Nederkoorn et al, 2006 and Nederkoorn et al, 2006). If so, the findings point the way towards a new form of intervention. Thus far, research has shown that automatic and controlled processes can be modified individually. For example, cognitive bias modification can be used to reduce approach biases for alcohol (Eberl et al., 2013;Wiers et al, 2011 and Wiers et al, 2010) and chocolate (Kemps, Tiggemann, Martin, & Elliott, 2013). Other research has shown that cognitive control training can increase inhibitory control for alcohol (Houben et al, 2011 and Houben et al, 2011), food in general (Houben, 2011), and chocolate cues (Houben & Jansen, 2011). However, it is likely that the effectiveness of these interventions will differ among individuals and, as yet, the combined effect has not been investigated in any population. Thus, it would be useful to determine whether a combined approach leads to a better overall success rate in clinical samples. If so, this form of implicit training might provide a useful addition to existing treatments, such as cognitive-behavioural therapy, which target the explicit aspect of cognitive processing (MacLeod, Matthews, 2012, Wiers et al, 2013 and Wiers et al, 2013).

In summary, the present study has demonstrated that a particular combination of automatic and controlled processing predicts unhealthy food intake in young women. These findings are consistent with dual-process models, and contribute to an understanding of the mechanisms that underlie unhealthy eating behaviour. Accordingly, the results have practical implications in suggesting that interventions modify both automatic and controlled processing for maximal effectiveness in discouraging unhealthy food intake. This is particularly important in contemporary Western society, which is characterised by unhealthy eating behaviour.

References

- Appelhans, B. M., Woolf, K., Pagoto, S. L., Schneider, K. L., Whited, M. C., & Liebman, R. (2011). Inhibiting food reward: Delay discounting, food reward sensitivity, and palatable food intake in overweight and obese women. *Obesity*, *19*, 2175-2182.
- Attwood, A. S., O'Sullivan, H., Leonards, U., Mackintosh, B., & Munafo, M. (2008).
 Attentional bias training and cue reactivity in cigarette smokers. *Addiction*, 103, 1875–1882.
- Berridge, K. C. (2009). 'Liking'and 'wanting' food rewards: Brain substrates and roles in eating disorders. *Physiology & Behavior*, *97*, 537–550.
- Bezdjian, S., Baker, L. A., Lozano, D. I., & Raine, A. (2009). Assessing inattention and impulsivity in children during the Go/NoGo task. *British Journal of Developmental Psychology*, 27, 365-383.
- Brignell, C., Griffiths, T., Bradley, B. P., & Mogg, K. (2009). Attentional and approach biases for pictorial food cues: Influence of external eating. *Appetite*, 52, 299–306.
- Burton, P., Smit, H. J., & Lightowler, H. J. (2007). The influence of restrained and external eating patterns on overeating. *Appetite*, *49*, 191–197.
- Carnell, S., Gibson, C., Benson, L., Ochner, C. N., & Geliebter, A. (2012). Neuroimaging
- and obesity: Current knowledge and future directions. *Obesity Reviews*, 13, 43-56.
- Castellanos, E. H., Charboneau, E., Dietrich, M. S., Park, S., Bradley, B. P., Mogg, K., et al. (2009). Obese adults have visual attention bias for food cue images: Evidence for altered reward system function. *International Journal of Obesity*, *33*, 1063–1073.

Cohen, J. (1992). A power primer. Psychological Bulletin, 112, 155–159.

Conway, A. R., Kane, M. J., & Engle, R. W. (2003) Working memory capacity and its relation to general intelligence. *Trends in Cognitive Science*, *7*, 547–552.

- Coskunpinar, A., & Cyders, M. A. (2013). Impulsivity and substance-related attentional bias: A meta-analytic review. *Drug and Alcohol Dependence*.
- Cousijn, J., Goudriaan, A. E., & Wiers, R. W. (2011). Reaching out towards cannabis: Approach-bias in heavy cannabis users predicts changes in cannabis use. *Addiction*, 106, 1667-1674.
- Field, M., & Cox, W. M. (2008). Attentional bias in addictive behaviors: A review of its development, causes, and consequences. *Drug and Alcohol Dependence*, 97, 1–20.
- Field, M., Duka, T., Eastwood, B., Child, R., Santarcangelo, M., & Gayton, M. (2007). Experimental manipulation of attentional biases in heavy drinkers: Do the effects generalise? *Psychopharmacology*, 195, 593–608.
- Field, M., Duka, T., Tyler, E., & Schoenmakers, T. (2009). Attentional bias modification in tobacco smokers. *Nicotine & Tobacco Research*, 11, 812-822.
- Field, M., & Eastwood, B. (2005). Experimental manipulation of attentional bias increases the motivation to drink alcohol. *Psychopharmacology*, *183*, 350–357.
- Guerrieri, R., Nederkoorn, C., & Jansen, A. (2007). How impulsiveness and variety influence food intake in a sample of healthy women. *Appetite*, *48*, 119-122.
- Guerrieri, R., Nederkoorn, C., Stankiewicz, K., Alberts, H., Geschwind, N., Martijn, C., et al. (2007). The influence of trait and induced state impulsivity on food intake in normal-weight healthy women. *Appetite*, *49*, 66–73.
- Hardman, C. A., Rogers, P. J., Etchells, K. A., Houstoun, K. V., & Munafò, M. R. (2013).The effects of food-related attentional bias training on appetite and food intake. *Appetite*, *71*, 295-300.
- Havermans, R. C., Giesen, J. C., Houben, K., & Jansen, A. (2011). Weight, gender, and snack appeal. *Eating behaviors*, *12*, 126-130.

- Hofmann, W., Friese, M., & Roefs, A. (2009). Three ways to resist temptation: The independent contributions of executive attention, inhibitory control, and affect regulation to the impulse control of eating behavior. *Journal of Experimental Social Psychology*, 45, 431-435.
- Hollitt, S., Kemps, E., Tiggemann, M., Smeets, E., &Mills, J. S. (2010). Components of attentional bias for food cues among restrained eaters. *Appetite*, *54*, 309–313.
- Hou, R., Mogg, K., Bradley, B. P., Moss-Morris, R., Peveler, R., & Roefs, A. (2011).External eating, impulsivity and attentional bias to food cues. *Appetite*, *56*, 424-427.
- Houben, K., & Jansen, A. (2011). Training inhibitory control. A recipe for resisting sweet temptations. *Appetite*, *56*, 345-349.
- Houben, K., Havermans, R. C., Nederkoorn, C., & Jansen, A. (2012). Beer à No-Go:
 Learning to stop responding to alcohol cues reduces alcohol intake via reduced affective associations rather than increased response inhibition. *Addiction*, 107, 1280-1287.
- Houben, K., Nederkoorn, C., Wiers, R. W., & Jansen, A. (2011). Resisting temptation:
 Decreasing alcohol-related affect and drinking behavior by training response
 inhibition. *Drug and Alcohol Dependence*, *116*, 132–136.
- Houben, K., & Wiers, R. W. (2009). Response inhibition moderates the relationship between implicit associations and drinking behavior. *Alcoholism: Clinical and Experimental Research*, 33, 626–633.
- Houben, K., Wiers, R. W., & Jansen, A. (2011). Getting a grip on drinking behavior training working memory to reduce alcohol abuse. *Psychological Science*, 22, 968–975.
- Kakoschke, N., Kemps, E., & Tiggemann, M. (2014). Attentional bias modification encourages healthy eating. *Eating Behaviors*, *15*, 120-124.

- Kemps, E., Tiggemann, M., & Hollitt, S. (2014). Biased attentional processing of food cues and modification in obese individuals. *Health Psychology*, 33, 1391-1401.
- Kemps, E., Tiggemann, M., Martin, R., & Elliott, M. (2013). Implicit approach–avoidance associations for craved food cues. *Journal of Experimental Psychology: Applied, 19*, 30-38.
- Kemps, E, Tiggemann, M., Orr, J. & Grear, J. (2014). Attentional re-training can reduce chocolate consumption. *Journal of Experimental Psychology: Applied*, 20, 94-102.
- Loeber, S., Grosshans, M., Korucuoglu, O., Vollmert, C., Vollstädt-Klein, S., Schneider, S.,
 ... & Kiefer, F. (2011). Impairment of inhibitory control in response to foodassociated cues and attentional bias of obese participants and normal-weight controls. *International Journal of Obesity*, *36*, 1334-1339.
- MacLeod, C., & Matthews, A. (2012). Cognitive bias modification approaches to anxiety. *Annual Review Clinical Psychology*, *8*, 189–217.
- Martin, A. A., & Davidson, T. L. (2014). Human cognitive function and the obesogenic environment. *Physiology & behavior*. Advance online publication. http://dx.doi.org/10.1016/j.physbeh.2014.02.062
- Meule, A., & Kübler, A. (2014). Double trouble. Trait food craving and impulsivity interactively predict food-cue affected behavioral inhibition. *Appetite*, *79*, 174-182.
- Meule, A., Lutz, A. P., Krawietz, V., Stützer, J., Vögele, C., & Kübler, A. (2014). Food-cue affected motor response inhibition and self-reported dieting success: a pictorial affective shifting task. *Frontiers in Psychology*, *5*, 1-17.
- Mogg, K., Bradley, B. P., Hyare, H., & Lee, S. (1998). Selective attention to food-related stimuli in hunger: Are attentional biases specific to emotional and pathological states, or are they found in normal drive states? *Behaviour Research and Therapy, 36*, 227–237.

- National Health and Medical Research Council [NHMRC] (2003). Dietary guidelines for Australian adults. Retrieved from. http://www.nhmrc.gov.au/publications/nhome. html
- Nederkoorn, C., Braet, C., Van Eijs, Y., Tanghe, A., & Jansen, A. (2006). Why obese children cannot resist food: the role of impulsivity. *Eating behaviors*, *7*, 315-322.
- Nederkoorn, C., Guerrieri, R., Havermans, R. C., Roefs, A., & Jansen, A. (2009). The interactive effect of hunger and impulsivity on food intake and purchase in a virtual supermarket. *International journal of obesity*, *33*, 905-912.
- Nederkoorn, C., Houben, K., Hofmann, W., Roefs, A., & Jansen, A. (2010). Control yourself or just eat what you like? Weight gain over a year is predicted by an interactive effect of response inhibition and implicit preference for snack foods. *Health Psychology*, 29, 389–393.
- Nederkoorn, C., Smulders, F. T., Havermans, R. C., Roefs, A., & Jansen, A. (2006). Impulsivity in obese women. *Appetite*, 47, 253-256.
- Nijs, I.M., Franken, I. H., & Muris, P. (2009). Enhanced processing of food-related pictures in female external eaters. *Appetite*, *53*, 376–383.
- Nijs, I. M., Franken, I. H., & Muris, P. (2010). Food-related Stroop interference in obese and normal-weight individuals: Behavioral and electrophysiological indices. *Eating Behaviors*, 11, 258-265.
- Nijs, I. M., Muris, P., Euser, A. S., & Franken, I. H. (2010). Differences in attention to food and food intake between overweight/obese and normal-weight females under conditions of hunger and satiety. *Appetite*, 54, 243–254.

- Price, R. B., Kuckertz, J. M., Siegle, G. J., Ladouceur, C. D., Silk, J. S., Ryan, N. D., ... & Amir, N. (November, 2014). Empirical recommendations for improving the stability of the dot-probe task in clinical research. *Psychological Assessment*. Advance Online Publication.
- Roefs, A., Werrij, M.Q., Smulders, F.T., & Jansen, A. (2006). The value of indirect measures for assessing food preferences in abnormal eating. *Journal for Consumer Protection and Food Safety*, 1, 180–186.
- Rinck, M., & Becker, E. S. (2007). Approach and avoidance in fear of spiders. *Journal of Behavior Therapy and Experimental Psychiatry*, 38, 105-120.
- Seibt, B., Häfner, M., & Deutsch, R. (2007). Prepared to eat: How immediate affective and motivational responses to food cues are influenced by food deprivation. *European Journal of Social Psychology*, *37*, 359-379.
- Sharbanee, J. M., Stritzke, W. G., Wiers, R. W., & MacLeod, C. (2013). Alcohol-related biases in selective attention and action tendency make distinct contributions to dysregulated drinking behaviour. *Addiction*, 108, 1758-1766.
- Schmukle, S. C. (2005). Unreliability of the dot probe task. *European Journal of Personality*, *19*(7), 595-605.
- Strack, F., & Deutsch, R. (2004). Reflective and impulsive determinants of social behavior. *Personality and social psychology review*, *8*, 220-247.
- Tabachnick, B. G., & Fidell, L. S. (2007). Using multivariate statistics (5th ed.). Boston,MA: Pearson/Allyn and Bacon.
- Veenstra, E. M., & de Jong, P. J. (2010). Restrained eaters show enhanced automatic approach tendencies towards food. *Appetite*, *55*, 30-36.

- Werthmann, J., Field, M., Roefs, A., Nederkoorn, C., & Jansen, A. (2014). Attention bias for chocolate increases chocolate consumption – An attention bias modification study. *Journal of Behavior Therapy and Experimental Psychiatry*, 45, 136-143.
- Werthmann, J., Roefs, A., Nederkoorn, C., Mogg, K., Bradley, B. P., & Jansen, A. (2011).Can (not) take my eyes off it: Attention bias for food in overweight participants.*Health Psychology*, *30*, 561.
- Wiers, R. W., Eberl, C., Rinck, M., Becker, E. S., & Lindenmeyer, J. (2011). Retraining automatic action tendencies changes alcoholic patients' approach bias for alcohol and improves treatment outcome. *Psychological Science*, 22, 490-497.
- Wiers, R. W., Gladwin, T. E., Hofmann, W., Salemink, E., & Ridderinkhof, K. R. (2013). Cognitive bias modification and cognitive control training in addiction and related psychopathology mechanisms, clinical perspectives, and ways forward. *Clinical Psychological Science*, 1, 192-212.
- Wiers, C. E., Kühn, S., Javadi, A. H., Korucuoglu, O., Wiers, R. W., Walter, H., ... & Bermpohl, F. (2013). Automatic approach bias towards smoking cues is present in smokers but not in ex-smokers. *Psychopharmacology*, 229, 187-197.
- Wiers, R. W., Rinck, M., Dictus, M., & Van Den Wildenberg, E. (2009). Relatively strong automatic appetitive action-tendencies in male carriers of the OPRM1 Gallele. *Genes, Brain and Behavior*, 8, 101-106.
- Wiers, R. W., Rinck, M., Kordts, R., Houben, K., & Strack, F. (2010). Retraining automatic action-tendencies to approach alcohol in hazardous drinkers. *Addiction*, 105, 279-287.
- World Health Organization [WHO] (2013).World Health Statistics 2013. Retrieved from. http://www.who.int

CHAPTER 3: STUDY 2

Implicit Food Evaluation and Emotional Eating Interact to Predict Unhealthy Food Choice

Naomi Kakoschke, Eva Kemps, and Marika Tiggemann School of Psychology, Flinders University, South Australia

Corresponding author: Naomi Kakoschke, School of Psychology, Flinders University, GPO Box 2100, Adelaide, 5001, South Australia. Email: naomi.kakoschke@flinders.edu.au **Funding:** This research was supported under the Australian Research Council's Discovery Project funding scheme (project number DP130103092).

Statement of co-authorship: All authors were involved in the formulation of the study concept and design. Naomi Kakoschke collected the data, and completed the data analysis and the initial draft of the manuscript. Eva Kemps and Marika Tiggemann edited multiple revisions of the manuscript.

Acknowledgements: We are grateful to Paul Douglas for developing the software for the computer administered tasks.

This manuscript is currently under review.

Abstract

The current study aimed to examine implicit evaluations of unhealthy food and an emotional eating style together in predicting unhealthy food choice. Participants were 80 undergraduate women aged 17 to 25 years who completed a self-report measure of emotional eating (the Dutch Eating Behaviour Questionnaire), a Single-Category Implicit Association Task, and a food choice task. Results indicated that a more positive implicit evaluation of unhealthy food predicted both higher trait emotional eating and unhealthy snack food choice. In addition, there was an interaction between implicit unhealthy food evaluation and emotional eating in predicting eating behaviour, such that the relationship between a more positive implicit food evaluation and a greater number of unhealthy snack foods chosen was stronger for those lower, rather than higher, in trait emotional eating. It was concluded that both implicit food evaluation and emotional eating play a role in determining unhealthy eating behaviour.

1. Introduction

The present study focused on the relationship between implicit evaluation of food and unhealthy eating behaviour. In particular, we were interested in the moderating role of a trait emotional eating style on this relationship. Previous research has found that a more positive implicit evaluation of unhealthy food predicts unhealthy eating behaviour, such as increased intake and snack choice. Likewise, higher trait emotional eating has been related to unhealthy eating behaviour. However, few studies have examined implicit food evaluations and emotional eating together. The present study aimed to extend the literature by examining both the additive and interactive effects of implicit food evaluation and emotional eating in predicting unhealthy eating behaviour.

Unhealthy eating behaviour, such as consuming too much food containing a high fat, salt and sugar content, has been linked to weight gain, as well as consequent overweight and obesity (WHO, 2014). One factor that has contributed to the increasing rates of overweight and obesity over the last few decades is an "obesogenic" environment, in which high-calorie food is easily accessible and readily available (Hill, Wyatt, Reed, & Peters, 2003). It has been proposed that the automatic processing of environmental cues associated with such food may result in unhealthy eating behaviour (Cohen & Babey, 2012; Kemps, Tiggemann & Hollitt, 2014; Marteau et al., 2012). One automatic mechanism that has been shown to contribute to such behaviour is an implicit evaluation of food, which refers to an association between food stimuli and positive affect at an automatic level (Czyzewska & Graham, 2008). It is important to examine implicit evaluations of unhealthy food given the role of processing of such cues in predicting unhealthy food intake (Hofmann, Gschwendner, Friese, Wiers, & Schmitt, 2008).

The suggestion that an affective reaction to unhealthy food cues may influence eating behaviour corresponds to a key theoretical prediction of contemporary dual-process models. Specifically, these models propose that both automatic and controlled processes play an important role in determining behaviour (Strack & Deutsch, 2004). On the one hand, the automatic system involves processing that is fast, effortless, and associative. This system guides behaviour based on an automatic appraisal of the affective properties of a stimulus and is influenced by emotion (Strack & Deutsch, 2004). On the other hand, the reflective system involves processing that is slow, controlled, and conscious. This system guides behaviour based on long-term goals and personal standards (Strack & Deutsch, 2004). Thus, it may be that automatically associating unhealthy food with positive affect compromises conscious reflection on long-term goals such as weight loss, resulting in unhealthy eating behaviour.

Implicit evaluations have commonly been measured using tasks such as The Implicit Association Test (IAT; Greenwald, McGhee, & Schwartz, 1998; Friese, Hofmann, & Wanke, 2008). In recent years, a number of studies using the IAT have shown that a more positive, implicit evaluation of unhealthy food is associated with increased choice of unhealthy food over healthy food, as well as higher self-reported and laboratory-based consumption of unhealthy food, and higher BMI (Conner, Perugini, O'Gorman, Ayres, & Prestwich, 2007; Friese et al., 2008; Haynes, Kemps, Moffitt, & Mohr, 2015a; Hofmann & Friese, 2008; Hofmann et al., 2008; Hofmann, Rauch, & Gawronski, 2007; Perugini, 2005; Richetin, Perugini, Prestwich, & O'Gorman, 2007). Thus, a large body of evidence suggests that a more positive implicit evaluation of unhealthy food predicts unhealthy eating behaviour.

Another factor that has been shown to influence unhealthy eating behaviour is a trait eating style, in particular, emotional eating. Emotional eating refers to the tendency to overeat in response to emotional cues, such as negative affect (e.g., distress or boredom; van Strien, Schippers, & Cox, 1995; Macht & Simons, 2000). Rather than a general tendency to eat more, emotional eating is triggered by a negative emotional state and is typically assessed with a subscale of the Dutch Eating Behavior Questionnaire (DEBQ; van Strien, Frijters, Bergers, & Defares 1986). Studies have found that self-identified female emotional eaters reported greater consumption of unhealthy snack food in response to a negative emotional state than low emotional eaters (Elfhag, Tholin, & Rasmussen, 2008; de Lauzon et al., 2004; Newman, O'Connor, & Conner, 2007; Wallis & Hetherington, 2009). Evidence also suggests that emotional eating may increase the risk of overweight and obesity as individuals who eat to cope with negative emotions are more likely to choose high-calorie foods (Macht, 1999; Bennett, Greene, & Schwartz-Barcot, 2013), which have been shown to reduce stress (Mercer & Holder, 1997). However, consuming too much high-calorie food can result in an energy imbalance (Fay & Finlayson, 2011) and as such, trait emotional eating has been associated with a higher BMI (Verhoeven et al., 2015) and weight gain (Greene et al., 2011).

However, not all studies have documented increased unhealthy eating behaviour in emotional eaters. Some correlational research has found that emotional eating was unrelated to unhealthy snack food consumption (Adriaanse, de Ridder, & Evers, 2011; Anschutz, van Strien, van de Ven, & Engels, 2009; Lluch, Herbeth, Mejean, & Siest, 2000; Wardle et al., 1992). Studies have also shown that emotional eating did not predict being overweight in a sample of adolescent girls (Snoek, van Strien, Janssens, & Engels, 2007) or weight gain in healthy weight undergraduate students (Lowe, Annunziato, & Markowitz, 2006). In addition, some experimental studies found no difference between high and low emotional eaters in unhealthy snack food intake after a sad versus neutral mood induction (Evers, de Ridder, & Adriannse, 2009; Werthmann et al., 2014). The mixed evidence suggests that it is important to determine the conditions under which emotions may predict unhealthy eating behaviour.

While both implicit food evaluation and emotional eating have been shown to predict eating behaviour independently, few researchers have examined both of these factors within the same study. Two relevant studies conducted in the laboratory investigated whether more positive implicit food evaluations guide eating behaviour under different emotional states. In these studies, emotional eating was conceptualised as eating in response to an experimentally manipulated mood state. Friese, Hofmann and Schmidt (2008) instructed participants to either suppress their emotions or let them flow while watching an emotional film segment. They found that positive implicit evaluations (of chips) predicted increased consumption of chips following exposure to the film, but only for the participants who were instructed to suppress their emotions. More recently, Holland, de Vries, Hermsen, and Knippenberg (2012) manipulated mood by using both happy and sad film segments. They found that implicit food evaluation predicted eating behaviour for individuals who were in an induced positive mood, but not for those in a negative mood. Specifically, implicit food evaluations were predictive of unhealthy food choice (a candy bar as opposed to an apple) when people were in a happy, but not a sad mood. In summary, implicit evaluation was less predictive of eating behaviour under negative mood states.

Thus far, only one study has examined the role of individual differences in trait emotional eating in the link between implicit food evaluation and eating behaviour (Ayres, Prestwich, Connor, & Smith, 2011). Specifically, Ayres et al. found an association between a more positive implicit food evaluation (i.e., increased liking of chocolate) and self-reported daily chocolate intake over the following week. They also found that emotional eating scores were positively correlated with implicit liking of chocolate, but not with self-reported intake. Importantly, Ayres et al. found that implicit evaluations and trait emotional eating interacted in predicting eating behaviour, such that liking of chocolate was more predictive of habitual chocolate intake in emotional eaters. As the emotional eating subscale indicates a tendency to eat in response to negative affect, this result contrasts with the findings of the two previous laboratory studies, which showed that implicit food evaluations predicted eating behaviour in individuals who suppressed their emotions (Friese et al., 2008) or were in a positive mood state (Holland et al., 2012). However, it should be noted that Ayres et al. measured selfreported consumption, unlike in the previous two experimental studies. Indeed, the authors suggested that it is important to confirm their findings using a behavioural measure of eating, such as food choice.

The aim of the current study was to further investigate the role of implicit unhealthy food evaluation and emotional eating together on unhealthy eating behaviour. Specifically, it was predicted that implicit evaluation of food and emotional eating would interact to predict unhealthy food choice. This hypothesis was based on the idea that associative processing is influenced by emotion, as well as evidence which suggests that increased automatic eating behaviour occurs in emotional eaters. Following Ayres et al.'s (2011) suggestion, emotional eating was conceptualized as a trait and a food choice task was used to provide a behavioural measure of unhealthy eating. Such a task allows insight into decision-making related to food (Jasinska et al., 2012).

2. Method

2.1 Participants

Participants were 80 women recruited from the undergraduate student population at Flinders University. They were aged 17 to 27 years (M = 20.28, SD = 2.31) and had a mean body mass index of 22.86 kg/m² (SD = 4.52), which is classified as being within the healthy weight range (i.e., 18.5–24.9 kg/m²). Only women were recruited, as they have shown higher levels of emotional eating than men (Keller & van der Horst, 2013) and have a greater tendency to overeat (Burton, Smit, & Lightowler, 2007). Participants were recruited if they spoke English as their first language, liked most foods, and did not have any food allergies, intolerances, or dietary requirements.

2.2 Procedure

Participants were recruited for a study entitled 'Food Preferences and Eating Habits in Women' through flyers posted around campus, and the School of Psychology's online study participation system. The testing session lasted approximately 30 minutes and was conducted in a quiet laboratory. After giving informed consent, participants provided some background information, and completed the Implicit Association Task, followed by the food choice task. As a trait measure, the emotional eating subscale of the DEBQ was administered last to ensure that food choice was not influenced by its completion. The study was approved by the University's Social and Behavioural Research Ethics Committee.

2.3 Measures

2.3.1 Emotional eating. Emotional eating was measured by the Emotional Eating subscale of the Dutch Eating Behaviour Questionnaire (DEBQ; van Strien et al., 1986). This subscale contains 13 items about eating in response to negative emotions (e.g., 'Do you have a desire to eat when you are emotionally upset?'). Participants were asked to indicate how well each of the items related to them on a five-point scale ranging from 'never' (= 1) to 'very often' (= 5). Responses for each item were averaged to produce an emotional eating score, with higher scores reflecting higher levels of emotional eating. Internal reliability was high in the present sample ($\alpha = .96$), comparable with a previous sample ($\alpha = .95$; Ayres et al., 2011, Study 2).

2.3.2 Implicit unhealthy food evaluation. Implicit evaluation of unhealthy food was measured using a single-category implicit association task (SC-IAT; Karpinski & Steinman, 2006). The target category was unhealthy food and the evaluative categories were 'I like' and 'I dislike' (Olson & Fazio, 2004). Participants were asked to categorise pictures (presented one at a time in the centre of the computer screen) into one of three categories: unhealthy food (target), positive (evaluative), or negative (evaluative). The target stimuli were six pictures of unhealthy food (i.e., cake, chocolate, ice-cream, chips, hamburger, and pizza). Evaluative stimuli were six positive and six negative pictures (unrelated to food) selected

from the IAPS¹ (Lang, Bradley & Cuthbert, 2001). These pictures have been used in previous studies (e.g., Nederkoorn et al., 2010).

The SC-IAT consisted of three phases. In the first phase, participants practiced classifying the positive and negative pictures into the evaluative categories of "I like" and "I dislike" using two response keys (i.e., left = 'E' and right = 'I'). This phase consisted of 24 trials with the six positive and the six negative pictures each presented twice. In the second practice phase, participants were asked to categorise the pictures into both of the evaluative categories (positive and negative) as well as the target category (unhealthy food). This phase consisted of 36 trials with the six positive, six negative and six unhealthy food pictures each presented twice. The unhealthy food pictures shared a response key with positive pictures in half of the trials and with negative pictures in the other half. The key side assigned to the response to the target category was counterbalanced across participants. The third phase was the same as the second, but the number of trials increased to 144 with each positive, negative, and food picture presented eight times. Participants were given a short break at half way.

IAT scores were calculated using the D600-algorithm (Greenwald, Nosek, & Banaji, 2003), modified for application to SC-IATs (Karpinski & Steinman, 2006). Mean response times of food paired with positive trials were subtracted from mean response times of food paired with negative trials. The difference between mean response times on these two trial types was divided by the standard deviation of all correct response times on food trials, with higher scores indicative of a more positive implicit evaluation of unhealthy food. Incorrect responses were discarded, following Karpinski and Steinman (2006). Response times below 300ms and above 3000ms were also removed as these times are regarded as anticipatory or delayed, respectively (Palfai & Ostafin, 2003).

¹ IAPS picture numbers: 1300, 1603, 2070, 2550, 5480, 5623, 6550, 6570, 8200, 9220, 9340, 9600

2.3.3 Food choice. Following Veling, Aarts, and Stroebe (2013), we used a touchscreen food-choice task to assess participants' hypothetical selection of foods that they would like to obtain to take home and consume. The food pictures comprised eight healthy (i.e., carrots, bananas, strawberries, almonds, crispbread, avocado, apple, fruit salad) and eight unhealthy (i.e., cheese, M&Ms, cookie, corn chips, chocolate, muffin, potato crisps, pretzels) snacks, which were arrayed in a 4 x 4 square grid on a computer touch-screen. Participants were asked to select eight of these 16 foods by touching the picture of the food on the screen. They were given a time limit of 15 seconds to make their selections as food choices are often made under time pressure. A touch-screen was used to represent reaching out to select a food, similar to using buttons to select a snack from a vending machine. The main dependent variable was the number (out of 8) of unhealthy snack foods selected.

3. Results

3.1 Descriptive Statistics

Participants' implicit unhealthy food evaluation scores ranged from -1.07 to 1.04 (M = -.04, SD = .46), which was comparable to Ayres et al.'s (2011; Study 2) sample (M = -.17, SD = .39). The mean score for emotional eating (M = 2.89, SD = 1.05) was also similar to Ayres et al.'s sample (M = 2.72, SD = .80). For the food choice task, unhealthy snack food choices ranged from 1 to 7 (M = 3.54, SD = 1.66).

3.2 Relationships between implicit evaluation, emotional eating, and food choice

A series of correlational analyses was conducted to examine the inter-relationships between implicit positive food evaluation, emotional eating and unhealthy food choice. There was a significant positive correlation between implicit food evaluation and both emotional eating, r = .25, p = .03, and unhealthy food choice, r = .24, p = .04. The positive correlation between emotional eating and unhealthy food choice was not significant, r = .16, p = .15.

3.3 Combined effect of implicit evaluation and emotional eating on food choice

A moderated regression analysis was conducted to examine whether implicit food evaluation and an emotional eating style interacted in predicting unhealthy snack food choice. Centered implicit food evaluation and emotional eating scores were entered in Step 1. The product term was entered in Step 2. In Step 1, unhealthy food choice was not predicted by either implicit unhealthy food evaluation, B = .689, p = .094, or emotional eating, B = .212, p= .238, F (2, 71) = 2.83, p = .07, $R^2 = .074$. In Step 2, however, the product term (implicit food evaluation x emotional eating) explained significant additional variance in unhealthy snack food choice, B = -1.04, p = .015, $F_{Change} (1, 70) = 6.16$, p = .015, $R^2_{Change} = .075$.

The above analysis indicates that implicit food evaluation and an emotional eating style interact in predicting unhealthy snack food choice. Simple slopes analyses were conducted to determine the form of this significant interaction. As can be seen in Fig. 1, implicit food evaluation was positively associated with unhealthy food choice in participants lower in emotional eating (one SD below the mean), B = .792, t(70) = 2.01, p = .048, but was unrelated to food choice in participants higher in emotional eating (one SD above the mean), B = .308, t(70) = .090, p = .466.



Fig 1. Unhealthy food choice as a function of implicit unhealthy food evaluation and trait emotional eating.

4. Discussion

The current study aimed to examine the potential interactive effects of implicit food evaluation and an emotional eating style on unhealthy snack food choice. As expected, a more positive implicit evaluation of unhealthy food was associated with higher emotional eating and increased unhealthy snack food choice. In contrast, emotional eating was not correlated with unhealthy food choice. Most importantly, implicit evaluations of unhealthy food and emotional eating interacted in predicting unhealthy food choice, such that positive implicit food evaluation was more predictive of snack food choice in those with lower emotional eating scores.

Not surprisingly, an implicit positive evaluation (i.e., liking) of unhealthy food was positively correlated with unhealthy snack food choice. This finding shows that individuals who implicitly like unhealthy snack food more have a greater tendency to choose such foods when given the option of choosing from an array of healthy and unhealthy snack foods. This finding is consistent with previous research (Ayres et al., 2011; Conner et al., 2007; Dube, 2007; Friese et al., 2008; Hofmann et al., 2008; Perugini, 2005; Prestwich et al., 2011; Richetin et al., 2007) and supports contemporary dual-process models, which posit that behaviour is partly determined by automatic processing (Strack & Deutsch, 2004). Clearly, appetitive food cues elicit automatic affective reactions for some individuals.

In addition, higher emotional eating was associated with greater implicit liking of unhealthy food. This finding supports the results of Ayres et al. (2011), who likewise found that higher emotional eating was associated with greater implicit liking of chocolate, and extends it to implicit evaluations of unhealthy food more generally. Together, the findings suggest that the tendency to eat for emotional reasons is related to a greater implicit liking of unhealthy food.

We did not find a significant association between emotional eating and unhealthy

snack food choice. This finding is not surprising given the previous inconsistent results, with some studies showing that an emotional eating style predicts increased unhealthy eating behaviour (e.g., Elfhag, et al., 2008; de Lauzon et al., 2004; Newman, et al., 2007; Wallis & Hetherington, 2009), and other studies finding no such relationship (Adriaanse, et al., 2011; Anschutz, et al., 2009; Lluch, et al., 2000; Wardle et al., 1992). In particular, Ayres et al. (2011) found no relationship between emotional eating and self-reported chocolate intake. Thus, an emotional eating style on its own may not predict increased unhealthy food choice or intake.

The main contribution of the present study, however, was to investigate the combined effect of implicit evaluation of unhealthy food and emotional eating on unhealthy snack food choice. Specifically, our interaction finding shows that implicit food evaluation was more predictive of unhealthy eating behaviour in individuals with lower, rather than higher, trait emotional eating. This finding contrasts with that of Ayres et al. (2011), who showed the opposite pattern of results also using the emotional eating subscale. However, the current finding aligns better with the results of the two laboratory-based studies, which showed that individuals who supressed their emotions ate less food (Friese et al., 2008) and those in an induced negative mood were less likely to choose an unhealthy snack (Holland et al., 2012).

Thus, it remains to be determined as to why the present result differs from that of Ayres et al. (2011). One possible explanation lies in the nature of the measure used to assess eating behaviour. Specifically, both the current study and the two previous experimental studies (Friese et al., 2008; Holland et al., 2012) used a behavioural measure collected in the laboratory, while Ayres et al. used a self-report measure of food intake which may be less reliable (e.g., Giuliani, Tomiyama, Mann, & Berkman, 2015). Another possibility is that Ayres et al. investigated responses to chocolate, rather than unhealthy food in general, as in the present study. Rogers and Smit (2000) have argued that chocolate has a special place

among foods in contemporary Western society because it is both highly palatable and can even be addictive, while generating ambivalence (i.e., 'naughty but nice'). Chocolate is also valued as a reward and is associated with gift-giving (Hetherington & MacDiarmid, 1993), and thus may function differently from unhealthy foods in general. Future research might investigate both chocolate and unhealthy food more generally in the same study protocol.

It is also important to note that questions have recently been raised about the nature of the emotional eating subscale. Specifically, the scale assesses whether people tend to eat in response to negative emotions. However, individuals can also overeat in response to positive emotions (Bongers, Jansen, Havermans, Roefs, & Nederkoorn, 2013). Indeed, Holland et al. (2012) showed that those in a happy mood state were more likely to choose an unhealthy snack than people who were in a sad mood state. As a set, the studies support the importance of determining how different emotions might influence the eating behaviour of self-identified emotional eaters.

The present study has some important practical implications. The finding that greater implicit liking of unhealthy food predicts increased unhealthy snack food choice provides support for interventions that target implicit food evaluation. Several tasks can be used to retrain implicit evaluations, such as evaluative conditioning protocols or a modified version of the IAT. Both of these tasks involve consistently pairing unhealthy food stimuli with negative affective cues (Ebert, Steffens, von Stulpnagel, & Jelenec, 2009; Hollands, Prestwich, & Marteau, 2011). Previous studies have shown that these tasks can reduce implicit liking of unhealthy snack food (Ebert et al., 2009; Haynes, Kemps, & Moffitt, 2015b; Hollands et al., 2011; Lebens et al., 2011), as well as subsequent unhealthy snack food intake (Haynes et al., 2015b) and choice (Hollands et al., 2011). However, some studies have found no effect on subsequent food choice (Eberl, et al., 2009; Lebens et al., 2011). The present results suggest that one reason for the mixed findings may be due to the trait level of emotional eating in the particular samples. Indeed, Hensels and Baines (2016) recently showed that an evaluative conditioning task was effective at changing implicit food evaluations and snack choice for participants with lower, but not those with higher trait emotional eating scores. Thus, future studies could usefully consider the role of emotional eating in protocols designed to modify implicit food evaluations.

In conclusion, the main finding of the present study is that a more positive implicit food evaluation predicts increased unhealthy snack food choice for individuals lower, rather than higher, in emotional eating. The findings contribute to our theoretical understanding of how both implicit food evaluation and trait eating styles can influence unhealthy eating behaviour.

References

- Adriaanse, M. A., De Ridder, D. T. D., & Evers, C. (2011). Emotional eating: Eating when emotional or emotional about eating? *Psychology & Health*, *26*, 23-39.
- Anschutz, D. J., van Strien, T., Van De Ven, M. O., & Engels, R. C. (2009). Eating styles and energy intake in young women. *Appetite*, *53*, 119–22.
- Ayres, K., Prestwich, A., Conner, M. T., & Smith, P. (2011). Emotional eating moderates the relationship between implicit measures of attitudes and chocolate consumption. *European Journal of Personality*, 25, 317-325.
- Bennett, J., Greene, G., & Schwartz-Barcott, D. (2013). Perceptions of emotional eating behavior. A qualitative study of college students. *Appetite*, 60, 187-192.
- Bongers, P., Jansen, A., Havermans, R., Roefs, A., & Nederkoorn, C. (2013). Happy eating. The underestimated role of overeating in a positive mood. *Appetite*, *67*, 74–80.
- Burton, P., Smit, H. J., & Lightowler, H. J. (2007). The influence of restrained and external eating patterns on overeating. *Appetite*, *49*, 191–197.
- Cohen, D.A., & Farley, T.A. (2008). Eating as an automatic behavior. *Preventing Chronic Disease*, *5*, 1-7.
- Conner, M. T., Perugini, M., O'Gorman, R., Ayres, K., & Prestwich, A. (2007). Relations between implicit and explicit measures of attitudes and measures of behavior:
 Evidence of moderation by individual difference variables. *Personality and Social Psychology Bulletin*, *33*, 1727-1740.
- Cox, W. M., Klinger, E., & Fadardi, J. S. (2015). The motivational basis of cognitive determinants of addictive behaviors. *Addictive Behaviors*, 44, 16-22.
- Czyzewska, M., & Graham, R. (2008). Implicit and explicit attitudes to high-and low-calorie food in females with different BMI status. *Eating behaviors*, *9*, 303-312.

- de Lauzon, B., Romon, M., Deschamps, V., Lafay, L., Borys, J. M., Karlsson, J., ... & Charles, M. A. (2004). The Three-Factor Eating Questionnaire-R18 is able to distinguish among different eating patterns in a general population. *The Journal of Nutrition*, 134, 2372-2380.
- Elfhag, K., Tholin, S., & Rasmussen, F. (2008). Consumption of fruit, vegetables, sweets and soft drinks are associated with psychological dimensions of eating behaviour in parents and their 12-year-old children. *Public Health Nutrition*, *11*, 914-923.
- Evers, C., de Ridder, D. T., & Adriaanse, M. A. (2009). Assessing yourself as an emotional eater: Mission impossible?. *Health Psychology*, 28, 717-725.
- Fay, S. H., & Finlayson, G. (2011). Negative affect-induced food intake in non-dieting women is reward driven and associated with restrained–disinhibited eating subtype. *Appetite*, 56, 682-688.
- Friese, M., Hofmann, W., & Wänke, M. (2008). When impulses take over: Moderated predictive validity of explicit and implicit attitude measures in predicting food choice and consumption behaviour. *British Journal of Social Psychology*, 47, 397-419.
- Giuliani, N. R., Tomiyama, A. J., Mann, T., & Berkman, E. T. (2015). Prediction of daily food intake as a function of measurement modality and restriction status. *Psychosomatic Medicine*, 77, 583–590.
- Greene, G. W., Schembre, S. M., White, A. A., Hoerr, S. L., Lohse, B., et al. (2011).
 Identifying clusters of college students at elevated health risk based on eating and exercise behaviors and psychosocial determinants of body weight. *Journal American Dietetic Association*, 111, 394–400.
- Greenwald, A. G., McGhee, D. E., & Schwartz, J. L. (1998). Measuring individual differences in implicit cognition: the implicit association test. *Journal of Personality* and Social Psychology, 74, 1464-1480.

- Greenwald, A. G., Nosek, B. A., & Banaji, M. R. (2003). Understanding and using the implicit association test: I. An improved scoring algorithm. *Journal of Personality* and Social Psychology, 85, 197-216.
- Haynes, A., Kemps, E. and Moffitt, R. (2015). Inhibitory self-control moderates the effect of changed implicit food evaluations on snack food consumption. *Appetite*, 90, 114-122.
- Haynes, A., Kemps, E., Moffitt, R. and Mohr, P. (2015). Reduce temptation or resist it?
 Experienced temptation mediates the relationship between implicit evaluations of unhealthy snack foods and subsequent intake. Psychology and Health, 30, 534-550.
- Hensels, I. S., & Baines, S. (2016). Changing 'gut feelings' about food: An evaluative conditioning effect on implicit food evaluations and food choice. *Learning and Motivation*, 55, 31-44.
- Hetherington, M. M., & MacDiarmid, J. I. (1993). "Chocolate addiction": A preliminary study of its description and its relationship to problem eating. *Appetite*, *21*, 233-246.
- Hill, J. O., Wyatt, H. R., Reed, G. W., & Peters, J. C. (2003). Obesity and the environment: where do we go from here? *Science*, *299*, 853-855.
- Hofmann, W., & Friese, M. (2008). Impulses got the better of me: Alcohol moderates the influence of implicit attitudes toward food cues on eating behavior. *Journal of Abnormal Psychology*, 117, 420-427.
- Hofmann, W., Gschwendner, T., Friese, M., Wiers, R. W., & Schmitt, M. (2008). Working memory capacity and self-regulatory behavior: Toward an individual differences perspective on behavior determination by automatic versus controlled processes. *Journal of Personality and Social Psychology*, 95, 962-977.

- Hofmann, W., Rauch, W., & Gawronski, B. (2007). And deplete us not into temptation:
 Automatic attitudes, dietary restraint, and self-regulatory resources as determinants of eating behavior. *Journal of Experimental Social Psychology*, 43, 497-504.
- Holland, R. W., de Vries, M., Hermsen, B., & van Knippenberg, A. (2012). Mood and the attitude–behavior link: The happy act on impulse, the sad think twice. *Social Psychological and Personality Science*, *3*, 356-364.
- Hollands, G. J., Prestwich, A., & Marteau, T. M. (2011). Using aversive images to enhance healthy food choices and implicit attitudes: An experimental test of evaluative conditioning. *Health Psychology*, *30*, 195-203.
- Jasinska, A. J., Yasuda, M., Burant, C. F., Gregor, N., Khatri, S., Sweet, M., & Falk, E. B. (2012). Impulsivity and inhibitory control deficits are associated with unhealthy eating in young adults. *Appetite*, 59, 738-747.
- Karpinski, A., & Steinman, R. B. (2006). The single category implicit association test as a measure of implicit social cognition. *Journal of Personality and Social Psychology*, 91, 16-32.
- Keller, C., & van der Horst, K. (2013). Dietary restraint, ambivalence toward eating, and the valence and content of spontaneous associations with eating. *Appetite*, *62*, 150-159.
- Lang, P., Bradley, M. M., & Cuthbert, B. N. (2001). International affective picture system (IAPS): Instruction manual and affective ratings. (Tech. Rep. No. A-5.) The Center for Research in Psychophysiology, University of Florida.
- Lebens, H., Roefs, A., Martijn, C., Houben, K., Nederkoorn, C., & Jansen, A. (2011).Making implicit measures of associations with snack foods more negative through evaluative conditioning. *Eating Behaviors, 12*, 249-253.
- Lluch, A., Herbeth, B., Mejean, L., & Siest, G. (2000). Dietary intakes, eating style and overweight in the Stanislas Family Study. *International Journal of Obesity*, 24, 1493-1499.
- Lowe, M. R., Annunziato, R. A., Markowitz, J. T. et al. (2006). Multiple types of dieting prospectively predict weight gain during the freshman year of college. *Appetite*, *47*, 83–90.
- Macht, M. (1999). Characteristics of eating in anger, fear, sadness and joy. *Appetite*, *33*, 129-139.
- Macht, M., & Simons, G. (2000). Emotions and eating in everyday life. *Appetite*, *35*, 65–71.
- Nederkoorn, C., Houben, K., Hofmann, W., Roefs, A., & Jansen, A. (2010). Control yourself or just eat what you like? Weight gain over a year is predicted by an interactive effect of response inhibition and implicit preference for snack foods. *Health Psychology*, 29, 389–393.
- Newman, E., O'Connor, D. B., & Conner, M. (2007). Daily hassles and eating behaviour: the role of cortisol reactivity status. *Psychoneuroendocrinology*, *32*, 125-132.
- Oliver, G., Wardle, J., & Gibson, E. L. (2000). Stress and food choice: A laboratory study. *Psychosomatic Medicine*, *62*, 853-865.
- Olson, M. A., & Fazio, R. H. (2004). Reducing the influence of extrapersonal associations on the implicit association test: Personalizing the IAT. *Journal of Personality and Social Psychology*, 86, 653–667.
- Palfai, T. P., & Ostafin, B. D. (2003). Alcohol-related motivational tendencies in hazardous drinkers: Assessing implicit response tendencies using the modified-IAT. *Behaviour Research and Therapy*, 41, 1149-1162.

- Perugini, M. (2005). Predictive models of implicit and explicit attitudes. *British Journal of Social Psychology*, 44, 29-45.
- Richetin, J., Perugini, M., Prestwich, A., & O'Gorman, R. (2007). The IAT as a predictor of food choice: The case of fruits versus snacks. *International Journal of Psychology*, 42, 166-173.
- Rogers, P. J., & Smit, H. J. (2000). Food craving and food "addiction": A critical review of the evidence from a biopsychosocial perspective. *Pharmacology Biochemistry and Behavior*, 66, 3-14.
- Snoek, H. M., Van Strien, T., Janssens, J. M., & Engels, R. C. (2007). Emotional, external, restrained eating and overweight in Dutch adolescents. *Scandinavian Journal of Psychology*, 48, 23-32.
- Strack, F., & Deutsch, R. (2004). Reflective and impulsive determinants of social behavior. *Personality and Social Psychology Review*, 8, 220-247.
- van Strien, T., Frijters, J. E., Bergers, G. P., & Defares, P. B. (1986). The Dutch Eating Behavior Questionnaire (DEBQ) for assessment of restrained, emotional, and external eating behavior. *International Journal Eating Disorders*, *5*, 295–315
- van Strien, T., Schippers, G. M., & Cox, W. M. (1995). On the relationship between emotional and external eating behavior. *Addictive Behaviors*, 20, 585-594.
- Veling, H., Aarts, H., & Stroebe, W. (2013). Using stop signals to reduce impulsive choices for palatable unhealthy foods. *British Journal of Health Psychology*, *18*, 354-368.
- Verhoeven, A. A., Adriaanse, M. A., de Vet, E., Fennis, B. M., & de Ridder, D. T. (2015).It's my party and I eat if I want to. Reasons for unhealthy snacking. *Appetite*, 84, 20-27.
- Wallis, D. J., & Hetherington, M. M. (2009). Emotions and eating. Self-reported and experimentally induced changes in food intake under stress. *Appetite*, *52*, 355-362.

- Wardle, J., Marsland, L., Sheikh, Y., Quinn, M., Fedoroff, I., & Ogden, J. (1992). Eating style and eating behaviour in adolescents. *Appetite*, *18*, 167-183.
- Werthmann, J., Renner, F., Roefs, A., Huibers, M. J., Plumanns, L., Krott, N., & Jansen, A. (2014). Looking at food in sad mood: Do attention biases lead emotional eaters into overeating after a negative mood induction? *Eating Behaviors*, 15, 230-236.
- World Health Organization [WHO]. (2014). Obesity and overweight. Retrieved October 1, 2015 from http://www.who.int/mediacentre/factsheets/fs311/en/.

CHAPTER 4: STUDY 3

Differential Effects of Approach Bias and Eating Style on Unhealthy Food Consumption in Overweight and Normal Weight Women

Naomi Kakoschke, Eva Kemps, and Marika Tiggemann School of Psychology, Flinders University, South Australia

Corresponding author: Naomi Kakoschke, School of Psychology, Flinders University, GPO Box 2100, Adelaide, 5001, South Australia. Email: naomi.kakoschke@flinders.edu.au **Funding:** This research was supported under the Australian Research Council's Discovery Project funding scheme (project number DP130103092).

Statement of co-authorship: All authors were involved in the formulation of the study concept and design. Naomi Kakoschke collected the data, and completed the data analysis and the initial draft of the manuscript. Eva Kemps and Marika Tiggemann edited multiple revisions of the manuscript.

Acknowledgements: We are grateful to Paul Douglas for developing the software for the computer administered Approach-Avoidance task.

Abstract

Objective: The current study aimed to examine the effects of approach bias for unhealthy food and trait eating style on consumption of unhealthy food in overweight and normal weight individuals. **Method:** Participants were 245 undergraduate women aged 17 to 26 years. They completed an Approach-Avoidance Task, the Dutch Eating Behaviour Questionnaire (to assess restrained, emotional, and external eating), and a taste test to measure consumption of unhealthy food. **Results:** An external eating style predicted increased consumption of unhealthy food. Among overweight participants, external and emotional eating style individually moderated the relationship between approach bias for unhealthy food and subsequent consumption. Specifically, approach bias was positively related to consumption in high external and emotional eaters, but negatively related to consumption in low external and emotional eaters. These interactions were not observed among normal weight participants. **Conclusion:** Practically, the results suggest that overweight individuals who are external or emotional eaters may benefit from interventions that aim to modify approach bias toward unhealthy food cues to reduce problematic eating behaviour.

Introduction

The prevalence of overweight and obesity has more than doubled during the last few decades in contemporary Western societies. Worldwide, 39% of adults are now classified as overweight and 13% as obese (WHO, 2014). Individuals in the unhealthy weight range (body mass index [BMI] >25 kg/m²) are at an increased risk of developing chronic health conditions, such as cardiovascular disease, diabetes, and cancer (WHO, 2014). Excess weight gain is partly driven by unhealthy eating behaviour, such as overeating or a tendency to consume foods high in fat, salt, and/or sugar (Hill, Wyatt, Reed, & Peters, 2003). One of the contributors to unhealthy eating behaviour is a continual exposure to an abundance of appetitive, high-caloric food cues through advertising on the internet, television, billboards, and in magazines (Brunner, van der Horst, Siegrist, 2010). Exposure to appetitive food cues in this 'obesogenic' environment is thought to contribute to unhealthy consumption behaviour through biased cognitive processing of such cues (Marteau, Hollands, & Fletcher, 2012). However, not all individuals overeat and gain weight in such an environment, which suggests that individual differences may play a role in the responsiveness to appetitive food and eating-related cues (Carnell, Kym, & Pryor, 2012).

Contemporary dual-process models suggest that biased cognitive processes are crucial for understanding health behaviour (Strack & Deutsch, 2004). According to such models, behaviour is determined by two cognitive processing systems: automatic processing, which is fast, implicit, and effortless, and controlled processing, which is slow, explicit, and effortful. Dual process models posit that automatic processing generally guides our behaviour unless the controlled system is able to regulate such processing. Furthermore, it has been proposed that the automatic processing system may be driven by cognitive biases, such as an approach bias (the tendency to reach out toward rather than move away from an appetitive stimulus), which can override conscious control and contribute to the consumption of appetitive substances (Wiers, Gladwin, Hofmann, Salemink, & Ridderinkhof, 2013).

In support, a number of studies have demonstrated an approach bias toward a variety of food cues in both normal weight (e.g., Brockmeyer, Hahn, Reetz, Schmidt, & Friederich, 2015; Kemps, Tiggemann, Martin, & Elliott, 2013) and overweight or obese samples (Craeynest, Crombez, Koster, Haerens, & De Bourdeaudhuij, 2008; Havermans, Giesen, Houben, & Jansen, 2011; Mogg et al., 2012). However, only one study to date has directly compared approach bias toward food cues between normal weight and overweight individuals. Kemps and Tiggemann (2015) demonstrated an approach bias for food among obese women, but not among normal weight women. This finding suggests that automatic approach tendencies toward food cues may be strongest in overweight and obese individuals. A few studies have also shown a link between approach bias for unhealthy food cues and increased consumption of such food in mostly normal weight undergraduate student samples (Hofmann, Gschwendner, Wiers, Friese, & Schmitt, 2008; Nederkoorn, Houben, Hofmann, Roefs, & Jansen, 2010). As yet, this relationship has not been tested in overweight samples.

A different kind of contributor to unhealthy eating behaviour may be individual differences in eating style. Research has commonly used the Dutch Eating Behaviour Questionnaire (DEBQ; van Strien, Frijters, Bergers, & Defares, 1986) to measure three different types of trait eating style: restrained, emotional, and external eating. Restrained eating is the tendency to restrict food intake. Emotional eating is the tendency to eat when experiencing negative emotions. External eating is the tendency to eat in response to external food cues, such as the sight or smell of food. Although a number of studies have shown that all three types of eating style are associated with increased unhealthy food consumption (Anschutz, van Strien, van De Ven, & Engels, 2011; Elfhag, Tholin, & Rasmussen, 2008; van Strien, 2000; van Strien, Herman & Verheijden, 2012; Wardle et al., 2002), this has been

shown most reliably for external eating. For example, some studies have found that emotional eating did not predict increased unhealthy food consumption (Anschutz et al., 2011; van Strien et al., 2009; Wardle et al., 2002).

Trait eating style has also been associated with higher BMI in unselected undergraduate student samples (Burton, Smit, & Lightowler, 2007; Elfhag & Linne, 2005; Greene et al., 2011; Jasinska et al., 2012; Snoek, van Strien, Janssens, Rutger, & Engels, 2007; Verhoeven et al., 2015), although not all studies have found this for restrained eating (Boschi, Iorio, Margiotta, D'Orsi, & Falconi, 2001). Research has also shown that overweight individuals are more likely to display emotional or external eating than their normal weight counterparts (Braet & van Strien, 1997; Braet et al., 2008; de Lauzon-Guillain et al., 2006; Konttinen, Haukkala, Sarlio-Lähteenkorva, Silventoinen, & Jousilahti, 2009; van Strien, Herman, & Verheijden, 2009). Taken together, the findings suggest that trait eating style contributes to the eating behaviour of both normal weight and overweight people. As findings appear most consistent for the role of external eating in unhealthy food consumption and excess weight, this may be a particularly problematic trait.

The evidence suggests that both automatic approach tendencies toward food cues and trait eating style individually contribute to unhealthy eating behaviour, but their combination has not been investigated. As both approach bias for food and trait eating style tend to be greater in overweight than in normal weight individuals, their combination may be especially important for overweight and obese individuals. Furthermore, it has been suggested that automatic approach tendencies for food may play a particularly important role in the eating behaviour of individuals susceptible to eating in response to negative emotions or external food-related cues (Brockmeyer et al., 2015; Nijs & Franken, 2012) or after a period of restricted eating (Roefs & Jansen, 2002; Veenstra & de Jong, 2010). If so, then trait eating style may also play a moderating role in the effect of approach bias for food on consumption.

To date, no study has examined approach bias and eating styles together in predicting unhealthy eating behaviour; nor have the combined effects been compared between normal weight and overweight individuals.

Thus, the main aim of our study was to investigate the combined effect of approach bias for food and trait eating style on unhealthy food consumption in normal weight and overweight individuals. To obtain sufficient overweight participants, we combined three existing datasets of unselected samples. It was expected that approach bias and trait eating styles would better predict unhealthy food consumption in overweight people compared with normal weight people, and that trait eating style would moderate the effect of approach bias for food on consumption.

Method

Participants

The pooled sample included 245 participants from three studies that were conducted for different research purposes (Kakoschke et al., 2015; under review [control group]; submitted [control group]). All studies included measures of approach bias, trait eating style and snack food consumption. Participants were women aged 17 to 26 years (M = 20.20, SD =2.47) recruited from the undergraduate student population at Flinders University. Participants' weight and height were measured to calculate BMI (weight [kg]/height² [m²]). They were subsequently categorised as normal weight (BMI 18.5 – 24.9 kg/m²) or overweight (BMI >25 kg/m²) on the basis of the current accepted cut-off (WHO, 2013). This resulted in 180 normal weight participants (BMI M = 20.77, SD = 2.3) and 65 overweight participants (BMI M =29.90, SD = 5.0).

Materials

Approach-Avoidance Task. The Approach-Avoidance Task (AAT) was used to measure approach bias for unhealthy food based on a previous protocol (Wiers, Eberl, Rinck,

Becker, & Lindenmeyer, 2011). The stimuli were 40 digital coloured photographs (presented in a resolution of 1024 x 768 pixels), comprising 20 pictures of unhealthy foods and 20 pictures of non-food stimuli (depicting animals not normally eaten in Western society). Animals were chosen for the non-food comparison category as they, like food, are overall appealing. A portrait (aspect ratio 3:4) and landscape (aspect ratio 4:3) format of each image was created. The unhealthy food and animal pictures were obtained from a pilot test in which 21 women aged 17-45 years (M = 23.67, SD = 8.28) rated 590 pictures of unhealthy food and animals on 9-point pleasure and arousal scales (Kemps, Tiggemann, & Hollitt, 2014). Pictures were selected on the basis that categories did not significantly differ on mean ratings of pleasure and arousal (all ps > .05). Another 12 images of common objects were used for practice trials preceding the task.

The approach-avoidance task consisted of 160 trials. On each trial, participants began by pressing the start button on the top of a joystick. A picture of an unhealthy food or an animal then appeared in the centre of the screen. Participants were instructed to pull (approach) or push (avoid) the joystick according to whether the picture was presented in portrait or landscape format. These instructions were counterbalanced (i.e., half of the participants pulled for portrait and pushed for landscape and half vice versa). Pulling the joystick increased the picture size (simulating approach), while pushing the joystick decreased the picture size (simulating avoidance). The picture disappeared once the participant had pulled or pushed the joystick. Participants were asked to respond as quickly and accurately as possible. Each picture was presented four times, twice in portrait format and twice in landscape format. Thus, participants pulled and pushed the food and animal pictures with equal frequency (50/50).

For each participant, approach-avoidance scores for unhealthy food were calculated by subtracting median reaction times for pulling from pushing unhealthy food pictures (Wiers et al., 2011). Positive scores indicated an approach bias for food (i.e., faster pull than push), whereas negative scores indicated an avoidance bias away from food (i.e., faster push than pull).

Dutch Eating Behaviour Questionnaire (DEBQ). Trait eating styles were measured by the widely used DEBQ (van Strien et al., 1986), which comprises three subscales. Restrained Eating consists of 10 items (e.g., 'Do you try to eat less at mealtimes than you would like to eat?'). Emotional Eating consists of 13 items (e.g., 'Do you have a desire to eat when you are emotionally upset?'). External Eating consists of 10 items (e.g., 'If foods smells and looks good, do you eat more than usual?'). Each item is rated on a five-point scale ranging from 'never' (1) to 'very often' (5). For each subscale, a mean score is calculated, with higher scores indicating a greater level of restrained, emotional, or external eating. In the present sample, internal reliability was good for restrained eating ($\alpha = .93$) and emotional eating ($\alpha = .95$), while it was acceptable for external eating ($\alpha = .78$), which is comparable to a previous sample (van Strien et al., 1986).

Taste Test. Unhealthy snack food consumption was measured using a so-called taste test. Although the snacks in the taste test differed somewhat across the studies, all studies included chocolate M&Ms and potato crisps. These snacks were chosen as they are frequently consumed and are bite-sized to facilitate eating. They were presented in equally-filled bowls. Participants were instructed to taste as much or as little of the food as they liked so that they could rate each of the snacks on several characteristics (e.g., 'How sweet is this food?'). They were given 10 minutes to complete their ratings after which time the bowls were taken away. The amount of each food consumed was calculated by subtracting the weight (in grams) of the food after the taste test from the weight before the taste test. The weight of each food was then converted into the number of calories consumed and summed to obtain a total measure of snack consumption.

Procedure

All participants were tested in the Food Laboratory in the School of Psychology at Flinders University during a single session. Although other tasks were included in the different experiments, participants always completed the three tasks reported here in the same order, i.e., the Approach-Avoidance Task first, then the taste test, and finally, the measure of eating style (DEBQ). As a trait measure, the DEBQ was administered last to ensure that food consumption was not influenced by its completion.

Results

Sample Characteristics

As can be seen in Table 1, approach bias scores for unhealthy food were positive for both normal weight and overweight participants. When these were tested by one-sample *t*tests, results showed that overweight participants had a significant approach bias (different from zero), t(64) = 2.48, p = .016, while normal weight participants did not, t(181) = 1.64, p= .102. However, when scores were compared between groups, the difference was not significant. For trait eating style, Table 1 shows minimal differences between normal weight and overweight participants. Finally, for consumption, although the overweight participants ate more unhealthy food, this difference was not statistically significant.

Table 1

Means (and standard deviations) and inferential tests of differences between normal weight and overweight participants.

	Normal Weight	Inferential Statistics	
Variables	<i>M</i> (SD)	<i>M</i> (SD)	<i>t</i> (245)
Approach bias for unhealthy food	11.18 (91.88)	26.36 (85.68)	1.16, <i>p</i> = .246
DEBQ: Restrained eating score	2.79 (0.93)	2.80 (1.00)	0.06, <i>p</i> = .951
DEBQ Emotional eating score	3.31 (0.57)	3.34 (0.77)	0.56, <i>p</i> = .576
DEBQ External eating score	2.93 (1.03)	3.01 (1.01)	0.39, <i>p</i> = .697
Consumption of unhealthy food	84.41 (70.05)	97.87 (75.21)	1.31, <i>p</i> = .193

Effect of approach bias and eating style on food intake in normal weight and overweight women

A series of hierarchical regression analyses was conducted to examine whether approach bias for food and trait eating style predicted unhealthy snack food consumption. BMI group was included to determine whether the effects differed between normal weight and overweight participants. These analyses were performed separately for restrained, emotional, and external eating. For each analysis, centred approach bias and eating style scores, as well as BMI group were entered in Step 1. The two-way product terms were entered in Step 2 and the three-way product term was entered in Step 3. The resulting descriptive and inferential statistics are presented in Table 2. The significant interactions are indicated by change in R^2 (ΔR^2).

For restrained eating, results showed that none of Step 1, Step 2, or Step 3 was significant. For emotional eating, Step 1 and Step 2 were not significant. However, Step 3 was significant as the three-way product term (emotional eating x approach bias x BMI group) explained unique additional variance in unhealthy food consumption. For external eating, Step 1 was significant. In particular, it can be seen that external eating uniquely predicted unhealthy food consumption. Step 2 was not significant, but Step 3 was; the three-way product term (external eating x approach bias x BMI group) explained additional variance in consumption.

The three-way interactions involving BMI indicate that the results were different for normal weight and overweight participants. Thus, separate regression analyses were conducted for the two weight groups. Results (Table 3) showed no significant main effects or interactions for normal weight participants. In contrast, for overweight participants, there was a significant interaction between approach bias and emotional eating, and approach bias and external eating, but not between approach bias and restrained eating. As indicated by total R^2 for each group, the amount of variance explained in consumption was significant in the overweight group for emotional (19.9% and external eating (15.3%), but was not significant in the normal weight group (1.9% and 0.3% respectively).

Figure 1 displays the interactions for emotional and external eating style, significant in the above regression analyses. In line with what can be seen in Figure 1, simple slopes analyses showed no significant effects for normal weight participants. For overweight participants, however, there was a clear pattern. For participants high in emotional eating (one *SD* above the mean), approach bias was positively related to consumption, $\beta = .302$, t(61) = 2.24, p = .029, and for those low in emotional eating (one *SD* below the mean), approach bias was negatively related to consumption, $\beta = -.432$, t(61) = -2.38, p = .020. Likewise, approach bias was positively related to consumption for participants high in external eating, $\beta = .318$, t(60) = 2.24, p = .021, and was negatively related for those low in external eating, $\beta = -.273 t(60) = -.313$, p = .069 (although the latter did not reach statistical significance).

Table 2

Regression analyses of approach bias for food and scores on subscales of the DEBQ with BMI group in predicting unhealthy food consumption (calories).

		Restrained Eating				Emotional Eating				External Eating			
Ste	p Predictors	В	SE	b	р	В	SE	b	р	В	SE	b	р
1	Approach bias	.027	.051	.033	.603	.024	.051	.031	.638	.013	.051	.017	.796
	Eating style	6.45	4.82	.086	.181	897	4.58	013	.845	19.38	7.32	.169	.009*
	BMI group	11.32	10.39	.070	.277	11.86	10.62	.072	.265	11.16	10.55	.067	.291
		$R^2(3, 241) = .014, F = 1.13, p = .338$				$R^2(3, 240) = .006, \Delta F = .520, p = .668$				R^2 (3, 239) = .034, ΔF = .283, p = .039*			
2	Approach bias × eating style	601	.061	067	.317	.111	.054	.135	.041*	.161	.085	.121	.061
	Approach bias × BMI group	.048	.123	.079	.698	028	.124	047	.820	.039	.119	.065	.742
	Eating style × BMI group	-1.804	10.87	033	.868	2.71	10.73	.050	.801	3.73	15.27	.047	.807
		ΔR^2 (3, 238) = .005, F = .401, p = .753				ΔR^2 (3, 237) = .018, ΔF = 1.50, p = .218				ΔR^2 (3, 236) = .016, ΔF = 1.33, p = .266			
3	Approach bias × eating style × BMI group	143	.133	217	.282	.324	.126	.525	.011*	.452	.181	.478	.013*
		$\Delta R^2 (1, 237) = .005, F = 1.16, p = .282$				$\Delta R^2 (1, 240) = .026, \Delta F = 6.56, p = .011*$				$\Delta R^2 (1, 235) = .025, \Delta F = 6.22, p = .013*$			
	Total	$R^2(7, 244) = .024, F = .819, p = .573$				R^2 (7, 243) = .051, $F = 1.82, p = .084$				R^2 (7, 242) = .075, $F = 2.72, p = .010*$			

Table 3

Regression analyses of approach bias for food and scores on subscales of the DEBQ in predicting unhealthy food consumption (calories) separately for normal weight and overweight participants.

		Normal Weight				Overweight				
Step	Predictors	В	SE	b	р	В	SE	b	р	
1	Approach bias	.018	.058	.102	.762	.048	.113	.055	.671	
	Restrained eating	7.65	5.64	.102	.177	4.03	9.61	.054	.676	
		$R^2(2, 177) = .011, F = .1.004, p = .368$ $R^2(2, 62) = .005, F = .15$						F = .154, p	.857	
2	Approach bias x restrained eating	018	.072	019	.801	161	.117	180	.173	
		$\Delta R^2 (1, 1)$	76) = .000,	, $\Delta F = .064$, <i>p</i> = .801	ΔR^2 (1, 6	51) = .030,	$\Delta F = 1.90,$	<i>p</i> = .173	
	Total	Total $R^2(3, 176) = .012, F = .687, p = .561$ $R^2(3, 61) = .035, F = .738, p = .561$						9 = .534		
1	Approach bias	.019	.057	.025	.743	.038	.115	.043	.745	
	Emotional eating	-2.06	5.16	030	.691	2.54	10.03	.034	.801	
		$R^2(2, 1)$	79) = .001,	F = .123, F	<i>p</i> = .884	884 $R^2(2, 59) = .004, F = .105, p = .900$				
2	Approach bias x emotional eating	.036	.060	.045	.550	.360	.112	.413	.002*	
		ΔR^2 (1, 1)	78) = .002,	, $\Delta F = .358$, <i>p</i> = .550	50 $\Delta R^2 (1, 58) = .150, \Delta F = 10.27, p = .00$				
	Total	R^2 (3, 17)	78) = .003,	F = .201, F	<i>p</i> = .895	$R^2(3, 58) = .153, F = 3.51, p = .021*$				
1	Approach bias	.003	.057	.004	.953	.055	.111	.063	.624	
	External Eating	16.74	9.15	.137	.069	24.45	12.72	.245	.059	
		R^2 (2, 17)	79) = .019,	F = 1.72, F	<i>p</i> = .182	$R^{2}(2, 5)$	58) = 062,	F = 1.92, p	= .156	
2	Approach bias x external eating	.017	.102	.012	.872	.469	.150	.375	.003*	
		ΔR^2 (1, 1	78) = .000	, $\Delta F = .026$	5, <i>p</i> = 872	ΔR^2 (1, 5'	7) = .137, 4	$\Delta F = 9.74,$	<i>p</i> = .003*	
	Total	R^2 (3, 17)	78) = .019,	F = 1.15, F	<i>p</i> = .331	R^2 (3, 5'	7) = .199, 1	F = 4.72, p	= .005*	

(a) Emotional Eating



(b) External Eating



Figure 1. Interactions between approach bias and (a) emotional and (b) external eating on unhealthy food consumption (calories) separately for normal weight and overweight women.

Discussion

The current study aimed to examine the combined effects of approach bias for unhealthy food and trait eating style in predicting consumption of unhealthy food in normal weight and overweight individuals. Contrary to predictions, there were no mean differences in approach bias, eating style, or snack food consumption between weight groups. The results did show that trait external eating predicted increased consumption of unhealthy food. Importantly, there were differences in the combined prediction of unhealthy food consumption between normal weight and overweight participants. Among overweight participants, external and emotional eating style individually moderated the relationship between approach bias and consumption. Specifically, approach bias for unhealthy food was positively related to consumption in high external and emotional eaters, but negatively related to consumption in low external and emotional eaters. These interactions were not observed among normal weight participants.

The results indicated that the overweight group showed a significant approach bias for unhealthy food while the normal weight group did not. This finding is consistent with that of a previous study, which showed an approach bias for food in obese, but not in normal weight, individuals (Kemps & Tiggemann, 2015). Our failure to find a group difference in approach bias might be due to our less extreme overweight sample, which comprised mostly of people (63%) in the overweight category (BMI = 25-30) rather than in the obese category (BMI > 30). Future research could specifically investigate differences between normal weight, overweight and obese individuals in their approach bias for unhealthy food and subsequent consumption.

In terms of individual predictors of eating behaviour, the current finding that an approach bias for unhealthy food did not predict unhealthy food consumption is at odds with two studies that showed this effect in unselected student samples (Hofmann & Gschwendner et al., 2008; Nederkoorn et al., 2010). However, these latter studies also showed that the automatic approach bias was more predictive of eating behaviour for individuals with poor self-regulatory control. Thus, it might be that approach bias toward unhealthy food cues results in increased unhealthy food consumption in combination with other predictors of eating behaviour.

For trait eating style, the finding that restrained or emotional eating were not unique predictors of unhealthy food consumption is not altogether surprising as the previous research has been somewhat mixed. Importantly, however, external eating did predict consumption of unhealthy food. This is consistent with the literature showing that external eating is the most robust eating style for predicting unhealthy food consumption (e.g., Anschutz et al., 2011; Elfhag et al., 2008; Wardle et al., 2002). This finding that a heightened responsiveness to external food-related cues predicts unhealthy eating supports the underlying premise of Schachter's classic externality theory (Schachter, 1968), which posits that external eaters are more susceptible to eating in response to cues such as the sight and smell of food, rather than internal cues (hunger). This is particularly important in the current 'obesogenic' environment where such individuals might find it particularly difficult to resist the lure of advertising and ready availability of fast food.

The major contribution of the present study, however, was that the combined effect of approach bias for food and trait eating style better predicted consumption of unhealthy food in overweight than in normal weight people, and that eating style moderated the effect of approach bias on consumption in the overweight group. Specifically, it was found that approach bias for unhealthy food together with high external or emotional eating predicted increased consumption of unhealthy food. In regards to external eating, it is likely that a heightened responsiveness to external food-related cues and a greater tendency to approach unhealthy food will be particularly problematic for overeating and weight gain in an environment that continually exposes people to an abundance of appetitive food and related cues. In terms of emotional eating, a predisposition to eat in response to negative emotions together with a greater tendency to approach unhealthy food likely contributes to excessive food consumption and weight gain. It is important to note that while the measure of emotional eating used in the present study (DEBQ) assessed the tendency to eat in response to negative emotions, some recent studies have demonstrated that overeating also occurs in response to positive emotions (Bongers, Jansen, Havermans, Roefs, & Nederkoorn, 2013; Holland, de Vries, Hermsen, & van Knippenberg, 2012). Future research could examine how experiencing different emotions might influence the eating behaviour of overweight people in combination with an approach bias toward unhealthy food.

Although we did not find that approach bias and eating style (except for external eating) individually predicted unhealthy eating behaviour, the results highlight that their combined effect is important for predicting consumption in overweight, but not in normal weight, individuals. The absence of findings for normal weight people is somewhat surprising based on the results of some previous studies (e.g., Hofmann & Gschwendner et al., 2008; Nederkoorn et al., 2010). One potential reason for these mixed findings is that unselected undergraduate student samples undoubtedly contain overweight individuals as well as normal weight individuals, given that more than 52% of the general population are overweight or obese (WHO, 2014). Our results suggest the intriguing possibility that it may be the overweight people who are driving the observed effects of approach bias for food and eating style in the prior unselected samples.

Theoretically, the current findings lend some support to the proposition that unhealthy eating behaviour is partly determined by automatic processing of appetitive food cues (Strack & Deutsch, 2004). Although approach bias for food did not predict eating behaviour alone, it did so in some individuals. In particular, the current results suggest that overweight people

who tend to eat high-calorie food when they are feeling sad or distressed (emotional eaters), as well as those who tend to eat when they see or smell an appealing food, such as a piece of chocolate cake (external eaters), might be especially susceptible. Thus, appetitive food and related cues in the environment likely do elicit automatic approach tendencies, but more so for certain individuals.

The present study has some important practical implications. The finding that approach bias interacted with trait eating style in predicting consumption in overweight people suggests that approach bias modification training might be useful for emotional and external eaters. The most commonly used task for re-training approach bias is the Approach-Avoidance Task (Rinck & Becker, 2007). In this task, participants are trained to avoid appetitive pictures (e.g., unhealthy food) by consistently presenting such cues in a format that requires them to be pushed (avoided) and control pictures in a format that requires them to be pulled (approached). Studies have shown that successfully reducing approach bias for unhealthy food can decrease unhealthy snack choice and consumption of chocolate (Becker, Jostmann, Wiers, & Holland, 2015, Study 1; Fishbach & Shah, 2006, Study 5; Kakoschke et al., under review; Schumacher, Kemps, & Tiggemann, 2016). However, other studies have found no effect of approach bias re-training on food consumption or choice (Becker et al., 2015, Study 2, Study 3; Dickson, Kavanagh, & MacLeod, 2015). Thus, the mixed findings might be due to individual differences in weight status and/or eating style not being taken into account. The current results suggest that overweight individuals may benefit more from approach bias re-training, particularly if they are external or emotional eaters. Future studies might usefully pursue the identification of individuals most likely to benefit from such retraining.

In conclusion, the present study has clearly shown combined effects of approach bias for food and external or emotional eating in predicting unhealthy food consumption in overweight, but not in normal weight, individuals. The findings contribute to a theoretical understanding of the automatic processes underlying unhealthy eating behaviour for overweight and obese people. At a practical level, approach bias for food presents a potential target for modifying unhealthy eating behaviour, particularly among overweight individuals who have a tendency to eat in response to negative emotions or when exposed to external food cues.

References

- Anschutz, D. J., van Strien, T., Van De Ven, M. O., & Engels, R. C. (2009). Eating styles and energy intake in young women. *Appetite*, *53*, 119–22.
- Becker, D., Jostmann, N. B., Wiers, R. W., & Holland, R. W. (2015). Approach avoidance training in the eating domain: Testing the effectiveness across three single session studies. *Appetite*, 85, 58–65.
- Bongers, P., Jansen, A., Havermans, R., Roefs, A., & Nederkoorn, C. (2013). Happy eating. The underestimated role of overeating in a positive mood. *Appetite*, *67*, 74–80.
- Boschi, V., Iorio, D., Margiotta, N., Orsi, P. D', & Falconi, C. (2001). The three-factor eating questionnaire in the evaluation of eating behaviour in subjects seeking participation in a dietotherapy programme. *Annals of Nutrition & Metabolism*, *45*, 72–77.
- Braet, C., & Van Strien, T. (1997). Assessment of emotional, externally induced and restrained eating behaviour in nine to twelve-year-old obese and non-obese children. *Behaviour Research and Therapy*, 35, 863-873.
- Braet, C., Claus, L., Goossens, L., Moens, E., Van Vlierberghe, L., & Soetens, B. (2008).
 Differences in eating style between overweight and normal-weight youngsters. *Journal of Health Psychology*, *13*, 733-743.
- Brockmeyer, T., Hahn, C., Reetz, C., Schmidt, U., & Friederich, H.-C. (2015). Approach bias and cue reactivity towards food in people with high versus low levels of food craving. *Appetite*, 95, 197–202.
- Brunner, T. A., van der Horst, K., & Siegrist, M. (2010). Convenience food products. Drivers for consumption. *Appetite*, *55*, 498–506.
- Burton, P., Smit, H. J., & Lightowler, H. J. (2007). The influence of restrained and external eating patterns on overeating. *Appetite*, *49*, 191–197.

- Carnell, S., Kim, Y., & Pryor, K. (2012). Fat brains, greedy genes, and parent power. A biobehavioural risk model of child and adult obesity. *International Review of Psychiatry*, 24, 189–199.
- Craeynest, M., Crombez, G., Koster, E. H., Haerens, L., & De Bourdeaudhuij, I. (2008).
 Cognitive-motivational determinants of fat food consumption in overweight and obese youngsters: The implicit association between fat food and arousal. *Journal of Behavior Therapy and Experimental Psychiatry*, *39*, 354-368.
- de Lauzon-Guillain, B., Basdevant, A., Romon, M., Karlsson, J., Borys, J. M., Charles, M.A., & The FLVS Study Group. (2006). Is restrained eating a risk factor for weight gain in a general population? *American Journal of Clinical Nutrition*, 83, 132-138.
- Dickson, H., Kavanagh, D. J., & MacLeod, C. (2015). The pulling power of chocolate: Effects of approach-avoidance training on approach bias and consumption. *Appetite*, *99*, 46-51.
- Elfhag, K., & Linné, Y. (2005). Gender differences in associations of eating pathology between mothers and their adolescent offspring. *Obesity Research*, *13*, 1070–1076.
- Elfhag, K., Tholin, S., & Rasmussen, F. (2008). Consumption of fruit, vegetables, sweets and soft drinks are associated with psychological dimensions of eating behaviour in parents and their 12-year-old children. *Public Health Nutrition*, *11*, 914-923.
- Evers, C., de Ridder, D. T. D., & Adriaanse, M. A. (2009). Assessing yourself as an emotional eater: Mission impossible? *Health Psychology*, 28, 717-725.
- Greene, G. W., Schembre, S. M., White, A. A., Hoerr, S. L., Lohse, B., et al. (2011).
 Identifying clusters of college students at elevated health risk based on eating and exercise behaviors and psychosocial determinants of body weight. *Journal American Dietetic Association*, 111, 394–400.

- Havermans, R. C., Giesen, J. C., Houben, K., & Jansen, A. (2011). Weight, gender, and snack appeal. *Eating behaviors*, *12*, 126-130.
- Hill, J. O., Wyatt, H. R., Reed, G. W., & Peters, J. C. (2003). Obesity and the environment: where do we go from here? *Science*, *299*, 853-855.
- Hofmann, W., Gschwendner, T., Wiers, R., Friese, M., & Schmitt, M. (2008b). Working memory capacity and self-regulation. Towards an individual differences perspective on behavior determination by automatic versus controlled processes. *Journal of Personality and Social Psychology*, 95, 962–977
- Holland, R. W., de Vries, M., Hermsen, B., & van Knippenberg, A. (2012). Mood and the attitude–behavior link: The happy act on impulse, the sad think twice. *Social Psychological and Personality Science*, *3*, 356-364.
- Jasinska, A. J., Yasuda, M., Burant, C. F., Gregor, N., Khatri, S., Sweet, M., & Falk, E. B. (2012). Impulsivity and inhibitory control deficits are associated with unhealthy eating in young adults. *Appetite*, 59, 738–747.
- Kakoschke, N., Kemps, E., & Tiggemann, M. (2015). Combined effects of cognitive bias for food cues and poor inhibitory control on unhealthy food intake. *Appetite*, *87*, 358-364.
- Kemps, E., & Tiggemann, M. (2015). Approach bias for food cues in obese individuals. *Psychology & Health, 30*, 370-380.
- Kemps, E., Tiggemann, M., Martin, R., & Elliott, M. (2013). Implicit approach–avoidance associations for craved food cues. Journal of Experimental Psychology: Applied, 19, 30-38.
- Konttinen, H., Haukkala, A., Sarlio-Lähteenkorva, S., Silventoinen, K., & Jousilahti, P.
 (2009). Eating styles, self-control and obesity indicators: The moderating role of obesity status and dieting history on restrained eating. *Appetite*, *53*, 131-134.

- Marteau, T. M., Hollands, G. J., & Fletcher, P. C. (2012). Changing human behavior to prevent disease: The importance of targeting automatic processes. *Science*, 337, 1492-1495.
- Mogg, K., Bradley, B. P., O'Neill, B., Bani, M., Merlo-Pich, E., Koch, A., ... & Nathan, P. J.
 (2012). Effect of dopamine D3 receptor antagonism on approach responses to food cues in overweight and obese individuals. *Behavioural Pharmacology*, 23, 603-608.
- Nederkoorn, C., Houben, K., Hofmann, W., Roefs, A., & Jansen, A. (2010). Control yourself or just eat what you like? Weight gain over a year is predicted by an interactive effect of response inhibition and implicit preference for snack foods. *Health Psychology*, 29, 389–393.
- Nijs, I. M. T., & Franken, I. H. A. (2012). Attentional processing of food cues in overweight and obese individuals. *Current Obesity Reports*, *1*, 106–113.
- Ouwehand, C., & Papies, E. K. (2010). Eat it or beat it. The differential effects of food temptations on overweight and normal-weight restrained eaters. *Appetite*, *55*, 56-60.
- Rinck, M., & Becker, E. S. (2007). Approach and avoidance in fear of spiders. *Journal of Behavior Therapy and Experimental Psychiatry*, 38, 105–120.
- Roefs, A., & Jansen, A. (2002). Implicit and explicit attitudes toward high-fat foods in obesity. *Journal of Abnormal Psychology*, *111*, 517–521.
- Schachter, S. (1968). Obesity and eating. Science, 161, 751-756.
- Schumacher, S. E., Kemps, E., & Tiggemann, M. (2016). Bias modification training can alter approach bias and chocolate consumption. *Appetite*, *96*, 219–224.
- Seibt, B., Hafner, M., & Deutsch, R. (2007). Prepared to eat: How immediate affective and motivational responses to food cues are influenced by food deprivation. *European Journal of Social Psychology*, 37, 359-379.

Snoek, H. M., Van Strien, T., Janssens, J. M. A. M., & Engels, R. C. M. E. (2007). Emotional, external, restrained eating and overweight in Dutch adolescents. *Scandinavian Journal of Psychology*, 48, 23–32.

- Stunkard, A. J., & Messick, S. (1985). The Three Factor Eating Questionnaire to measure dietary restraint, disinhibition and hunger. *Journal of Psychosomatic Research*, 29, 71–83.
- Strack, F., & Deutsch, R. (2004). Reflective and impulsive determinants of social behavior. Personality and Social Psychology Review, 8, 220-247.
- Tuschl, R. J. (1990). From dietary restraint to binge eating: Some theoretical considerations. *Appetite*, *14*, 105–109.
- van Strien, T. (2000). Ice-cream consumption, tendency toward overeating, and personality. *The International Journal of Eating Disorders*, 28, 460–464.
- van Strien, T., Frijters, J. E., Bergers, G. P., & Defares, P. B. (1986). The Dutch Eating Behavior Questionnaire (DEBQ) for assessment of restrained, emotional, and external eating behavior. *International Journal Eating Disorders*, *5*, 295–315
- van Strien, T., Herman, C. P., & Verheijden, M. W. (2009). Eating style, overeating, and overweight in a representative Dutch sample: Does external eating play a role? *Appetite*, 52, 380-387.
- van Strien, T., Herman, C. P., & Verheijden, M. W. (2012). Eating style, overeating and weight gain: A prospective 2-year follow-up study in a representative Dutch sample. *Appetite*, 59, 782-789.
- Veenstra, E. M., & de Jong, P. J. (2010). Restrained eaters show enhanced automatic approach tendencies towards food. *Appetite*, *55*, 30-36.

- Verhoeven, A. A., Adriaanse, M. A., de Vet, E., Fennis, B. M., & de Ridder, D. T. (2015).
 It's my party and I eat if I want to. Reasons for unhealthy snacking. *Appetite*, 84, 20-27.
- Wardle, J., Marsland, L., Sheikh, Y., Quinn, M., Fedoroff, I., & Ogden, J. (1992). Eating style and eating behaviour in adolescents. *Appetite*, 18, 167-183.
- Wiers, R. W., Eberl, C., Rinck, M., Becker, E. S., & Lindenmeyer, J. (2011). Retraining automatic action tendencies changes alcoholic patients' approach bias for alcohol and improves treatment outcome. *Psychological Science*, 22, 490–497.
- Wiers, R. W., Gladwin, T. E., Hofmann, W., Salemink, E., & Ridderinkhof, K. R. (2013).
 Cognitive bias modification and cognitive control training in addiction and related psychopathology mechanisms, clinical perspectives, and ways forward. *Clinical Psychological Science*, 1, 192-212.
- World Health Organization [WHO]. (2013). Body Mass Index (BMI) classifications. Retrieved November 14 2016, from

http://www.who.int/bmi/index.jsp?introPage=intro_3.html.

World Health Organization [WHO]. (2014). Obesity and overweight. Retrieved October 1, 2015 from http://www.who.int/mediacentre/factsheets/fs311/en/.

CHAPTER 5: LITERATURE REVIEW

Approach Bias Modification Training and Consumption: A Review of the Literature

Naomi Kakoschke, Eva Kemps, and Marika Tiggemann School of Psychology, Flinders University, South Australia

Corresponding author: Naomi Kakoschke, School of Psychology, Flinders University, GPO Box 2100, Adelaide, 5001, South Australia. Email: naomi.kakoschke@flinders.edu.au **Funding:** This research received no specific grant from any funding agencies in the public, commercial, or not-for-profit sector.

Statement of co-authorship: All authors were involved in the formulation of the review concept and design. Naomi Kakoschke conducted the literature searches and completed the initial draft of the manuscript. Eva Kemps and Marika Tiggemann edited multiple revisions of the manuscript.

This manuscript has been published as:

Kakoschke, N., Kemps, E., & Tiggemann, M. (2017). Approach bias modification training and consumption: A review of the literature. *Addictive Behaviors*, *64*, 21-28. DOI 10.1016/j.addbeh.2016.08.007

Abstract

Recent theoretical perspectives and empirical evidence have suggested that biased cognitive processing is an important contributor to unhealthy behaviour. Approach bias modification is a novel intervention in which approach biases for appetitive cues are modified. The current review of the literature aimed to evaluate the effectiveness of modifying approach bias for harmful consumption behaviours, including alcohol use, cigarette smoking, and unhealthy eating. Relevant publications were identified through a search of four electronic databases (PsycINFO, Google Scholar, ScienceDirect and Scopus) that were conducted between October and December 2015. Eligibility criteria included the use of a human adult sample, at least one session of avoidance training, and an outcome measure related to the behaviour of interest. The fifteen identified publications (comprising 18 individual studies) were coded on a number of characteristics, including consumption behaviour, participants, task, training and control conditions, number of training sessions and trials, outcome measure, and results. The results generally showed positive effects of approach-avoidance training, including reduced consumption behaviour in the laboratory, lower relapse rates, and improvements in self-reported measures of behaviour. Importantly, all studies (with one exception) that reported favourable consumption outcomes also demonstrated successful reduction of the approach bias for appetitive cues. Thus, the current review concluded that approach bias modification is effective for reducing both approach bias and unhealthy consumption behaviour.

1. Introduction

Engagement in damaging health behaviours including excessive alcohol use, smoking and unhealthy eating can lead to the development of chronic health conditions (WHO, 2016). In particular, the current Western diet, characterised by the overconsumption of high-calorie food, has been linked to obesity, while the use of alcohol and tobacco is related to the three major causes of death in the world, namely heart disease, cancer and accidents (WHO, 2016). Therefore, it is important to develop interventions that aim to modify unhealthy behaviours to improve health outcomes. Health behaviour change interventions are typically informationbased and aim to encourage individuals to reflect upon and monitor their own behaviour; however, such interventions are not always effective (Marteau, Hollands, & Fletcher, 2012). One reason is that health behaviour interventions occur within a larger context, namely in an environment in which advertisements for alcohol, tobacco and unhealthy food are abundant (Amanzadeh et al., 2015, Bestman et al., 2015; Zimmerman & Shimoga, 2014). It has been further argued that the automatic processing of cues in such advertisements can lead to harmful consumption behaviours (Cohen & Babey, 2012; Kemps et al., 2014; Marteau et al., 2012), a consideration of which could improve the effectiveness of these interventions (Hofmann, Friese, & Wiers, 2008).

One automatic mechanism implicated in the development of unhealthy behaviour is biased cognitive processing of appetitive cues (Marteau et al., 2012). A cognitive bias refers to the selective processing of relevant cues over other cues in the environment (MacLeod & Mathews, 2012). Contemporary dual-process models of behaviour posit that both automatic and controlled processing determine health behaviour (Strack & Deutsch, 2004). Automatic processing, the focus of the current review, is fast, implicit, and effortless, while controlled processing is slower and involves conscious reflection based on long-term goals and personal standards. Moreover, automatic processing can involve biases, including both attending to (attentional bias) and approaching (approach bias) unhealthy appetitive cues (e.g., a slice of chocolate cake) in the environment (Wiers, Gladwin, Hofmann, Salemink, & Ridderinkoff, 2013). Most research to date on cognitive biases has focused on attentional bias, which refers to an automatic allocation of attention to particular cues in the environment (Beard et al., 2012; MacLeod & Mathews, 2012). More recently, however, researchers have turned their focus toward approach bias, which is the automatic tendency to reach out toward (i.e., approach) rather than move away from (i.e., avoid) appetitive cues (Wiers, Gladwin, et al., 2013). Approach bias toward appetitive cues is likely to be an important contributor to unhealthy consumption due to the additional behavioural component over and above attentional bias.

The existence of approach biases has been documented for a variety of appetitive cues, including alcohol (e.g., Wiers et al., 2009; Wiers et al., 2010), cigarettes (e.g., Bradley et al., 2008; Wiers et al., 2013), cannabis (Cousijn et al., 2011; Field et al., 2006) and unhealthy food (e.g., Havermans et al., 2011; Veenstra and de Jong, 2010). Furthermore, some researchers have demonstrated that approach biases are linked to increased consumption of alcohol (Wiers et al., 2009, Wiers et al., 2010; Palfai & Ostafin, 2003), cigarettes (Watson et al., 2013; Wiers et al., 2013), cannabis (Cousijn et al., 2011; Cousijn et al., 2012) and unhealthy food (Kakoschke et al., 2015; Nederkoorn et al., 2010). Thus, evidence suggests that the automatic tendency to approach rather than avoid appetitive cues may play a key role in unhealthy consumption behaviour.

Importantly, researchers have begun to investigate whether approach biases can be modified. While several specific tasks can be used to modify approach bias, most researchers have used the Approach-Avoidance Task (AAT), originally designed for the anxiety domain (Rinck & Becker, 2007). In this task, participants are instructed to respond to a picture based on an irrelevant feature (e.g., portrait versus landscape orientation) by pushing or pulling a joystick. Participants are asked to respond based on a feature of the picture, rather than its content, to ensure that the task captures automatic cognitive processing (Wiers, Gladwin, et al., 2013). The AAT also has a 'zooming' feature, such that pulling the joystick will increase the picture size while pushing the joystick will decrease the picture size, which gives participants a sense of respectively approaching or avoiding the cue (Neumann & Strack, 2000).

An avoidance of appetitive cues can be trained using the AAT by manipulating the contingencies of the target (appetitive) and control pictures. Specifically, the target pictures are consistently presented in a format that requires them to be pushed (avoided) and control pictures in a format that requires them to be pulled (approached). In contrast, approach training involves the reverse contingencies, while what has been termed sham-training (a neutral condition) is when the target and control pictures are approached and avoided with equal frequency. Avoidance-training is considered successful if training reduces the approach bias for a specific appetitive substance. Although there has been considerable variability in the protocols used (e.g., the number of training trials varies from 48 to 600), there is evidence to suggest that the AAT can reliably modify approach bias (e.g., Eberl et al., 2013).

The focus of the present literature review is to evaluate the effectiveness of modifying approach biases for appetitive substances. In their more general review of cognitive training techniques for addictive behaviours, Wiers, Gladwin, et al. (2013) included three studies concerning approach bias for alcohol cues. The review concluded that approach-avoidance training was effective for re-training approach bias to alcohol cues and consumption behaviour in both undergraduate and clinical samples. However, since then, approachavoidance training has been examined for a range of unhealthy consumption behaviours beyond (excessive) alcohol consumption, in particular, for cigarette smoking and unhealthy eating. Therefore, the current review of the approach bias modification literature extends upon the review of Wiers et al. by including more recent studies on alcohol, as well as studies on cigarette smoking and eating.

2. Method

2.1 Search strategy

The Preferred Reporting Items for Systematic Review (PRISMA) guidelines (Moher, Liberati, Tetzlaff, & Altman, 2009) were followed. Relevant articles were identified through a search of four electronic databases (PsycINFO, Google Scholar, ScienceDirect and Scopus) conducted between October and December 2015. The search terms used were: (approach bias modification OR approach avoidance training) AND (eating OR eating behaviour OR food OR smoking OR cigarettes OR alcohol OR drinking). To the best of our knowledge, no studies had been published prior to 2006, so searches were limited to studies published during the last 10 years (2006 to current). Articles were included if they had been peer-reviewed, published, and contained an original empirical study. Papers were excluded if they were a review article, conference paper, study protocol, abstract, or an unpublished dissertation or thesis. Searches were also limited to articles available in the English language. In addition, manual searches were conducted of the reference sections of all included articles to identify other potentially eligible studies, which resulted in the identification of one additional paper (Fishbach & Shah, 2006). An outline of the full literature search strategy for the four electronic databases, including any search limits used, is provided in Appendix A.

2.2 Eligibility criteria

To be included in the current literature review, studies needed to meet the following eligibility criteria. First, studies had to recruit a sample of human adult participants (i.e., aged over 18 years). Second, studies had to use at least one session of approach-avoidance training to re-train participants to avoid appetitive cues. Finally, studies had to measure a behavioural outcome, such as self-reported consumption or intake during a laboratory-based taste test.

2.3 Study selection

In total, the literature search retrieved 437 articles. After removing duplicates, 263 articles were title screened. Based on the eligibility criteria previously outlined, 214 articles were excluded. Abstracts of the remaining 49 articles were examined to determine eligibility, resulting in the exclusion of a further 29 articles that did not include consumption behaviour. An additional five articles were excluded based on an in-depth assessment of the full text. Thus, 15 articles (comprising 18 individual studies) were identified for inclusion in the current literature review (refer to Figure 1 for a detailed explanation of each of the steps).



Figure 1. Flow diagram for the search and inclusion criteria of studies in the literature review.

3. Results

3.1 Study characteristics

Detailed information was extracted from the 15 articles (comprising 18 individual studies) that met the inclusion criteria for this review. Included articles were reviewed based on the following characteristics: consumption behaviour, participants, task type, training and control conditions, number of training sessions and trials in each session, success of approach bias re-training, outcome measure, and results (see Table 1 for a detailed description). The included articles examined alcohol consumption (n = 7), cigarette smoking (n = 3), and eating behaviour (n = 5). Samples comprised both undergraduate students and clinical inpatients, and ranged in size from 24 to 475. The majority of the included studies used the AAT as their training protocol (only two used a different task). As per one of the eligibility criteria, all of the studies included a training condition that was designed to train an avoidance of appetitive cues. However, the comparison condition varied between studies (approach training, shamtraining, or no training). The number of training sessions varied between 1 and 12, while the number of trials in each session ranged from 100 to 600. The outcomes comprised subjective (self-reported consumption) and objective measures (consumption in a laboratory 'taste test').

3.2 Alcohol consumption

Seven articles, comprising eight studies, focused on alcohol consumption. All of these used the AAT to train avoidance of alcohol cues. Three studies with clinical samples of alcohol-dependent inpatients compared avoidance training with a no-training condition (Eberl et al., 2013, Eberl et al., 2014; Wiers et al., 2011). Three studies compared avoidance training with sham-training in self-identified heavy drinkers recruited from undergraduate students (Lindgren et al., 2015, Study 1 and Study 2) or the community (Wiers et al., 2015). The remaining two studies compared avoidance-training and approach-training in undergraduate samples with self-identified heavy drinkers (Sharbanee et al., 2014; Wiers et al., 2010).
Table 1

Characteristics and Effect Sizes of Studies Included in the Literature Review.

Authors/Year	Consumption	Participants	Task	Training	Control	Sessions; trials	Was approach	Outcome	Result (training
	Behaviour			condition	condition	in each session	bias modified?	measure	versus control)
Becker et al.	Eating	52 students	AAT	Avoid	Sham-training	1;360	No	Self-reported and	Training group
(2015; Study 1)	behaviour			unhealthy				behavioural	chose healthier
				food				food choice	snacks*; $d = .74$
Becker et al.	Eating	104 students	AAT	Avoid	Sham-training	1;360	No	Self-reported	n.s.; $d = .14$
(2015; Study 2)	behaviour			unhealthy				food choice	
				food					
Becker et al.	Eating	103 students	AAT	Avoid	Sham-training	1; 320	No	Chocolate	Training group
(2015; Study 3)	behaviour			unhealthy				consumption	consumed more
				food				in taste test	chocolate; $d = .46$
Brockmeyer	Eating	30 female	AAT	Avoid	No control	10 across 5	Yes	Self-reported	Training group had
et al. (2015)	behaviour	students; trait		unhealthy	group	weeks; 402		trait food	reduced cravings,
		food cravers		food				craving; daily	d = 1.11; improved
								eating habits	daily eating habits
Eberl et al.	Alcohol	475 alcohol	AAT	Avoid	No training	12 across 12	Yes	Relapse rates at	Training group
(2013)	Consumption	dependent		Alcohol		weeks; 200		one-year	had lower relapse
		patients						follow-up	rates; $\eta^2 = .09$
Eberl et al.	Alcohol	111 alcohol	AAT	Avoid	No training	12 across 12	Yes	Relapse rates at	Training group
(2014)	Consumption	dependent		Alcohol		weeks; 200		one-year	had lower relapse
		patients						follow-up	rates; n/a
Fishbach &	Eating	24 female	Joystick	Push	Push healthy	1; 120	Yes	Behavioural	Training group
Shah (2006;	behaviour	students	category	unhealthy	food			food choice	made healthier
Study 5)			judgement	food					choices; <i>d</i> = .96
Kemps et al.	Eating	96 female	IAT	Avoid	Approach	1;240	Yes	Self-reported	Training group had
(2013; Study 2)	behaviour	students		chocolate	chocolate			chocolate	lower cravings; n.s.
								craving	<i>d</i> = .21

Kong et al. (2015)	Cigarette Smoking	60 adolescent smokers	AAT	Avoid cigarettes	Sham-training	4 across 4 weeks; 260	No	Self-reported 7-day cigarette consumption	n.s., <i>d</i> = .54
Lindgren et al. (2015; Study1)	Alcohol Consumption	300 students; social drinkers	AAT	Avoid alcohol	Sham-training	2 across 2 weeks; 600	No	Self-reported drinking intention	n.s; $\eta^2 = .02$
Lindgren et al. (2015; Study 2)	Alcohol Consumption	288 students; problem drinkers	AAT	Avoid alcohol	Sham-training	2 across 1 week; 600	No	Self-reported drinking intention	n.s; $\eta^2 = .04$
Machulska et al. (2016)	Cigarette Smoking	139 inpatient psychiatric adult smokers	AAT	Avoid cigarettes	Sham-training	4 across 4 consecutive days; 250	No	Self-reported daily cigarette consumption	Training group smoked fewer cigarettes (3 month follow-up); $\eta^2 = .04$
Schumacher et al. (2016)	Eating Behaviour	120 female students	AAT	Avoid chocolate	Approach chocolate	1; 240	Yes	Chocolate consumption in taste test	Training group consumed less chocolate; $d = .37$
Sharbanee et al. (2014)	Alcohol Consumption	74 students; social drinkers	AAT	Avoid alcohol	Approach alcohol or sham-training	1; 384	Yes	Beer consumption in taste test	Training group consumed less beer*; n/a
Wiers et al. (2010)	Alcohol Consumption	42 students; male problem drinkers	AAT	Avoid alcohol	Approach alcohol	1; 440	Yes	Beer consumption in taste test	Training group consumed less beer*; n/a
Wiers et al. (2011)	Alcohol Consumption	214 alcohol dependent patients	AAT	Avoid alcohol	Sham-training or no training	4 over 4 days; 200	Yes	Relapse rates at one-year follow-up	Training group had lower relapse rates; $d = .28$
Wiers et al. (2015)	Alcohol Consumption	136 problem drinkers	AAT	Avoid alcohol	Sham-training	4 across 2-14 days; 220	No (approach bias data was not reported)	Self-reported alcohol consumption	n.s.; n/a
Wittekind et al. (2015)	Cigarette Smoking	257 smokers	AAT	Avoid cigarettes	No training	1; 100	Approach bias not measured	Self-reported cigarette smoking	Training group smoked fewer cigarettes; $\eta^2 = .06$

Note. AAT = Approach-Avoidance Task; IAT = Implicit Association Task; [#] bias mediated effect of training group on outcome; n.s. = not statistically significant at .05 level; * = significant results obtained only for successfully trained participants. Effect sizes were partial eta² (η^2) for ANOVA and Cohen's *d* for t-tests; n/a = insufficient data available.

In terms of the results, five of the eight studies showed successful re-training of the approach bias (at least for some participants). All of these also showed that approach bias change translated into effects on consumption. Specifically, the three studies with alcohol-dependent inpatients found that those in the training group were less likely to have relapsed at one-year follow-up than those who received treatment as usual (Eberl et al., 2013; Eberl et al., 2014; Wiers et al., 2011). Although the two studies with undergraduate students found no main effect of training condition on alcohol consumption during a laboratory taste test, they did find the effect for a subset of participants who were successfully re-trained toward an avoidance of alcohol cues (Sharbanee et al., 2014; Wiers et al., 2010). This subset of participants showed a change in AAT score in the predicted direction from pre- to post-training. In addition, Eberl et al. (2013) explicitly tested for mediation and found that a change in approach bias for alcohol cues fully mediated the effect of training on relapse rates one-year later in a clinical sample. Similarly, Sharbanee et al. (2015) found a significant indirect effect of training condition on alcohol consumption through approach bias.

The remaining three studies investigated samples of problem drinkers and used shamtraining as a control. Two found no significant effect of training condition on approach bias (Lindgren et al., Study 1 and Study 2), and one did not report approach bias scores (Wiers et al., 2015). Importantly, these three studies did not find any effect of training on consumption. Specifically, Lindgren et al. (2015; Study 1 and Study 2) found no difference between the training and control groups in self-reported cravings or intention to drink alcohol during the following week, while Wiers et al. (2015) found that self-reported alcohol consumption in the two-week period following training reduced equally for the training and control groups.

In summary, five out of eight studies (across undergraduate and clinical samples) found that the approach bias was successfully re-trained toward avoidance of alcohol cues (Eberl et al., 2013; Eberl et al., 2014; Sharbanee et al., 2015; Wiers et al., 2010; Wiers et al., 2011). In terms of behaviour, these same five studies also reported a significant group difference in alcohol consumption or relapse rates (at least in the successfully re-trained participants). Thus, it appears that the main difference between studies that did and those that did not produce a positive effect on consumption was the successful reduction of approach bias for alcohol. This interpretation is further strengthened by the two studies that showed that positive effects on behaviour were limited to successfully trained participants (Sharbanee et al., 201; Wiers et al., 2010), and by the two studies that explicitly tested and confirmed an indirect or mediating effect of training condition on behaviour through approach bias (Eberl et al. 2013; Sharbanee et al., 2015).

3.3 Cigarette smoking

All three studies that attempted to reduce cigarette smoking used the AAT. Kong et al. (2015) compared avoidance-training with sham-training over four weekly sessions in selfreported daily adolescent smokers with low nicotine dependence. They found no difference between groups in approach bias change or the number of cigarettes smoked in the week after training (smoking reduced in both groups). In contrast, Wittekind et al. (2015) compared a single online training session to a no-training control in a community sample of daily adult smokers who were moderately dependent and committed to quitting. They found that participants in the training group smoked significantly fewer cigarettes per day at four-week follow-up (they did not measure approach bias). Finally, Machulska et al. (2015) administered four sessions of avoidance or sham-training to psychiatric inpatients who were moderately dependent smokers. Participants had smoked during the past month and most were motivated to quit smoking. They observed an equal reduction in both approach bias for smoking cues and cigarette smoking in both groups after the last training session. However, at three-month follow-up, the training group reported smoking significantly fewer cigarettes per day than the control group (they did not measure approach bias at this later time). In all three studies, training led to an absolute reduction in cigarette smoking. In the two studies with moderately dependent adult smokers, the training group showed a greater reduction in cigarette smoking at four weeks (Wittekind et al., 2015) and three months (Machulska et al., 2015). Thus, the emerging evidence suggests that avoidance-training can be a useful intervention for the cessation of cigarette smoking over the long term. However, more studies are needed to corroborate this conclusion.

3.4 Eating behaviour

There were five articles that focused on eating behaviour, comprising seven studies. Five of the studies used the AAT, while one employed a Joystick Category Judgement Task (participants pulled a joystick toward them or pushed a joystick away from them in response to healthy and unhealthy food words) and one used an Implicit-Association Task (participants categorised stimuli into one of two concept categories [chocolate and non-chocolate] and one of two attribute categories [approach and avoid]). All seven studies used food stimuli: four used pictures of a range of unhealthy foods (Becker et al., 2015; Becker et al., 2015; Study 1 and Study 2; Brockmeyer et al., 2015; Fishbach and Shah, 2006), while three used only chocolate pictures (Becker et al., 2015; Becker et al., 2015, Study 3; Kemps et al., 2013; Schumacher et al., 2016). Three of the studies compared the training with sham-training (Becker, Jostmann, et al., 2015; Studies 1, 2 and 3), while three used an approach-food condition (Fishbach and Shah, 2006, Kemps et al., 2013; Schumacher et al., 2016), and one study did not have a control group (Brockmeyer et al., 2015). Most studies used a single training session (n = 6) and all used undergraduate samples.

Five out of the seven studies (all except for Becker, Jostmann, et al., 2015, Study 2, Study 3) successfully re-trained the food-related approach bias. Importantly, four of these also found a significant reduction in eating behaviour across the sample, while one study found this for successfully re-trained participants (Becker et al., 2015; Becker et al., 2015,

Study 1). Specifically, Brockmeyer et al. (2015) found that participants in the avoidancetraining group reported less intense food cravings and improved daily eating habits, while Fishbach and Shah (2006) and Becker, Jostmann, et al. (2015, Study 1) found positive effects on healthy food choice. Of the two studies that investigated chocolate, Kemps et al. (2013) found that the training group reported less intense cravings for chocolate than the control group, while Schumacher et al. (2016) found that the training group consumed less chocolate during a laboratory taste test.

The remaining two studies reported no difference in approach bias between the training and control groups (Becker, Jostmann, et al., 2015, Study 2 and Study 3). These studies also found no benefit of training on eating behaviour. Specifically, Becker, Jostmann, et al. (2015) found no difference between groups in unhealthy food choice (Study 2), and that the training group actually consumed more chocolate during a laboratory taste test (Study 3).

In sum, the studies that reported positive effects on consumption had successfully retrained approach bias toward an avoidance bias (Becker, Jostmann, et al., 2015, Study 1; Brockmeyer et al., 2015; Fishbach and Shah, 2006; Kemps et al., 2013; Schumacher et al., 2016). In contrast, the studies that did not report a positive effect on consumption also reported no difference between training and control groups on approach bias (Becker, Jostmann, et al., 2015, Study 2 and Study 3).

3.5 Summary of results

Across the domains of alcohol consumption, cigarette smoking, and eating behaviour, half of the included studies (n = 9) reported a significant difference between the training and control groups in approach bias, while the remainder reported no significant difference (n =7) or did not measure approach bias (n = 2). In addition, the majority of the studies reported positive effects on consumption that favoured the training group (n = 12); some reported no difference between groups (n = 5), and one study reported an effect in the opposite direction (Becker, Jostmann, et al., 2015, Study 3). As shown in Table 1, effect sizes (partial eta² and Cohen's d) for outcome measures varied. However, the overall conclusions drawn were the same across the three different consumption domains.

4. Discussion

The aim of the current review was to investigate the effectiveness of approach bias modification for reducing unhealthy consumption behaviours, including excessive alcohol use, cigarette smoking, and unhealthy eating. Specifically, we examined whether approach bias modification was effective at re-training approach biases for appetitive cues toward an avoidance bias, as well as subsequent effects on consumption. The main finding was that studies that successfully reduced the approach bias also changed unhealthy consumption behaviour. This appeared to be the case regardless of the number of training trials or sessions, and was evident across three different consumption domains. Thus, specific characteristics of the studies did not seem particularly important in determining the effectiveness of training.

The majority of studies included in this review found that approach bias modification was effective at changing the approach bias toward an avoidance bias for appetitive cues. This was particularly the case for alcohol and unhealthy food cues; as yet there is insufficient information on cigarette smoking. This finding suggests that approach biases for appetitive substances beyond alcohol (Wiers, Gladwin, et al., 2013) can be successfully modified with the Approach-Avoidance Task.

More importantly, the majority of studies reported positive effects of approach bias re-training on consumption. In particular, studies targeting alcohol consumption found that successfully re-trained participants consumed less alcohol during a laboratory taste test and alcohol-dependent inpatients had lower relapse rates one year after treatment than those in the control group. Similarly, participants trained to avoid smoking-related cues reported smoking fewer cigarettes over the longer-term. Finally, several studies from the eating domain found that modifying an approach bias for unhealthy food cues, including chocolate, encouraged healthier snack food choice and reduced food cravings and consumption in the laboratory. These findings extend the conclusions drawn by Wiers, Gladwin, et al. (2013) to show that approach bias modification is effective for reducing unhealthy consumption behaviours other than alcohol use, in particular, cigarette smoking and unhealthy eating.

Further, the current review was able to provide some insight into the main difference between the studies that reported positive effects on consumption behaviour (i.e., those that favoured the training group), and those that did not. It appears that avoidance-training was effective at reducing unhealthy consumption when the approach bias for appetitive cues was successfully changed toward an avoidance bias for such cues. For alcohol and tobacco, this was more apparent in samples with greater levels of substance dependence than in unselected samples. Nevertheless, the main conclusion of this review suggests that the mechanism underlying behaviour change is a reduction in approach bias, consistent with the theoretical predictions of contemporary dual-process models (Macleod et al., 2002). The finding that the effect of training condition on consumption was mediated by approach bias (Eberl et al., 2013) further supports this suggestion. Future studies may aim to replicate this finding. Importantly, the overall conclusion appeared to hold across a range of outcomes in three different domains, suggesting that the theoretical predictions of dual-process models apply to a number of appetitive substances.

The results of the current review have a number of practical and clinical implications. First, we found that effective re-training of the approach bias was what distinguished studies that successfully reduced consumption from those that did not; specific study characteristics (e.g., the number of training trials) did not affect outcomes. Nevertheless, it would be useful for future studies to determine the minimum number of trials needed to change approach bias and consumption. Second, the review showed that approach bias modification is successful across several different appetitive behaviours, including alcohol use, cigarette smoking, and eating behaviour. As suggested here and by dual-process models (Strack & Deutsch, 2004), if the mechanism underlying behaviour change is reduction in approach bias, then logically approach bias modification should be effective for other appetitive behaviours. Thus, future research should investigate the effectiveness of using approach bias modification for other substances, including cannabis (Field et al., 2006) and heroin (Zhou et al., 2012), for which approach biases have previously been documented.

While the majority of the reviewed studies were conducted in the laboratory, some studies delivered the training online. The delivery mode was not specific to the consumption domain and made no difference to the effectiveness of training on outcomes. However, future research could examine the optimal delivery mode for obtaining a positive effect on approach bias and behaviour and also determine which is the preferred delivery mode for participants. There may well be novel ways to implement effective training protocols, for example, via a smartphone app. Another possible avenue for future research would be the combined effect of avoidance training for unhealthy substances with approach training for healthy substances. This may provide additional benefit, as a recent study in the anxiety domain found that participants trained to approach positive cues showed a reduction in anxiety symptoms (Becker, Jostmann, et al., 2015). Using a similar 'positivity training' technique for consumption, participants could be simultaneously trained to approach a healthy substance, such as water or healthy food, and avoid an unhealthy substance, such as alcohol or chocolate. Interventions that not only discourage unhealthy choices but also promote healthy choices are likely to be more acceptable, an important consideration for cognitive training tasks (Wiers, Gladwin, et al., 2013).

The results of the current literature review are encouraging in terms of illustrating the effectiveness of approach bias modification for consumption behaviour. However, the body

of literature reviewed has some limitations as the studies used a range of different outcome measures, making direct comparison across studies difficult. Nevertheless, this did not appear to have any effect on the observed effectiveness of training on consumption behaviour, and so strengthens the overall conclusions drawn. In addition, most of the studies did not examine the longer-term effects of approach bias modification on either approach bias or consumption behaviour, as only four out of the 18 studies reported any follow-up data (Eberl et al., 2013; Eberl et al., 2014; Machulska et al., 2015; Wiers et al., 2011). The current review also has some limitations. The review included 15 different publications (comprising 18 individual studies) that fulfilled the selection criteria. However, as an emerging area, there may be a number of studies under review or in progress that have not been included. In addition, all intervention studies are subject to publication bias (Scherer, Langenberg, & Elm, 2005).

In conclusion, the findings of the current literature review suggest that approach bias modification is effective at re-training approach biases for several appetitive cues. The main finding was that this protocol is also effective at reducing unhealthy consumption behaviour, but seemingly only when the approach bias has been successfully modified. Future research might investigate whether approach bias re-training is useful for other unhealthy consumption behaviours, whether training people to approach healthy appetitive cues is (more) effective, and whether training effects persist longer-term. Determining the clinical utility of approach bias modification is particularly important for the treatment of individuals who engage in harmful behaviours such as excessive alcohol consumption, smoking, and unhealthy eating.

Appendix A. Search strategy

- PsycInfo: (approach avoidance or approach bias) and (modification or training)

- ScienceDirect: pub-date > 2005 and "approach bias" modification OR "approach

avoidance" training AND "eating behaviour" OR smoking OR alcohol

[Journals(Psychology)].

Google Scholar: "approach bias" modification OR "approach avoidance" training AND
 "eating behaviour" OR smoking OR alcohol

Scopus: TITLE-ABS-KEY (approach bias modification OR approach avoidance training)
 AND PUBYEAR AFT 2005; Search limits: English

References¹

- Amanzadeh, B., Sokal-Gutierrez, K., & Barker, J. C. (2015). An interpretive study of food, snack and beverage advertisements in rural and urban El Salvador. *BMC public health*, 15(1), 1-11.
- Beard, C., Sawyer, A. T., & Hofmann, S. G. (2012). Efficacy of attention bias modification using threat and appetitive stimuli: A meta-analytic review. *Behavior Therapy*, 43, 724–740.
- Becker, E. S., Ferentzi, H., Ferrari, G., Möbius, M., Brugman, S., Custers, J., ... Rinck, M. (2015). Always approach the bright side of life: A general positivity training reduces stress reactions in vulnerable individuals. *Cognitive Therapy and Research*. In press.
- *Becker, D., Jostmann, N. B., Wiers, R. W., & Holland, R. W. (2015). Approach avoidance training in the eating domain: Testing the effectiveness across three single session studies. *Appetite*, *85*, 58–65.
- Bestman, A., Thomas, S. L., Randle, M., & Thomas, S. D. (2015). Children's implicit recall of junk food, alcohol and gambling sponsorship in Australian sport. *BMC Public Health*, 15(1), 1-9.
- Bradley, B. P., Field, M., Healy, H., & Mogg, K. (2008). Do the affective properties of smoking-related cues influence attentional and approach biases in cigarette smokers? *Journal of Psychopharmacology*, 22, 737-745.
- *Brockmeyer, T., Hahn, C., Reetz, C., Schmidt, U., & Friederich, H.-C. (2015). Approach bias modification in food craving: A proof-of-concept study. *European Eating Disorders Review*, 23, 352–360.

¹ * means included in literature review

- Cohen, D. A., & Babey, S. H. (2012). Contextual influences on eating behaviours: Heuristic processing and dietary choices. *Obesity Reviews*, *13*, 766-779.
- Cousijn, J., Goudriaan, A. E., Ridderinkhof, K. R., van den Brink, W., Veltman, D. J., & Wiers, R. W. (2012). Approach-bias predicts development of cannabis problem severity in heavy cannabis users: results from a prospective FMRI study. *PloS One*, *7*, e42394.
- Cousijn, J., Goudriaan, A. E., & Wiers, R. W. (2011). Reaching out towards cannabis: approach-bias in heavy cannabis users predicts changes in cannabis use. *Addiction*, 106, 1667-1674.
- Field, M., Eastwood, B., Bradley, B. P., & Mogg, K. (2006). Selective processing of cannabis cues in regular cannabis users. *Drug and alcohol dependence*, 85, 75-82.
- *Eberl, C., Wiers, R. W., Pawelczack, S., Rinck, M., Becker, E. S., & Lindenmeyer, J.
 (2013). Approach bias modification in alcohol dependence: Do clinical effects replicate and for whom does it work best? *Developmental Cognitive Neuroscience*, 4, 38–51.
- *Eberl, C., Wiers, R. W., Pawelczack, S., Rinck, M., Becker, E. S., & Lindenmeyer, J. (2014). Implementation of approach bias re-training in alcoholism: How many sessions are needed? *Alcoholism: Clinical and Experimental Research*, *38*, 587–594.
- Field, M., Eastwood, B., Bradley, B. P., & Mogg, K. (2006). Selective processing of cannabis cues in regular cannabis users. *Drug and Alcohol Dependence*, 85, 75-82.
- *Fishbach, A., & Shah, J. Y. (2006). Self-control in action: Implicit dispositions toward goals and away from temptations. *Journal of Personality and Social Psychology*, 90, 820– 832.
- Havermans, R. C., Giesen, J. C., Houben, K., & Jansen, A. (2011). Weight, gender, and snack appeal. *Eating Behaviors*, *12*, 126-130.

- Hofmann, W., Friese, M., & Wiers, R. W. (2008). Impulsive versus reflective influences on health behavior: A theoretical framework and empirical review. *Health Psychology Review*, 2, 111-137.
- *Lindgren, K. P., Wiers, R. W., Teachman, B. A., Gasser, M. L., Westgate, E. C., Cousijn, J., ... Neighbors, C. (2015). Attempted training of alcohol approach and drinking identity associations in US undergraduate drinkers: Null results from two studies. *PloS One*, 10, e0134642.
- Jones, A., Di Lemma, L., Robinson, E., Christiansen, P., Nolan, S., Tudur-Smith, C., & Field, M. (2016). Inhibitory control training for appetitive behaviour change: A metaanalytic investigation of mechanisms of action and moderators of effectiveness. *Appetite*, 97, 16-28.
- Kakoschke, N., Kemps, E., & Tiggemann, M. (2015). Combined effects of cognitive bias for food cues and poor inhibitory control on unhealthy food intake. *Appetite*, *87*, 358-364.
- Kemps, E., & Tiggemann, M. (2015). Approach bias for food cues in obese individuals. *Psychology & Health, 30,* 370-380.
- Kemps, E., Tiggemann, M., & Hollitt, S. (2014). Exposure to television food advertising primes food-related cognitions and triggers motivation to eat. *Psychology & health*, 29, 1192-1205.
- *Kemps, E., Tiggemann, M., Martin, R., & Elliott, M. (2013). Implicit approach–avoidance associations for craved food cues. *Journal of Experimental Psychology: Applied*, 19, 30-38.
- *Kong, G., Larsen, H., Cavallo, D. A., Becker, D., Cousijn, J., Salemink, E., ... & Krishnan-Sarin, S. (2015). Re-training automatic action tendencies to approach cigarettes among adolescent smokers: A pilot study. *The American Journal of Drug and Alcohol Abuse*, 41, 425-432.

- Koster, E. H., Fox, E., & MacLeod, C. (2009). Introduction to the special section on cognitive bias modification in emotional disorders. *Journal of Abnormal Psychology*, *118*, 1-4.
- *Machulska, A., Zlomuzica, A., Rinck, M., Assion, H. J., & Margraf, J. (2015). Approach bias modification in inpatient psychiatric smokers. *Journal of Psychiatric Research*. In press.
- MacLeod, C., & Mathews, A. (2012). Cognitive bias modification approaches to anxiety. *Annual Review of Clinical Psychology*, *8*, 189-217.
- MacLeod, C., Rutherford, E., Campbell, L., Ebsworthy, G., & Holker, L. (2002). Selective attention and emotional vulnerability: Assessing the causal basis of their association through the experimental manipulation of attentional bias. *Journal of Abnormal Psychology*, 111, 107–123.
- Marteau, T. M., Hollands, G. J., & Fletcher, P. C. (2012). Changing human behavior to prevent disease: The importance of targeting automatic processes. Science, 337 1492-1495.
- Moher, D., Liberati, A., Tetzlaff, J., Altman, D. G., The PRISMA Group (2009). *Preferred Reporting Items for Systematic Reviews and Meta-Analyses: The PRISMA Statement. PLoS Med 6*: e1000097.
- Nederkoorn, C., Houben, K., Hofmann, W., Roefs, A., & Jansen, A. (2010). Control yourself or just eat what you like? Weight gain over a year is predicted by an interactive effect of response inhibition and implicit preference for snack foods. *Health Psychology*, 29, 389–393.
- Neumann, R., & Strack, F. (2000). Approach and avoidance: the influence of proprioceptive and exteroceptive cues on encoding of affective information. *Journal of Personality and Social Psychology*, *79*, 39-48.

- Rinck, M., & Becker, E. S. (2007). Approach and avoidance in fear of spiders. *Journal of Behavior Therapy and Experimental Psychiatry*, 38, 105–120.
- Scherer, R. W., Langenberg, P., & Elm, E. (2005). Full publication of results initially presented in abstracts. *The Cochrane Library*.
- *Schumacher, S. E., Kemps, E., & Tiggemann, M. (2016). Bias modification training can alter approach bias and chocolate consumption. *Appetite*, *96*, 219–224.
- *Sharbanee, J. M., Hu, L., Stritzke, W. G. K., Wiers, R. W., Rinck, M., & MacLeod, C. (2014). The effect of approach/avoidance training on alcohol consumption is mediated by change in alcohol action tendency. *PLoS One*, *9*, e85855.
- Strack, F., & Deutsch, R. (2004). Reflective and impulsive determinants of social behavior. Personality and Social Psychology Review, 8, 220–247.
- Veenstra, E. M., & de Jong, P. J. (2010). Restrained eaters show enhanced automatic approach tendencies towards food. *Appetite*, *55*, 30-36.
- Watson, P., de Wit, S., Cousijn, J., Hommel, B., & Wiers, R.W. (2013). Motivational mechanisms underlying the approach bias to cigarettes. *Journal of Experimental Psychopathology*, 4, 250–262.
- * Wiers, R. W., Eberl, C., Rinck, M., Becker, E. S., & Lindenmeyer, J. (2011). Retraining automatic action tendencies changes alcoholic patients' approach bias for alcohol and improves treatment outcome. *Psychological Science*, 22, 490–497.
- Wiers, R. W., Rinck, M., Dictus, M., & Van den Wildenberg, E. (2009). Relatively strong automatic appetitive action-tendencies in male carriers of the OPRM1 G-allele. *Genes, Brain and Behavior*, 8, 101-106.
- *Wiers, R. W., Rinck, M., Kordts, R., Houben, K., & Strack, F. (2010). Retraining automatic action-tendencies to approach alcohol in hazardous drinkers. *Addiction*, *105*, 279–287.

Wiers, R. W., Gladwin, T. E., Hofmann, W., Salemink, E., & Ridderinkhof, K. R. (2013).
Cognitive bias modification and cognitive control training in addiction and related psychopathology mechanisms, clinical perspectives, and ways forward. *Clinical Psychological Science*, 1, 192-212.

- *Wiers, R. W., Houben, K., Fadardi, J. S., van Beek, P., Rhemtulla, M., & Cox, W. M. (2015). Alcohol cognitive bias modification training for problem drinkers over the web. *Addictive Behaviors*, 40, 21–26.
- Wiers, C. E., Kühn, S., Javadi, A. H., Korucuoglu, O., Wiers, R. W., Walter, H., ... & Bermpohl, F. (2013). Automatic approach bias towards smoking cues is present in smokers but not in ex-smokers. *Psychopharmacology*, 229, 187-197.
- *Wittekind, C. E., Feist, A., Schneider, B. C., Moritz, S., & Fritzsche, A. (2015). The approach-avoidance task as an online intervention in cigarette smoking: A pilot study. *Journal of Behavior Therapy and Experimental Psychiatry*, 46, 115–120.
- World Health Organization [WHO] (2016). World Health Statistics 2016: Monitoring health for the SDGs. Retrieved from http://www.who.int
- Zhou, Y., Li, X., Zhang, M., Zhang, F., Zhu, C., & Shen, M. (2012). Behavioural approach tendencies to heroin-related stimuli in abstinent heroin abusers. *Psychopharmacology*, 221, 171-176.
- Zimmerman, F. J., & Shimoga, S. V. (2014). The effects of food advertising and cognitive load on food choices. *BMC Public Health*, 14, 1-10.

CHAPTER 6: STUDY 4

The Effect of Combined Avoidance and Control Training on Implicit Food Evaluation and Choice

Naomi Kakoschke, Eva Kemps, and Marika Tiggemann School of Psychology, Flinders University, South Australia

Corresponding author: Naomi Kakoschke, School of Psychology, Flinders University, GPO Box 2100, Adelaide, 5001, South Australia. Email: naomi.kakoschke@flinders.edu.au **Funding:** This research was supported under the Australian Research Council's Discovery Project funding scheme (project number DP130103092).

Statement of co-authorship: All authors were involved in the formulation of the study concept and design. Naomi Kakoschke collected the data, and completed the data analysis and the initial draft of the manuscript. Eva Kemps and Marika Tiggemann edited multiple revisions of the manuscript.

Acknowledgements: We are grateful to Paul Douglas for developing the software for the computer administered tasks.

This manuscript is currently under revision.

Abstract

Background: Continual exposure to food cues in the environment contributes to unhealthy eating behaviour. According to dual-process models, such behaviour is partly determined by automatic processing of unhealthy food cues (e.g., approach bias), which fails to be regulated by controlled processing (e.g., inhibitory control). The current study aimed to investigate the effect of combined avoidance and control training on implicit evaluation (liking), choice, and consumption of unhealthy snack food. Method: Participants were 240 undergraduate women who were randomly allocated to one of four experimental conditions of a 2 (avoidance training: training versus control) x 2 (control training: training versus control) betweensubjects design. Results: The combined training group had a more negative implicit evaluation of unhealthy food than either of the two training conditions alone or the control condition. In addition, participants trained to avoid unhealthy food cues subsequently made fewer unhealthy snack food choices. No significant group differences were found for food intake. Limitations: Participants were women generally of a healthy weight. Overweight or obese individuals may derive greater benefit from combined training. Conclusions: Results lend support to the theoretical predictions of dual-process models, as the combined training reduced implicit liking of unhealthy food. At a practical level, the findings have implications for the effectiveness of interventions targeting unhealthy eating behaviour.

1. Introduction

The increasing worldwide prevalence of overweight and obesity during the last few decades has become a primary health concern. In contemporary Western societies such as Australia, 64% of adults are now classified as overweight and 29% as obese (WHO, 2014). A major contributing factor to these high rates of overweight and obesity is unhealthy eating behaviour, in particular, the over-consumption of foods high in fat, salt, and sugar (WHO, 2014). One potential influence on unhealthy eating is exposure to a vast array of visual cues associated with food through advertising on the internet, billboards, magazines, and television (Havermans, 2013; Polivy, Herman & Coelho, 2008). Over time, exposure to unhealthy food cues can lead to biased automatic processing of such cues, which can translate into increased food intake if automatic responses to these cues are not inhibited (Cohen & Farley, 2008).

Contemporary dual-process models have been prominent in understanding why our health-related behaviours are not always consistent with long-term goals, such as weight loss (Hoffman, Friese, & Wiers, 2008; Strack & Deutsch, 2004). One of the key predictions is that behaviour is determined by a combination of automatic and controlled processing. Automatic processing is fast, effortless and associative. One such automatic process is an approach bias, which is the automatic action tendency to approach rather than avoid relevant cues in the environment (Wiers et al., 2013). In contrast, controlled processing is slow, controlled and conscious. One aspect of controlled processing is inhibitory control, which is 'the ability to inhibit a behavioural impulse in order to attain higher-order goals' (Houben, Nederkoorn, & Jansen, 2012, p. 550). Taken together, it may be that a rewarding stimulus (e.g., a slice of chocolate cake) in the environment elicits an automatic response, such as an approach action tendency, which can predict unhealthy choice or intake if this process occurs too quickly and effortlessly to be regulated by the slower controlled processing system.

Approach bias has been demonstrated for a range of appetitive substances, including alcohol (Wiers, Rinck, Kordts, Houben, & Strack, 2010), cigarettes (Wiers et al., 2013) and cannabis (Cousijn, Goudriaan, & Wiers, 2011). Importantly, approach bias has also been associated with increased consumption of some of these substances (alcohol: Wiers, Rinck, Dictus, & Van Den Wildenberg, 2009; Wiers, et al., 2010, cannabis: Cousijn, et al., 2011). In the eating domain, approach bias has been reliably demonstrated for a variety of unhealthy foods (Brignell, Griffiths, Bradley, & Mogg, 2009; Havermans, Giesen, Houben, & Jansen, 2011; Kemps & Tiggemann, 2015; Kemps, Tiggemann, Martin, & Elliott, 2013; Veenstra & de Jong; 2010), and has been associated with increased unhealthy food intake (Kakoschke, Kemps, & Tiggemann, 2015; Nederkoorn, Houben, Hofmann, Roefs, & Jansen, 2010). Thus, the evidence suggests that approach bias contributes to consumption of appetitive substances.

Furthermore, automatic processes underlying unhealthy behaviour can be manipulated using a computerised cognitive training paradigm. This is achieved in commonly used protocols such as the Approach-Avoidance Task (AAT), by presenting target pictures (e.g., unhealthy food) in a format that requires a push (i.e., avoidance) response and control pictures (e.g., animals) in a format that requires a pull (i.e., approach) response on the majority of trials. In the alcohol domain, the AAT has been used to train an avoidance of alcohol cues, which reduced implicit liking of such cues and subsequent beer consumption (Wiers et al., 2010). Importantly, these findings were extended to alcohol-dependent inpatients, whereby the training reduced relapse rates one year later (Wiers, Eberl, Rinck, Becker, & Lindenmeyer, 2011; Eberl et al., 2013).

Researchers in the eating domain have also begun to use the AAT. In an early study, Fishbach and Shah (2006) trained participants to avoid unhealthy food words (e.g., 'cookie') and approach healthy food words (e.g., 'apple'), which translated into healthier snack food choices. More recently, Brockmeyer, Hahn, Reetz, Schmidt and Friederich (2015) successfully re-trained approach bias for unhealthy food, which reduced food cravings. Similarly, Becker, Jostmann, Wiers and Holland (2015, Study 1) found that participants who were successfully trained to avoid unhealthy food were more likely to choose a healthy snack. Schumacher, Kemps and Tiggemann (2016) found that participants trained to avoid chocolate cues ate less chocolate, while Dickson, Kavanagh and MacLeod (2016) found no difference in chocolate consumption between groups. Finally, Becker et al. (2015, Study 3) found that their training group actually ate more chocolate than the control group. Thus, research shows that approach bias for unhealthy food can be reduced, but the effect on eating behaviour is less consistent.

In terms of controlled processing, poor inhibitory control has been linked to several unhealthy behaviours. For example, research on alcohol has shown that for participants with low inhibitory control, positive implicit evaluations for alcohol predicted increased alcohol consumption (Friese & Hofmann, 2009; Friese, Hofmann, & Wänke, 2008; Houben & Wiers, 2009; Thush et al., 2008). In the eating domain, studies have consistently shown that poor inhibitory control predicts unhealthy eating behaviour, such as increased unhealthy snack food choice (Jasinska et al., 2012) and intake (Appelhans et al., 2011; Guerrieri et al., 2007).

Inhibitory control can be increased using tasks that involve pairing appetitive stimuli with a no-go cue (Veling, Holland, & van Knippenberg, 2008) or stop-signal (Verbruggen & Logan, 2009). In the alcohol domain, inhibitory control training reduced implicit liking and consumption of alcohol (Houben et al., 2012; Houben, Nederkoorn, Wiers & Jansen, 2011; Jones & Field, 2013, Study 1). Similarly, in the eating domain, inhibitory control training reduced chocolate intake (Houben & Jansen, 2011), as well as implicit liking (Houben & Jansen, 2015; Lawrence et al., 2015b; Veling, Holland, & Knippenberg, 2008; Veling, Aarts, & Stroebe, 2013a), choice (Veling, Aarts, & Stroebe, 2013b; Veling, Stroebe, & Aarts, 2014), and unhealthy food intake (Lawrence, Verbruggen, Morrison, Adams, & Chambers, 2015a).

Although inhibitory control training appears to be an effective technique for changing eating behaviour, a recent meta-analysis found that the effect size on consumption is small, leading the authors to suggest that it may be useful to supplement inhibitory control training with another type of intervention (Allom, Mullan, & Hagger, 2015). Two recent studies tested the combination of inhibitory control training with implementation intention training, which aims to improve eating behaviour by reminding people of their dieting goal. One study found that participants who received both interventions lost more weight over four weeks than those who received either one alone (Veling, van Koningsbruggen, Aarts & Stroebe, 2014), while the other found that the combined training was no more beneficial than either training task alone at reducing the amount of sweets participants selected (Koningsbruggen, Veling, Stroebe, & Aarts, 2014).

According to dual-process models, it should be possible to change unhealthy eating behaviour by re-training either automatic (e.g., approach bias) or controlled processing (e.g., inhibitory control). To date, these two types of interventions have been used individually with mixed success. However, the key prediction of dual-process models is that training automatic and controlled processing together should be more effective at changing behaviour. There is some evidence to support this suggestion, as one correlational study has shown that women who had a stronger approach bias for unhealthy food cues combined with lower inhibitory control consumed more unhealthy snack food during a taste test (Kakoschke et al., 2015).

Thus, the current study aimed to determine whether combining avoidance training with inhibitory control training was more effective than either training task alone at reducing implicit liking, choice and intake of unhealthy food. It was predicted that participants trained to avoid unhealthy food cues and inhibit responses to such cues would show reduced implicit liking of unhealthy food, eat less food in a taste test, and be less likely to choose an unhealthy snack than those who received either training alone or those in the control group.

2. Material and Methods

2.1. Participants

The sample consisted of 240 women, aged 18 to 25 years (M = 20.61, SD = 2.43), recruited from the undergraduate student population at Flinders University. The majority were within the healthy weight range (18.5-24.9 kg/m²) with a mean BMI of 22.91 (SD = 4.90). Only women were recruited as they have shown a greater tendency to overeat (Burton, Smit, & Lightowler, 2007) and greater concern for weight and dieting goals (Keel, Baxter, Heatherton, & Joiner, 2007). Participants were included if they could speak English fluently, liked most foods, and did not have any food allergies, intolerances, or special dietary requirements. Participants were instructed to eat something two hours before their scheduled testing session to ensure that they were not hungry, as hunger has been associated with both a cognitive bias for unhealthy food cues (Mogg, Bradley, Hyare, & Lee, 1998; Seibt, Hafner, & Deutsch, 2007) and lower inhibitory control (Nederkoorn, Guerrieri, Havermans, Roefs, & Jansen, 2009). All participants reported having complied with this instruction, and subjective hunger ratings (100 mm visual analogue scale ranging from 'not hungry at all' to 'extremely hungry'; Grand, 1968), fell slightly below the mid-point of the scale (M = 47.02, SD = 24.37). 2.2. Design

Participants were randomly allocated to one of four experimental conditions of a 2 (avoidance training: training vs. control) x 2 (control training: training vs. control) betweensubjects design. The researcher was blind to condition during testing as participants were assigned a randomised login code for all computerised tasks. The dependent measures were implicit unhealthy food evaluation, consumption during the taste test, and first snack choice. *2.3. Materials*

2.3.1. Stimuli

The stimuli were 40 digital coloured photographs (presented in a resolution of approximately 1024 x 768 pixels) comprising 20 food and 20 control (animal) pictures, a subset of those used in a previous study (Kakoschke et al., 2015). The pictures of unhealthy food depicted food items high in sugar, salt and/or fat (e.g., chocolate, potato chips). Animals were chosen for the non-food comparison category of stimuli as they are, like food, overall appealing. Animal pictures depicted species that are not commonly consumed in Western society (e.g., koala, giraffe). The unhealthy food pictures and the animal pictures were matched on characteristics such as quality, brightness, and size, as well as ratings of pleasure and arousal. These ratings were obtained from a previously conducted pilot study, in which 21 women aged 17-45 years (M = 23.67, SD = 8.28) rated 590 pictures of food and animals on 9-point pleasure and arousal scales (Kemps, Tiggemann, & Hollitt, 2014). The same stimulus set was used for both the modified Approach-Avoidance and the Go/No-Go task. 2.3.2. Avoidance training with a modified AAT

Following Wiers et al. (2011), an approach-avoidance task was used to induce an avoidance tendency away from unhealthy food. The task consisted of two blocks (100 trials each) with a break half-way. Participants began each trial by pressing the start button on top of a joystick. An unhealthy food or animal picture then appeared in the centre of the screen. Participants were instructed to push or pull the joystick according to whether the picture was presented in portrait or landscape format. Instructions were counterbalanced (i.e., half of the participants pulled the joystick for portrait format and pushed the joystick for landscape and vice versa). Pulling the joystick increased the picture size (simulating approaching), while pushing the joystick decreased it (simulating avoidance). The picture disappeared once the participant pulled or pushed the joystick so that the picture was at its largest or smallest size,

respectively. Participants were asked to respond as quickly and accurately as possible. Pictures were presented in a different random order for each participant.

In the training condition, participants were trained to avoid pictures of unhealthy food. Specifically, unhealthy food pictures were presented in pull-format (approach) on 10% of trials and in push-format (avoid) on 90% of trials, with reversed contingencies for animal pictures. In the control condition, all of the pictures were presented in pull-format on 50% of trials and in push-format on 50% of trials.

For each participant, reaction time scores for correct responses were calculated for pulling and pushing the unhealthy food and animal pictures. An approach-avoidance score was calculated as the difference between median pushing and pulling reaction times for unhealthy food pictures (Wiers et al., 2011). Positive scores indicate an approach action tendency toward food (i.e., faster pull than push), whereas negative scores indicate an avoidance tendency away from food (i.e., faster push than pull).

2.3.3. Control training with a modified GNG task

Following Houben and Jansen (2011), a food specific version of the go/no-go (GNG) task was used to increase inhibitory control for unhealthy food cues. The task consisted of two blocks (160 trials each) with a break half-way. On each trial, a picture was presented together with a go or a no-go cue (the letters 'P' or 'F') for 1500 ms. Participants were instructed to press the space bar as quickly and accurately as possible when a go cue was presented together with the picture, and to refrain from pressing the space bar when a no-go cue was presented. Instructions regarding letter type ('F' versus 'P') and response assignment (go versus no-go) were counterbalanced. The go/no-go cues were displayed in black font type and presented in the corner of each picture (corner location was counterbalanced). Pictures were presented in a different random order for each participant.

In the training condition, participants were trained to inhibit responses to unhealthy food. Specifically, food pictures were paired with a no-go cue on 90% of trials and a go-cue on 10% of trials, with reversed contingencies for animal pictures. In the control condition, all pictures were paired with a go cue on 50% of trials and a no-go cue on 50% of trials.

Performance on the GNG task was assessed by calculating reaction times (i.e., correct responses on go trials) and the number of commission errors (i.e., incorrect responses on nogo trials) for unhealthy food cues. Faster reaction times and a higher number of commission errors are typically considered indicative of poor inhibitory control (Baker, Taylor & Leyva, 1995; Bezdjian, Baker, Lozano, & Raine, 2009; Halperin, Wolf, Greenblatt, & Young, 1991). *2.3.4. Implicit food evaluation*

Implicit evaluations of unhealthy food were assessed using a recoding-free (RF; Rothermund, Teige-Mocigemba, Gast, & Wentura, 2009) variant of the single-category implicit association task (SC-IAT; Karpinski & Steinman, 2006). In the IAT-RF, the response key assignment to the target category switches between trials rather than blocks, which has been shown to be a more valid assessment of implicit evaluations than the traditional IAT (Houben, Rothermund, & Wiers, 2009). Specifically, participants respond to the target category on the basis of semantic membership rather than salience (Rothermund et al., 2009).

The target category was unhealthy food and the evaluative categories were 'I like' and 'I dislike' (Olson & Fazio, 2004). These evaluative category labels were used to reduce the effects of 'extra-personal associations' (Olson & Fazio, 2004) and have been used in previous food and eating research (Nederkoorn et al., 2010). Participants were asked to categorise pictures (presented one at a time in the centre of the computer screen) into one of three categories: unhealthy food (target), positive (evaluative), or negative (evaluative). The target stimuli were six pictures of unhealthy food (i.e., cake, chocolate, ice-cream, chips, hamburger, pizza), and following Houben et al. (2012), were a subset of those used in the

AAT and GNG tasks. Evaluative stimuli were six positive and six negative pictures (unrelated to food) selected from the IAPS¹ (Lang, Bradley & Cuthbert, 2001). These pictures have been used in previous research (e.g., Nederkoorn et al., 2010).

The SC-IAT consisted of three blocks. In the first block, participants were asked to categorise the positive and negative pictures into the evaluative categories of 'I like' and 'I dislike' using two response keys (i.e., left = 'E' and right = 'I'). This block consisted of 24 trials with the six positive and the six negative pictures each presented twice. In the second block, participants were instructed to categorise pictures into both evaluative categories (i.e., positive and negative) and the target category (i.e., unhealthy food). This block consisted of 36 trials with each positive, negative and unhealthy food picture presented twice. The third block was the same as the second; however, the number of trials increased to 144 with each positive, negative, and food picture presented eight times. Participants were given a short break at half way. The response key (i.e., left or right) assigned to the evaluative categories remained constant throughout the task, and was counterbalanced between participants. However, the response key assigned to the target category switched randomly between trials, so that unhealthy food pictures shared a response key with the positive category on half of the trials and with the negative category on the other half.

Following standard protocols (e.g., Nederkoorn et al., 2010), scores on the IAT were calculated using the D600-algorithm (Greenwald, Nosek, & Banaji, 2003) and modified for application to SC-IAT (Karpinski & Steinman, 2006). Incorrect responses were replaced by 'mean (correct) + 600 ms'. Mean response times on food paired with positive trials were subtracted from mean response times on food paired with negative trials. The difference between response times on these two trial types was divided by the standard deviation of all correct responses on food trials, with higher scores indicative of a more positive implicit

¹ IAPS picture numbers: 1300, 1603, 2070, 2550, 5480, 5623, 6550, 6570, 8200, 9220, 9340, 9600

evaluation of unhealthy food. Response times below 300ms and above 3000ms were removed as these times are regarded as anticipatory or delayed, respectively (Palfai & Ostafin, 2003).

2.3.5. Food intake

Food intake was measured using a so-called taste test. Participants were presented with a platter comprising four snacks (M&Ms, chocolate-chip biscuits, potato chips, and pretzels). These snacks were chosen as they are commonly consumed and are bite-sized to facilitate eating. The snack foods were presented in four equally filled separate bowls. The presentation order of the bowls was counterbalanced across participants using a 4×4 Latin square. Participants were instructed to taste and rate each snack on several dimensions (e.g., flavour, likelihood of purchase). They were told that they could try as much of the food as they liked and were given 10 minutes to complete their ratings. The amount consumed of each food was calculated by subtracting the weight (in grams) of the snacks after the taste test from the weight before the taste test. The weight in grams for each food was then converted into the number of calories consumed and summed to obtain a total measure of food intake. *2.3.6. Food choice*

Following Veling, Aarts, and Stroebe (2013), we used a touch-screen food-choice task to assess participants' selection of foods that they would like to take home. Participants were asked to select eight of the 16 foods presented to them by touching the picture of the food on the screen. However, the first food chosen was used as an indicator of snack choice to ensure that an 'automatic' decision was captured (Furst, Connors, Bisogni, Sobal, & Falk, 1996). A touch-screen was used to represent reaching out to select a snack, similar to using buttons on a vending machine. The food pictures comprised eight healthy and eight unhealthy snacks, which were arrayed in a 4 x 4 square grid on a computer touch-screen. Participants were given a time limit of 15 seconds to make their selections as food choices are often made under time pressure.

2.4. Procedure

Participants were recruited for a study entitled 'Food Preferences and Eating Habits in Women' using flyers distributed around campus, and a listing on the School of Psychology's online study participation system. The testing session lasted approximately 60 minutes and was conducted in a quiet laboratory. After giving informed consent, participants provided background information. They then completed the Approach-Avoidance and the Go/No-Go tasks in counterbalanced order in the combined training and the control only conditions. In the two conditions involving both a control and a training task, the control task was always completed before the training task so as to not reduce the effect of the training on subsequent outcome measures. Participants subsequently completed the implicit association task, a taste test (to assess immediate consumption), and the food choice task. The study was approved by The University's Social and Behavioural Research Ethics Committee.

3. Results

3.1. Statistical considerations

Following standard protocols (e.g., Wiers et al., 2010), incorrect trials (5.28%) were excluded and approach bias scores that were more than 3 *SD* from the mean were identified as outliers and were changed to plus or minus one unit from the first non-outlier (Tabachnick & Fidell, 1989). An alpha value of .05 was used to determine significant *p* values. Effect size measures were partial eta² (η^2) for ANOVA and Cohen's *d* for t-tests. For η^2 , a value of .01 represents a small effect, .06, a medium effect, and .14, a large effect, while for Cohen's *d*, .20 represents a small effect, .50, a medium effect, and .80, a large effect (Cohen, 1992).

3.2. Manipulation check

3.2.1. Approach-avoidance task

An independent samples t-test revealed a significant group difference in approachavoidance scores, t(214.98) = -11.69, p < .001, d = .15. Participants in the avoidance training condition showed an avoidance tendency away from unhealthy food cues (M = -128.90, SD = 111.12), while those in the control condition showed an approach tendency toward such cues (M = 18.86, SD = 80.29).

3.2.2. Go/no-go task

An independent samples t-test revealed a significant group difference in response times on go trials, t(238) = 4.11, p < .001, d = .53. Participants in the control training condition were significantly slower to respond to unhealthy food cues (M = 621.71, SD =84.79) than those in the control condition (M = 579.31, SD = 74.74). Although means were in the predicted direction, there was no difference in the number of commission errors between the training (M = 1.89, SD = 2.32) and control group (M = 1.47, SD = 2.01), t(238) =1.06, p = .290, but participants made very few commission errors overall (1.57%).

3.3. Effect of training on implicit food evaluation

A 2 (AAT: training versus control) x 2 (GNG: training versus control) ANOVA was used to examine the effects of training on implicit food evaluation. Neither the main effect of AAT, F(1, 230) = .778, p = .379, $\eta^2 = .003$, nor GNG condition, F(1, 230) = .780, p = .378, $\eta^2 = .003$, was significant. However, as predicted, there was a significant interaction between AAT and GNG condition, F(1, 233) = 4.76, p = .030, $\eta^2 = .02$. As can be seen in Table 1, the combined training group showed a more negative implicit evaluation of unhealthy food than either training group alone or the control group. Thus, results indicate that the combination of AAT and GNG training was most effective at reducing implicit evaluation of unhealthy food. *3.4. Effect of training on food intake*

A 2 (AAT: training versus control) x 2 (GNG: training versus control) ANOVA was also used to examine the effects of training on food intake. Neither the main effect of AAT, F(1,189) = 2.35, p = .629, or GNG condition, F(1,189) = .392, p = .532, was significant (see Table 1 for descriptive statistics). There was also no significant interaction between AAT and GNG condition, F(1,189) = 1.92, p = .167. Thus, training condition did not affect food intake.

Table 1

Means (and standard deviations) for implicit food evaluation and consumption (calories), and observed count (and percentages) for healthy first food choice by AAT condition and GNG condition.

		AAT T	raining	AAT Control		
		GNG Training	GNG Control	GNG Training	GNG Control	
Implicit Food Evaluation	М	-0.17	-0.01	-0.01	-0.07	
	SD	(0.42)	(0.35)	(0.37)	(0.46)	
Food Consumption (Calories)	М	157.32	181.73	163.11	147.13	
	SD	(128.83)	(105.62)	(96.84)	(110.65)	
Healthy First Food Choice	Ν	41	38	28	35	
	%	(64.1%)	(67.9%)	(50.9%)	(54.7%)	

3.5. Effect of training on food choice

Overall, participants were slightly more likely to choose a healthy snack food first (n = 142, 59%). A logistic regression analysis was conducted to test group differences. In Step 1, the Wald criterion indicated that AAT condition was a significant predictor of healthy first food choice, B = .55, SE = .27, $\chi^2(1) = 4.24$, p = .039, while GNG condition was not, B = -.16, SE = .27, $\chi^2(1) = .36$, p = .549. Those who completed AAT training were 1.73 times more likely to first choose a healthy snack food than an unhealthy one, Exp(B) odds ratio = 1.73 (95% CI = 1.03 - 2.93). In Step 2, the interaction between AAT and GNG condition was not significant, B = -.02, SE = .53, $\chi^2(1) = .561$, p = .454, which indicates that the combined training did not provide any additional benefit (see Table 1 for descriptive statistics). Overall, the results indicate that participants who were trained to avoid unhealthy food cues were less likely to choose an unhealthy snack first (34.2%) than those in the control condition (47.1%).

4. Discussion

The aim of the current study was to investigate the effect of combined avoidance and control training. The main findings were that the combined training was more effective than either training alone at reducing implicit evaluation of unhealthy food. However, there were no individual or combined effects of training on unhealthy food consumption during the taste test. Finally, the avoidance training did have an effect on food choice, as the training group was less likely than the control group to choose an unhealthy snack first over a healthy one.

Our manipulation check indicated that avoidance training was successful, in that participants in the training condition showed an avoidance tendency away from unhealthy food cues, whereas those in the control condition showed approach toward such cues. The effectiveness of control training at increasing inhibitory control was less clear. Participants in the training condition were slower to respond to unhealthy food cues than those in the control condition, but they did not make fewer food-related commission errors. However, the overall rate of commission errors was very low, indicating good control in our sample. Perhaps samples with poorer inhibitory control would receive more benefit from the control training.

To the best of our knowledge, the current study is the first to examine the effect of combined avoidance and control training on implicit evaluation of unhealthy appetitive cues. As predicted, participants who were trained to both avoid and inhibit responses to unhealthy food cues showed less liking for such cues. We did not find that the individual training tasks reduced implicit liking, in contrast to some previous studies in the alcohol (Wiers et al., 2010; Wiers et al., 2011) and eating domains (Houben et al., 2012; Lawrence et al., 2015; Veling et al., 2008; Veling et al., 2013). Perhaps our individual training tasks were less extensive than in previous studies due to the inclusion of both tasks in the session. Nevertheless, the finding that combined training was most effective at reducing implicit liking suggests that it may be useful to supplement one task with the other, even when the individual training is successful.

Although the training protocols produced changes in implicit liking of unhealthy food, there were no training effects on unhealthy food intake during the taste test. This finding is at odds with some previous studies in the alcohol domain (Wiers et al., 2010; 2011; Eberl et al., 2013; Sharbanee et al., 2014). However, the current finding is not surprising given previous inconsistent effects of avoidance and control training on food consumption. Some studies have shown that avoidance training reduces chocolate consumption (Schumacher et al., 2016), whereas others have found no difference between the training and control groups (Dickson et al., 2016), or have even shown the reverse effect with participants trained to avoid chocolate actually consuming more chocolate (Becker et al., 2015, Study 3).

Dickson et al. (2016) have suggested that food may be different from alcohol in that it constitutes a larger and less specific category. Consequently, alcohol in the taste test is likely to more closely match pictures presented during training than is the case for a range of foods. The same consideration may apply to research examining the effect of control training on food intake. While previous studies in the alcohol domain have generally shown that control training successfully reduces consumption (Houben et al., 2011; Houben et al., 2012; Jones & Field, 2013), studies in the eating domain have reported mixed results, with some finding no effect on consumption (Allom & Mullan, 2015; Lawrence et al., 2015b).

For food choice, we found that avoidance training led to reduced choice for unhealthy food. Specifically, participants who were trained to avoid unhealthy food were more likely to choose a healthy snack than those in the control group. Thus, inducing an avoidance tendency away from unhealthy food cues appears to translate into individuals making a quick, healthy decision when both healthy and unhealthy snacks are presented. This result is consistent with previous studies in the eating domain (Fishbach & Shah, 2006; Becker et al., 2015, Study 1). One potential reason as to why this effect differs from our consumption result is that selecting a snack (particularly under time pressure) may be relatively spontaneous compared to food

intake, which may be more conscious. Another possible reason is that the food choice task included both healthy and unhealthy snacks, while the taste test included only unhealthy snacks. Finally, it is important to consider that the AAT may train cognitive control, as well as the acquisition of automatic avoidance tendencies (Koch, Holland, & van Knippenberg, 2008), which could explain the individual effect of avoidance training on snack food choice.

The current findings can be understood in the context of dual-process models (Strack & Deutsch, 2004). Specifically, approach bias (one aspect of automatic processing) is proposed to contribute to unhealthy eating, while inhibitory control (one aspect of controlled processing) is necessary for the successful regulation of automatic responses in predicting behaviour. In support, the current study demonstrated that combining avoidance with control training had interactive effects on inducing a more negative implicit evaluation of unhealthy food. More generally, it is possible that the varying findings for the different outcome measures used here reflect different proportions of automatic and controlled processing.

The current study has some important practical implications for the use of a combined avoidance and control training intervention. The present sample comprised primarily healthy weight women. Stronger training effects may be expected for individuals who are likely to be most at risk in an environment with an abundance of cues for unhealthy food. In particular, overweight or obese individuals may be more likely to benefit as they have been shown to have both a stronger approach bias for unhealthy food (Havermans, Giesen, Houben & Jansen, 2011; Kemps & Tiggemann, 2015) and lower inhibitory control (Batterink, Yokum & Stice, 2010; Nederkoorn, Smulders, Havermans, & Jansen, 2006). On the other hand, it is possible that it may be more difficult to modify the stronger approach tendencies and reduced inhibitory control of such individuals. Another important practical consideration concerns the stimuli used during training. An intervention that promotes healthy eating by training people to approach healthy food cues (rather than avoid unhealthy food cues) and respond to healthy food cues (rather than inhibit responses to unhealthy food cues) may prove a more positive acceptable protocol (Becker & Ferentzi et al., 2015).

In conclusion, the current study demonstrated that a combined training procedure can be used to modify approach-avoidance tendencies and inhibitory control for unhealthy food, aspects of automatic and controlled processing, respectively. In addition, the combination of avoidance and control training was most effective at reducing implicit liking for unhealthy food. While the combined training did not reduce consumption of unhealthy food, avoidance training on its own did reduce unhealthy snack food choice. These findings support some key theoretical predictions of dual-process models and contribute to developing an understanding of the effectiveness of training aimed at changing cognitive processes underlying unhealthy eating. Future research could extend the current findings to interventions that are aimed at encouraging individuals with problematic eating behaviour to make healthier food choices.
References

- Allom, V., & Mullan, B. (2015). Two inhibitory control training interventions designed to improve eating behaviour and determine mechanisms of change. *Appetite*, 89, 282-290.
- Allom, V., Mullan, B., & Hagger, M. (2015). Does inhibitory control training improve health behaviour? A meta-analysis. *Health Psychology Review*, 1-19.
- Appelhans, B. M., Woolf, K., Pagoto, S. L., Schneider, K. L., Whited, M. C., & Liebman, R. (2011). Inhibiting food reward: Delay discounting, food reward sensitivity, and palatable food intake in overweight and obese women. *Obesity*, *19*, 2175-2182.
- Baker, D. B. Taylor, C. J. Leyva, C. (1995). Continuous performance tests: A comparison of modalities. *Journal of Clinical Psychology*, 51, 548–551.
- Becker, E. S., Ferentzi, H., Ferrari, G., Möbius, M., Brugman, S., Custers, J., ... Rinck, M.(2015). Always approach the bright side of life: A general positivity training reduces stress reactions in vulnerable individuals. *Cognitive Therapy and Research*. In press.
- Becker, D., Jostmann, N. B., Wiers, R. W., & Holland, R. W. (2015). Approach avoidance training in the eating domain: Testing the effectiveness across three single session studies. *Appetite*, 85, 58-65.
- Bowley, C,. Faricy, C., Hegarty, B., Johnstone, S. J., Smith, J., Kelly, P., & Rushby, J.
 (2013). The effects of inhibitory control training on alcohol consumption, implicit alcohol-related cognitions and brain electrical activity. *International Journal of Psychophysiology*, 89, 342–348.
- Brignell, C., Griffiths, T., Bradley, B. P., & Mogg, K. (2009). Attentional and approach biases for pictorial food cues. Influence of external eating. *Appetite*, *52*, 299-306.

- Brockmeyer, T., Hahn, C., Reetz, C., Schmidt, U., & Friederich, H.-C. (2015). Approach bias modification in food craving: A proof-of-concept study. *European Eating Disorders Review*, 23, 352–360.
- Burton, P., Smit, H. J., & Lightowler, H. J. (2007). The influence of restrained and external eating patterns on overeating. *Appetite*, *49*, 191–197.

Cohen, J. (1992). A power primer. Psychological bulletin, 112, 155-159.

- Cohen, D., & Farley, T. A. (2008). Eating as an automatic behavior. *Preventing Chronic Disease*, *5*(1), 1-7.
- Cousijn, J., Goudriaan, A. E., & Wiers, R. W. (2011). Reaching out towards cannabis: Approach-bias in heavy cannabis users predicts changes in cannabis use. *Addiction*, *106*, 1667-1674.
- Dickson, H., Kavanagh, D. J., & MacLeod, C. (2016). The pulling power of chocolate: Effects of approach-avoidance training on approach bias and consumption. *Appetite*, *99*, 46-51.
- Eberl, C., Wiers, R. W., Pawelczack, S., Rinck, M., Becker, E. S., & Lindenmeyer, J. (2013). Approach bias modification in alcohol dependence: do clinical effects replicate and for whom does it work best? *Developmental Cognitive Neuroscience*, *4*, 38-51.
- Fishbach, A., & Shah, J. Y. (2006). Self-control in action: Implicit dispositions toward goals and away from temptations. *Journal of Personality and Social Psychology*, 90, 820-832.
- Friese, M., & Hofmann, W. (2009). Control me or I will control you: Impulses, trait selfcontrol, and the guidance of behavior. *Journal of Research in Personality*, 43, 795– 805.

- Friese, M., Hofmann, W., & Wänke, M. (2008). When impulses take over: Moderated predictive validity of explicit and implicit attitude measures in predicting food choice and consumption behaviour. *British Journal of Social Psychology*, 47, 397-419.
- Furst, T., Connors, M., Bisogni, C. A., Sobal, J., & Falk, L. W. (1996). Food choice: A conceptual model of the process. *Appetite*, 26, 247-266.
- Greenwald, A. G., Nosek, B. A., & Banaji, M. R. (2003). Understanding and using the implicit association test: I. An improved scoring algorithm. *Journal of Personality* and Social Psychology, 85, 197-216.
- Guerrieri, R., Nederkoorn, C., Stankiewicz, K., Alberts, H., Geschwind, N., Martijn, C., & Jansen, A. (2007). The influence of trait and induced state impulsivity on food intake in normal-weight healthy women. *Appetite*, *49*, 66-73.
- Halperin, J. M., Wolf, L. E., Greenblatt, E. R., & Young, G. (1991). Subtype analysis of commission errors on the continuous performance test in children. *Developmental Neuropsychology*, 7, 207–217.
- Havermans, R. C. (2013). Pavlovian craving and overeating: A conditioned incentive model. *Current Obesity Reports*, 2, 165-170.
- Havermans, R. C., Giesen, J. C., Houben, K., & Jansen, A. (2011). Weight, gender, and snack appeal. *Eating Behaviors*, *12*, 126-130.
- Hofmann, W., Friese, M., & Wiers, R. W. (2008). Impulsive versus reflective influences on health behavior: A theoretical framework and empirical review. *Health Psychology Review*, 2, 111-137.
- Houben, K. (2011). Overcoming the urge to splurge: Influencing eating behavior by manipulating inhibitory control. *Journal of Behavior Therapy and Experimental Psychiatry*, 42, 384-388.

- Houben, K., & Jansen, A. (2011). Training inhibitory control. A recipe for resisting sweet temptations. *Appetite*, *56*, 345-349.
- Houben, K., & Jansen, A. (2015). Chocolate equals stop. Chocolate-specific inhibition training reduces chocolate intake and go associations with chocolate. *Appetite*, 87, 318-323.
- Houben, K., Havermans, R. C., Nederkoorn, C., & Jansen, A. (2012). Beer à No-Go:
 Learning to stop responding to alcohol cues reduces alcohol intake via reduced
 affective associations rather than increased response inhibition. *Addiction*, *107*, 1280-1287.
- Houben, K., Nederkoorn, C., & Jansen, A. (2012). Too tempting to resist? Past success at weight control rather than dietary restraint determines exposure-induced disinhibited eating. *Appetite*, *59*, 550-555.
- Houben, K., Nederkoorn. C., Wiers, R. W., & Jansen, A. (2011). Resisting temptation:
 Decreasing alcohol-related affect and drinking behavior by training response
 inhibition. *Drug and Alcohol Dependence*, *116*, 132-136.
- Houben, K., Rothermund, K., & Wiers, R. W. (2009). Predicting alcohol use with a recodingfree variant of the Implicit Association Test. *Addictive Behaviors*, *34*, 487–489.
- Houben, K., & Wiers, R. W. (2009). Response inhibition moderates the influence of implicit associations on drinking behavior. *Alcoholism: Clinical and Experimental Research*, 33, 1–8.
- Jasinska, A. J., Yasuda, M., Burant, C. F., Gregor, N., Khatri, S., Sweet, M., & Falk, E. B. (2012). Impulsivity and inhibitory control deficits are associated with unhealthy eating in young adults. *Appetite*, 59, 738-747.

Jones A., Field, M., (2013). The effects of cue-specific inhibition training on alcohol consumption in heavy social drinkers. *Journal of Experimental Clinical Psychopharmacology*, 21, 8-16.

- Jones, A., Robinson., E., & Field, M. (2014). A comparison of three types of web-based inhibition training for the reduction of alcohol consumption in problem drinkers: Study protocol. *BMC Public Health*, 14, 796-896.
- Kakoschke, N., Kemps, E., & Tiggemann, M. (2015). Combined effects of cognitive bias for food cues and poor inhibitory control on unhealthy food intake. *Appetite*, *87*, 358-364.
- Karpinski, A., & Steinman, R. B. (2006). The single category implicit association test as a measure of implicit social cognition. *Journal of Personality and Social Psychology*, 91, 16-32.
- Keel, P. K., Baxter, M. G., Heatherton, T. F., & Joiner Jr, T. E. (2007). A 20-year longitudinal study of body weight, dieting, and eating disorder symptoms. *Journal of Abnormal Psychology*, 116, 422-432.
- Kemps, E., & Tiggemann, M. (2015). Approach bias for food cues in obese individuals. *Psychology & Health, 30,* 370-380.
- Kemps, E., Tiggemann, M., & Hollitt, S. (2014). Biased attentional processing of food cues and modification in obese individuals. *Health Psychology*, *33*, 1391–1401.
- Kemps, E., Tiggemann, M., Martin, R., & Elliott, M. (2013). Implicit approach–avoidance associations for craved food cues. *Journal of Experimental Psychology: Applied*, 19, 30-38.
- Koch, S., Holland, R. W., & van Knippenberg, A. (2008). Regulating cognitive control through approach–avoidance motor actions. *Cognition*, 109, 133–142.

- Koster, E. H., Fox, E., & MacLeod, C. (2009). Introduction to the special section on cognitive bias modification in emotional disorders. *Journal of Abnormal Psychology*, *118*, 1-4.
- Lawrence, N. S., O'Sullivan, J., Parslow, D., Javaid, M., Adams, R. C., Chambers, C. D., ...
 & Verbruggen, F. (2015). Training response inhibition to food is associated with weight loss and reduced energy intake. *Appetite*, 95, 17-28.
- Lawrence, N. S., Verbruggen, F., Morrison, S., Adams, R. C., & Chambers, C. D. (2015a). Stopping to food can reduce intake. Effects of stimulus-specificity and individual differences in dietary restraint. *Appetite*, 85, 91-103.
- Loeber, S., Grosshans, M., Korucuoglu, O., Vollmert, C., Vollstädt-Klein, S., Schneider, S., ... & Kiefer, F. (2012). Impairment of inhibitory control in response to foodassociated cues and attentional bias of obese participants and normal-weight controls. *International Journal of Obesity*, *36*, 1334-1339.
- Marteau, T. M., Hollands, G. J., & Fletcher, P. C. (2012). Changing human behavior to prevent disease: The importance of targeting automatic processes. *Science*, *337*, 1492-1495.
- Mogg, K., Bradley, B. P., Hyare, H., & Lee, S. (1998). Selective attention to food-related stimuli in hunger: Are attentional biases specific to emotional and pathological states, or are they found in normal drive states? *Behaviour Research and Therapy*, *36*, 227–237.
- Mogg, K., Holmes, A., Garner, M., & Bradley, B. P. (2008). Effects of threat cues on attentional shifting, disengagement and response slowing in anxious individuals. *Behaviour Research and Therapy*, 46, 656-667.

Nederkoorn, C., Guerrieri, R., Havermans, R. C., Roefs, A., & Jansen, A. (2009). The interactive effect of hunger and impulsivity on food intake and purchase in a virtual supermarket. *International Journal of Obesity*, *33*, 905-912.

- Nederkoorn, C., Houben, K., Hofmann, W., Roefs, A., & Jansen, A. (2010). Control yourself or just eat what you like? Weight gain over a year is predicted by an interactive effect of response inhibition and implicit preference for snack foods. *Health Psychology*, 29, 389–393.
- Olson, M. A., & Fazio, R. H. (2004). Reducing the influence of extrapersonal associations on the Implicit Association Test: personalizing the IAT. *Journal of Personality and Social Psychology*, 86, 653-667.
- Palfai, T. P., & Ostafin, B. D. (2003). Alcohol-related motivational tendencies in hazardous drinkers: Assessing implicit response tendencies using the modified-IAT. *Behaviour Research and Therapy*, 41, 1149-1162.
- Polivy, J., Herman, C. P., & Coelho, J. S. (2008). Caloric restriction in the presence of attractive food cues: External cues, eating, and weight. *Physiology & Behavior*, 94, 729-733.
- Rothermund, K., Teige-Mocigemba, S., Gast, A., & Wentura, D. (2009). Minimizing the influence of recoding in the implicit association test. The recoding-free implicit association test (IAT-RF). *The Quarterly Journal of Experimental Psychology*, 62, 84–98.
- Seibt, B., Häfner, M., & Deutsch, R. (2007). Prepared to eat: How immediate affective and motivational responses to food cues are influenced by food deprivation. *European Journal of Social Psychology*, *37*, 359-379.

- Sharbanee, J. M., Hu, L., Stritzke, W. G., Wiers, R. W., Rinck, M., & MacLeod, C. (2014). The effect of approach/avoidance training on alcohol consumption is mediated by change in alcohol action tendency. *PloS one*, 9(1), 1-10.
- Strack, F., & Deutsch, R. (2004). Reflective and impulsive determinants of social behavior. *Personality and Social Psychology Review*, 8, 220–247.
- Tabachnick, B. G., & Fidell, L. S. (2007). Using multivariate statistics (5th ed.). Boston,MA: Pearson/Allyn and Bacon.
- Thush, C., Wiers, R. W., Ames, S. L., Grenard, J. L., Sussman, S., & Stacy, A. W. (2008).
 Interactions between implicit and explicit cognition and working memory capacity in the prediction of alcohol use in at-risk adolescents. *Drug and Alcohol Dependence*, 94, 116–124.
- Van Koningsbruggen, G. M., Veling, H., Stroebe, W., & Aarts, H. (2014). Comparing two psychological interventions in reducing impulsive processes of eating behaviour:
 Effects on self-selected portion size. *British Journal of Health Psychology, 19*, 767-782.
- Veenstra, E. M., & de Jong, P. J. (2010). Restrained eaters show enhanced automatic approach tendencies towards food. *Appetite*, *55*, 30-36.
- Veling, H., Aarts, H., & Stroebe, W. (2013a). Stop signals decrease choices for palatable foods through decreased food evaluation. *Frontiers in Psychology*, 4 (875), 1-7.
- Veling, H., Aarts, H., & Stroebe, W. (2013b). Using stop signals to reduce impulsive choices for palatable unhealthy foods. *British Journal of Health Psychology*, 18, 354-368.
- Veling, H., Holland, R. W., & van Knippenberg, A. (2008). When approach motivation and behavioral inhibition collide: Behavior regulation through stimulus devaluation. *Journal of Experimental Social Psychology*, 44, 1013–1019.

- Veling, H., van Koningsbruggen, G. M., Aarts, H., & Stroebe, W. (2014). Targeting impulsive processes of eating behavior via the internet. Effects on body weight. *Appetite*, 78, 102-109.
- Verbruggen, F., & Logan, G. D. (2009). Automaticity of cognitive control: Goal priming in response-inhibition paradigms. *Journal of Experimental Psychology: Learning Memory and Cognition*, 35, 1381–1388.
- Wiers, R. W., Ames, S. L., Hofmann, W., Krank, M., & Stacy, A. W. (2010). Impulsivity, impulsive and reflective processes and the development of alcohol use and misuse in adolescents and young adults. *Frontiers in Psychology*, 1, 1–12.
- Wiers, R. W., Eberl, C., Rinck, M., Becker, E. S., & Lindenmeyer, J. (2011). Retraining automatic action tendencies changes alcoholic patients' approach bias for alcohol and improves treatment outcome. *Psychological Science*, 22, 490-497.
- Wiers, R. W., Gladwin, T. E., Hofmann, W., Salemink, E., & Ridderinkhof, K. R. (2013).
 Cognitive bias modification and cognitive control training in addiction and related psychopathology mechanisms, clinical perspectives, and ways forward. *Clinical Psychological Science*, 1, 192-212.
- Wiers, R. W., Houben, K., Fadardi, J. S., van Beek, P., Rhemtulla, M., & Cox, W. M. (2015).Alcohol cognitive bias modification training for problem drinkers over the web.*Addictive Behaviors*, 40, 21-26.
- Wiers, C. E., Kühn, S., Javadi, A. H., Korucuoglu, O., Wiers, R. W., Walter, H., ... & Bermpohl, F. (2013). Automatic approach bias towards smoking cues is present in smokers but not in ex-smokers. *Psychopharmacology*, 229, 187-197.
- Wiers, R. W., Rinck, M., Dictus, M., & Van den Wildenberg, E. (2009). Relatively strong automatic appetitive action-tendencies in male carriers of the OPRM1 G-allele. *Genes, Brain and Behavior*, 8, 101-106.

- Wiers, R. W., Rinck, M., Kordts, R., Houben, K., & Strack, F. (2010). Retraining automatic action-tendencies to approach alcohol in hazardous drinkers. *Addiction*, 105, 279-287.
- World Health Organization [WHO]. (2014). Obesity and overweight. Retrieved October 1, 2015 from http://www.who.int/mediacentre/factsheets/fs311/en/.

CHAPTER 7: STUDY 5

Impulsivity Moderates the Effect of Approach Bias Modification on Healthy Food Consumption

Naomi Kakoschke, Eva Kemps, and Marika Tiggemann School of Psychology, Flinders University, South Australia

Corresponding author: Naomi Kakoschke, School of Psychology, Flinders University, GPO Box 2100, Adelaide, 5001, South Australia. Email: naomi.kakoschke@flinders.edu.au **Funding:** This research was supported under the Australian Research Council's Discovery Project funding scheme (project number DP130103092).

Statement of co-authorship: All authors were involved in the formulation of the study concept and design. Naomi Kakoschke collected the data, and completed the data analysis and the initial draft of the manuscript. Eva Kemps and Marika Tiggemann edited multiple revisions of the manuscript.

Acknowledgements: We are grateful to Paul Douglas for developing the software for the computer administered tasks.

This manuscript is currently under review.

Abstract

The study aimed to modify approach bias for healthy and unhealthy food and to determine its effect on subsequent food consumption. In addition, we investigated the potential moderating role of impulsivity in the effect of approach bias re-training on consumption. Participants were 200 undergraduate women (17-26 years) who were randomly allocated to one of five conditions of an approach-avoidance task varying in the training of an approach bias for healthy food, unhealthy food, and non-food cues. Outcome variables were approach bias for healthy and unhealthy food and the proportion of healthy relative to unhealthy snack food consumed. As predicted, approach bias for healthy food significantly increased in the 'avoid unhealthy /approach healthy food condition. Importantly, the effect of training on consumption was moderated by trait impulsivity. Participants high in impulsivity consumed a greater proportion of healthy snack food following the 'avoid unhealthy/ approach healthy food following the 'avoid unhealthy/ approach healthy food following the 'avoid unhealthy/ approach healthy food suggestion that automatic processing of appetitive cues has a greater influence on consumption behaviour in individuals with poor self-regulatory control.

Introduction

The contemporary Western environment provides continual exposure to an abundance of unhealthy food cues through advertising on the internet, TV, billboards, and in magazines (Havermans, 2013). An 'obesogenic' environment has been linked to consuming too much food high in fat, salt, and sugar, and not enough fruit and vegetables (Hill & Peters, 1998). Unhealthy eating behaviour is a key contributor to the increasing rates of overweight and obesity, which have doubled during the last few decades (Cohen, 2008). It is estimated that 35% of adults can now be classified as overweight and 11% as obese (WHO, 2014). Excess body weight can lead to negative health consequences such as cancer, cardiovascular disease and diabetes (Must et al., 1999). Therefore, it is important to identify the mechanisms by which exposure to appetitive food cues in the environment can affect unhealthy eating behaviour.

One such mechanism implicated in the development of unhealthy eating behaviour is biased automatic processing of appetitive cues (Marteau, Hollands, & Fletcher, 2012). Recent dual-process models posit that health behaviours are determined by two types of processing: automatic and controlled (Strack & Deutsch, 2004). Automatic processing is fast, implicit, and effortless, while controlled processing is slower and involves conscious reflection based on long-term goals and personal standards. Moreover, automatic processing is thought to involve cognitive biases, such as an approach bias, which is the automatic tendency to reach out toward (i.e., approach) rather than move away from (i.e., avoid) attractive cues (e.g., a piece of chocolate cake) in the environment (Wiers, Gladwin, Hofmann, Salemink, & Ridderinkoff, 2013). The influence of automatic processes on behaviour is argued to be regulated by controlled processes; however, there exist individual differences in the ability to regulate automatic processing. For example, trait impulsivity, which refers to 'a general tendency to act without deliberation' may allow automatic processing of cues to exert a greater influence on behaviour due to poor self-regulatory control (Hofmann, Friese, & Wiers, 2008, p. 113). Indeed, research shows that impulsivity is associated with increased unhealthy food intake (Guerrieri, Nederkoorn, & Jansen, 2007; Nederkoorn, Braet, Van Eijs, Tanghe, & Jansen, 2006; Kakoschke, Kemps, & Tiggemman et al., 2015b).

Researchers have documented approach biases for a variety of appetitive substances, including alcohol (Wiers, Rinck, Kordts, Houben, & Strack, 2010), cigarettes (Wiers & Kuhn et al., 2013) and cannabis (Cousijn, Goudriaan, & Wiers, 2011). Furthermore, research shows that approach bias predicts increased consumption of alcohol (Wiers, Rinck, Dictus, & Van Den Wildenberg, 2009; Wiers et al., 2010) and cannabis (Cousijn et al., 2011). Similarly, in the eating domain, approach bias has been reliably demonstrated for a variety of unhealthy foods (Brignell, Griffiths, Bradley, & Mogg, 2009; Havermans, Giesen, Houben, & Jansen, 2011; Kemps & Tiggemann, 2015; Kemps, Tiggemann, Martin, & Elliott, 2013; Veenstra & de Jong; 2010). Importantly, approach bias for food has been shown to be linked to increased unhealthy food intake (Hofmann, Gschwendner, Wiers, Friese, & Schmitt, 2008; Kakoschke, Kemps, & Tiggemann, 2015a; Nederkoorn, Houben, Hofmann, Roefs, & Jansen, 2010). The evidence suggests that approach bias contributes to the consumption of appetitive substances.

Recently, researchers have begun to investigate whether approach bias for appetitive substances can be modified using a computerised cognitive training paradigm. In commonly used protocols such as the Approach-Avoidance Task (AAT), participants are instructed to respond to a picture based on an irrelevant feature (e.g., portrait or landscape orientation) by pushing or pulling a joystick. Responses are based on a feature of the picture, rather than its content, to ensure that the task captures automatic processing (Wiers & Gladwin et al., 2013). An avoidance of appetitive cues can be trained using a modified AAT, which manipulates the contingencies of target (appetitive) and control pictures. Specifically, target pictures are presented in a format that consistently requires them to be pushed (avoided) and control

pictures in a format that consistently requires them to be pulled (approached). In contrast, 'approach training' involves the reverse contingencies, while what has been termed 'shamtraining' (a neutral or control condition) is when the target and control pictures are approached and avoided with equal frequency.

Previous studies have shown that approach bias for appetitive cues can be modified by manipulating the contingencies of the AAT. Specifically, research shows that the AAT can be used to successfully reduce the approach bias for alcohol (e.g., Wiers et al., 2010; Wiers et al., 2011) and chocolate (Schumacher, Kemps, & Tiggemann, 2016; Dickson, Kavanagh, & Macleod, 2016). Furthermore, such training has been found to reduce subsequent consumption of alcohol (Wiers et al., 2010) and chocolate (Schumacher et al., 2016) in laboratory taste tests. A recent review of the literature concluded that approach bias modification is an effective intervention for reducing approach biases for unhealthy substances and discouraging the consumption of alcohol, cigarettes, and unhealthy food (Kakoschke, Kemps, & Tiggemann, 2017).

It is clearly possible to avoid unhealthy substances such as alcohol and cigarettes because there is no biological requirement to consume them. However, food is a substance that is essential for human survival; complete avoidance is not possible. Instead, a healthy diet is about developing the right balance between eating enough healthy food and avoiding eating too much unhealthy food. It may be that an approach bias modification protocol that simultaneously encourages the avoidance of unhealthy food and the approach of healthy food lends itself best to promoting a healthy diet. Moreover, interventions that not only discourage an unhealthy behaviour, but also promote a healthy one, are likely to be more attractive and acceptable to participants, an important consideration for cognitive bias modification tasks (Wiers et al., 2013). A few studies have investigated the use of approach bias modification to promote healthy eating. In an early study, Fishbach and Shah (2006, Study 5) trained participants to approach healthy and avoid unhealthy food or alternatively, to approach unhealthy and avoid healthy food. They found that participants in the approach healthy/avoid unhealthy food condition subsequently made healthier snack choices. More recently, Dickson et al. (2016) compared approach healthy food/avoid chocolate training with approach chocolate/avoid healthy food training. Although approach bias was trained in the expected direction, there was no group difference in chocolate consumption. In another recent study, Maas, Keijers, Rinck, Tanis and Becker (2015) found that approach healthy/avoid unhealthy food training resulted in increased avoidance of unhealthy and approach of healthy food. However, they did not measure eating behaviour. Finally, Becker et al. (2015, Study 1) found no difference between approach healthy/avoid unhealthy food training and a control group (sham-training) in healthy snack choice. Thus, the evidence for approach bias modification of healthy eating is inconsistent.

One potential methodological explanation for the mixed findings lies in the particular comparison condition used. Similar to previous studies in the alcohol domain, Fishbach and Shah (2006, Study 5), who found a positive result, compared two extreme training conditions i.e., approach healthy/avoid unhealthy food versus avoid healthy food/approach unhealthy food. In contrast, Becker et al. (2015), who did not obtain positive results, compared approach healthy/avoid unhealthy food training with a less extreme condition (i.e., sham-training). Thus, it may be that a more extreme comparison is necessary to obtain a significant difference in healthy eating. To date, no previous study has compared all three conditions (i.e., 'approach', 'avoidance', and 'sham' training).

In addition, the approach healthy/avoid unhealthy food training has two interwoven components: approach healthy food, avoid unhealthy food. To determine which component is

most important for effective re-training of the approach bias and subsequent consumption, we included two further training conditions. In one, approach of healthy food was paired with avoidance of a non-food category, whereas in the other one, avoidance of unhealthy food was paired with approach of a non-food category. Thus, in total, the present study included five training conditions: simultaneous approach of healthy and avoidance of unhealthy food; a reverse training condition (i.e., simultaneous approach of unhealthy and avoidance of healthy food); a control condition in which approach-avoidance of healthy and unhealthy food was equal (i.e., sham-training); an avoid unhealthy food (approach non-food) condition and an approach healthy food (avoid non-food) condition.

A different kind of factor that may contribute to the observed inconsistent effects of approach bias re-training on food consumption is individual differences, which might make re-training differentially effective for different people. In particular, individual differences in aspects of self-regulatory control, such as trait impulsivity, have been shown to moderate the influence of impulses in general (Hofmann & Friese, 2008; Thush et al., 2008). Thus, trait impulsivity may predict whether or not the training promotes successful regulation of approach bias in determining consumption behaviour. To date, no study has tested the potential moderating role of trait impulsivity in the effect of approach bias training on consumption. Here, we predicted that participants with higher levels of impulsivity would show greater benefit from the training because automatic processes likely play a more important role for individuals who tend to have difficulty with self-regulatory control.

In sum, the main aim of the study was to investigate whether approach bias for both healthy food and unhealthy food can be re-trained using approach bias modification, and to determine the effect of such training on subsequent food consumption. Specifically, it was predicted that participants trained to approach healthy food and avoid unhealthy food would have a greater increase in approach bias for healthy food and decrease in approach bias for unhealthy food compared to those trained to approach unhealthy food and avoid healthy food, or those in the control condition (i.e., sham-training). It was expected that participants trained to approach healthy food and avoid unhealthy food would consume the greatest proportion of healthy rather than unhealthy snack food. The second aim was to investigate the potential moderating role of trait impulsivity. Specifically, it was predicted that training effects would be greater for highly impulsive individuals.

Method

Participants

A total of 200 women were recruited from the undergraduate student population at Flinders University. They were aged 17 to 26 years (M = 20.16, SD = 2.24) and had a mean body mass index of 23.12 kg/m² (SD = 4.83), which is classified as being within the healthy weight range (i.e., 18.5–24.9 kg/m²). Only women were recruited because they have a greater tendency to overeat than men (Burton, Smit, & Lightowler, 2007). Participants were recruited if they could speak English fluently, liked most foods, and did not have any food allergies, intolerances, or special dietary requirements. Participants were instructed to eat something two hours before their scheduled testing session in the laboratory to ensure that they were not hungry as hunger has been shown to confound approach bias for food cues (Seibt, Hafner, & Deutsch, 2007). Most participants reported having complied with this instruction as the mean time period since participants had last eaten was 2.40 hours (SD = .98). Participants also rated their current hunger level on a 100 mm visual analogue scale ranging from 'not hungry at all' to 'extremely hungry' (Grand, 1968). Mean hunger ratings fell slightly below the mid-point of the scale (M = 47.24, SD = 22.07).

Design

The study used a 5 (AAT training condition) x 2 (picture: unhealthy food, healthy food) x 2 (time: pre-training, post-training) mixed experimental design. Participants were

randomly allocated to one of the five training conditions: (1) avoid unhealthy food and approach healthy food; (2) avoid unhealthy food and approach non-food (i.e., animals); (3) approach healthy food and avoid non-food (i.e., animals); (4) approach unhealthy food and avoid healthy food; (5) approach and avoid healthy and unhealthy food equally (i.e., control).

Materials

Stimulus materials. Following Weirs et al. (2010), the Approach-Avoidance Task was adapted to measure and modify approach bias for healthy food and unhealthy food. The stimuli were 60 digital coloured photographs (presented in a resolution of 1024 x 768 pixels), comprising 20 images of healthy foods, 20 images of unhealthy foods and 20 images of nonfood stimuli depicting animals not normally eaten in Western society. Animals were chosen for the non-food stimuli as they, like food, are overall appealing. A portrait (aspect ratio 3:4) and landscape (aspect ratio 4:3) format of each image was created. The healthy food pictures were selected from a subset of those used in a previous study (Kakoschke, Kemps, & Tiggemann, 2014). The ratings were obtained from a pilot test with 20 women aged 18–25 years (M = 21.60, SD = 1.50) in which participants were asked to rate 36 pictures of healthy food on 9-point pleasure and arousal scales. The ratings for the unhealthy food and animal pictures were obtained from a pilot test in which 21 women aged 17-45 years (M = 23.67, SD = 8.28) rated 590 pictures of unhealthy food and animals on 9-point pleasure and arousal scales (Kemps, Tiggemann, & Hollitt, 2014). The pictures were selected on the basis that the healthy food and unhealthy food, and animal categories did not significantly differ on mean ratings of pleasure and arousal (all ps > .05). Another 12 images of common objects (e.g., ball, flower) were used for the practice trials preceding the task.

Approach-Avoidance Task. Based on standard procedures (e.g., Wiers et al., 2010), a computerized Approach-Avoidance Task was used. The protocol consisted of three phases: (1) a pre-training phase in which participants' approach bias for healthy and unhealthy food was measured; (2) a training phase in which participants were trained to approach or avoid healthy food, unhealthy food and/or animals; and (3) a post-training phase in which participants' approach bias for healthy and unhealthy food was again measured.

On each trial of the pre- and post-training phases, participants began by pressing the start button on the top of the joystick. A picture of a healthy food or an unhealthy food then appeared in the centre of the screen. Participants were instructed to push or pull the joystick according to whether the picture was presented in portrait versus landscape format. These instructions were counterbalanced (i.e., half of the participants pulled portrait pictures and pushed landscape, and vice versa). When participants pulled the joystick, the picture size increased (simulating approach), while pushing the joystick decreased the picture size (simulating avoidance; Neumann & Strack, 2000). Participants were asked to respond as quickly and accurately as possible. Prior to the pre-training phase, 12 practice trials were used so that participants could learn to push and pull the joystick in response to the picture format. During the pre-training phase, each of the 40 images (20 healthy and 20 unhealthy food) were shown twice, once in the format participants were instructed to pull and once in the format they were instructed to push, resulting in 80 trials. Thus, participants pushed and pulled the healthy and unhealthy food pictures equally often.

In the training phase, participants completed a modified Approach-Avoidance Task. Specifically, the push-pull contingencies of healthy food, unhealthy food, and animal pictures were manipulated to create five training conditions (see Table 1). Participants in all five conditions pushed and pulled pictures from the two stimulus categories with equal frequency. Following previous studies (Schumacher et al., 2016), the 40 images used in each condition were presented six times resulting in 240 trials.

In the post-training phase, participants again undertook the assessment version of the Approach-Avoidance Task, as they did in the pre-training phase.

	Contin	gencies
Approach-Avoidance Training Condition	Push (avoid)	Pull (approach)
Avoid unhealthy food /Approach healthy food	90% = unhealthy	10% = unhealthy
	10% = healthy	90% = healthy
Avoid unhealthy food/Approach animals	90% = unhealthy	10% = unhealthy
	10% = animals	90% = animals
Approach healthy food/Avoid animals	10% = healthy	90% = healthy
	90% = animals	10% = animals
Approach unhealthy food/Avoid healthy food	10% = unhealthy	90% = unhealthy
	90% = healthy	10% = healthy
Approach/avoid healthy/unhealthy food equally	50% = healthy	50% = healthy
	50% = unhealthy	50% = unhealthy

Push/pull contingencies for each training condition on the Approach-Avoidance Task.

For the trials from the pre- and post-training phases, median reaction times were calculated for the four combinations of pushing versus pulling healthy and unhealthy food pictures. Reaction times on pull (approach) trials were subtracted from reaction times on push (avoidance) trials, resulting in positive bias scores that indicate relative approach and negative bias scores that indicate relative avoidance for each of the two types of pictures.

Taste test. A so-called taste test was used to assess the effect of the training on healthy and unhealthy food consumption after the post-training phase of the Approach-Avoidance task. Participants were presented with a platter comprising two healthy (grapes and almonds) and two unhealthy snacks (chocolate M&Ms and potato crisps). The snacks were presented in equally-filled separate bowls and were chosen as they are commonly consumed and are bite-sized to facilitate eating. The presentation order of the four bowls was counterbalanced across participants using a 4×4 Latin square. Participants were instructed to taste as much or as little of the food as they liked so that they could rate each snack on several characteristics (e.g., 'How sweet is this food?'). They were given 10 minutes to complete their ratings after which time the platter was taken away. The amount of each food consumed was calculated by subtracting the weight (in grams) of the food after the taste test from the weight of the food before the taste test. The weight in grams for each food was then converted into the number of calories consumed. An overall measure which included healthy and unhealthy food was calculated as the proportion of healthy food consumed relative to total food consumption.

Barratt Impulsiveness Scale (BIS-11). Trait impulsivity was assessed by the widely used BIS-11 (Patton et al., 1995). The BIS-11 comprises 30 items designed to assess different impulsivity aspects such as attentional (e.g., 'I am restless at the theatre or lectures'), motor (e.g., 'I do things without thinking'), and non-planning (e.g., 'I am more interested in the present than the future'). Items are scored on a 4-point Likert-scale ranging from 1 = 'rarely/ never' to 4 = 'almost always/always'. Scores are summed to provide a total score, with scores of 72 or above indicating high impulsivity (Stanford et al., 2009). In the current study, 49 participants were classified as highly impulsive and 151 were classified as low. The BIS-11 has good test-retest reliability and internal reliability (Patton et al. 1995; Stanford et al., 2009). In the present study, internal reliability for the total scale was also good ($\alpha = .85$). **Procedure**

Participants were tested in the Food Laboratory in the School of Psychology at Flinders University during a single one-hour session. After providing informed consent, participants completed a brief demographics questionnaire, followed by the Approach-Avoidance Task. After completing the computer task, participants underwent the taste test, and finally, they completed the Barratt Impulsiveness Scale. All participants were debriefed via e-mail once data collection was completed.

Results

Statistical considerations

An alpha value of .05 was used to determine significant *p* values. Effect size measures were partial eta² (η^2) for ANOVA and Cohen's *d* for t-tests. For η^2 , a value of .01 represents a small effect, .06, a medium effect, and .14, a large effect, while for Cohen's *d*, .20 represents a small effect, .50, a medium effect, and .80, a large effect (Cohen, 1992).

Effect of training on approach bias

To assess the effect of the training on approach bias for each of the two picture types (i.e., healthy and unhealthy food), reaction times at pre-training were compared with those at post-training. Following standard protocols (e.g., Wiers et al., 2010), incorrect responses and extreme reaction times (i.e., <300ms or >3000 ms) were removed. Individual reaction times that were more than 2.5 *SD* from the mean were identified as outliers and were changed to plus or minus one unit from the first non-outlier (Tabachnick & Fidell, 1989).

Changes in approach bias scores were assessed using a 5 (AAT condition: avoid unhealthy/approach healthy food, avoid unhealthy food/ approach animals, approach healthy/ avoid animals, approach unhealthy/avoid healthy food, approach/avoid healthy/unhealthy food equally) x 2 (picture: healthy food, unhealthy food) x 2 (time: pre-training, posttraining) mixed model ANOVA. Results revealed significant condition x picture, F(1, 195) =10.05, p < .001, $\eta^2 = .170$, and picture x time interactions, F(1, 195) = 5.52, p = .020, $\eta^2 =$.028. Importantly, there was a significant condition x picture x time interaction, F(1, 195) =9.032, p < .001, $\eta^2 = .156$.

The nature of the three-way interaction was further examined using separate pairwise comparisons for healthy and unhealthy food to assess change in approach bias from pre- to post-training in each condition. For unhealthy food (Figure 1a), approach bias significantly decreased from pre- to post-training in the avoid unhealthy/approach healthy food condition,

 $F(1,195) = 16.48, p < .001, \eta^2 = .078$, and in the avoid unhealthy food/approach animals condition, $F(1, 195) = 9.84, p = .002, \eta^2 = .048$. In addition, approach bias for unhealthy food significantly increased in the approach unhealthy/avoid healthy food condition, F(1, 148) = $9.06, p = .003, \eta^2 = .044$. These results showed that it is the 'avoid unhealthy food', rather than the 'approach healthy food', component that seems to be crucial for the training effect for unhealthy food.

For healthy food (Figure 1b), approach bias significantly increased from pre- to posttraining in the avoid unhealthy/approach healthy food condition, F(1,195) = 4.47, p = .036, $\eta^2 = .022$. Approach bias for healthy food also trended toward a decrease from pre- to posttraining in the approach unhealthy/avoid healthy food condition, F(1,195) = 3.31, p = 0.70, $\eta^2 = .017$. There were no other significant changes in approach bias for healthy food cues. These results showed that both the 'avoid unhealthy food' and 'approach healthy food' components appear to be important for the training effect on healthy food. Overall, it seems that it was easier to train the avoidance of unhealthy food than the approach toward healthy food cues.



(a) Unhealthy Food





AAT Training Condition

Fig 1. Mean approach bias scores (with 95% confidence intervals [CI]) for AAT training condition at pre- and post-training for (a) unhealthy and (b) healthy food. Within-subjects 95% CIs were calculated using formulae from Masson and Loftus (2003). *p < .05, *p = .07.

Effect of training on consumption

A one-way ANOVA was used to assess the effect of the training on the proportion of healthy food consumed (see Table 2 for descriptive statistics). Results revealed no significant differences in consumption between conditions, F(4, 196) = .425, p = .791, $\eta^2 = .009$. This result indicates that training did not have an overall effect on the relative amount of healthy food consumed. As can be seen from the means, participants in all of the training conditions consumed about 50% healthy food. The only condition in which participants ate more healthy than unhealthy food was the 'avoid unhealthy/approach healthy food' condition (53%), but pairwise analyses showed that this did not differ significantly from any other condition (all p's >.25).

Table 2

Means and standard deviations for the proportion of healthy food consumed for each training condition on the Approach-Avoidance Task.

	Proportion of Healthy Food Consumed	
Approach-Avoidance Training Condition	Mean	Standard Deviation
Avoid unhealthy food /Approach healthy food	.53	.23
Avoid unhealthy food/Approach animals	.48	.16
Approach healthy food/Avoid animals	.48	.23
Approach unhealthy food/Avoid healthy food	.50	.19
Approach/avoid healthy/unhealthy food equally	.47	.18

Moderation of training effect by impulsivity

Further analyses were performed to determine whether impulsivity group (high/low, based on the established cut-off score) moderated the effect of training on approach bias or consumption. For approach bias, results showed no significant main effect of impulsivity group, F(1, 190) = .091, p = .764, $\eta^2 = .001$, nor any interactions involving impulsivity (all p's >.10). This indicates that the training had equivalent effects on approach biases for healthy food and unhealthy food regardless of participants' level of trait impulsivity.

For consumption, on the other hand, results showed likewise no main effect of impulsivity group, F(1, 190) = .017, p = .998, $\eta^2 = .001$, but a significant interaction between condition and impulsivity group, F(4, 190) = 2.47, p = .046, $\eta^2 = .049$ (see Figure 2). Separate analyses for the impulsivity groups revealed that there was no significant difference in the proportion of healthy food consumed across conditions for participants with low impulsivity, F(4, 190) = .304, p = .875, $\eta^2 = .006$. In contrast, for those with high impulsivity, there was a significant difference between conditions, F(4, 190) = 2.596, p = .038, $\eta^2 = .001$. Pairwise comparisons showed that the proportion of healthy food consumed was significantly greater in the avoid unhealthy/approach healthy food condition than in the approach unhealthy/avoid healthy food condition, p = .005, d = 1.15, and in the approach/avoid healthy/unhealthy food equally condition, p = .010, d = 1.06. This indicates that the effect of the training on the relative amount of healthy food consumed was found only in participants with high impulsivity.



AAT Training Condition

Fig. 2. Proportion of healthy food consumption in calories (with standard error bars) as a function of AAT training condition and trait impulsivity. *p < .05

Discussion

The current study aimed to use approach bias modification to encourage healthier eating behaviour. As predicted, participants trained to avoid unhealthy and approach healthy food showed a bias away from unhealthy food and a bias toward healthy food following the training, whereas those trained to approach unhealthy and avoid healthy food showed a bias toward unhealthy food and a (non-significant) bias away from healthy food. In addition, for consumption, training interacted with trait impulsivity to predict relative healthy food intake in a subsequent taste test. Specifically, only among highly impulsive participants, did those trained to avoid unhealthy and approach healthy food eat a greater proportion of healthy food. As expected, participants allocated to the avoid unhealthy and approach healthy food condition showed an increase in their approach bias for healthy food and a decrease in their approach bias for unhealthy food from pre- to post-training. These results are consistent with previous studies showing that an increased approach bias for healthy food cues can be trained while simultaneously training an avoidance of unhealthy food (Fishbach & Shah, 2006, Study 5; Maas et al., 2015; Dickson et al., 2015). Conversely, in the current study, participants in the approach unhealthy and avoid healthy food condition showed an increased approach bias for unhealthy food. Previous studies have similarly shown an increased bias for unhealthy food after training toward unhealthy food (Fishbach & Shah, 2006; Schumacher et al., 2016). As expected, there were no approach bias changes for healthy or unhealthy food in the condition in which healthy and unhealthy food were equally approached and avoided.

The design of our study was novel in that it included two further training conditions, which examined either avoidance of unhealthy food or approach of healthy food compared to animals (a non-food category). The finding that approach bias for unhealthy food reduced in participants trained to avoid unhealthy food and approach animals fits with one previous study in the food domain (Brockmeyer, Hahn, Reetz, Schmidt, & Friederich, 2015), but our results indicate that it is the 'avoid unhealthy food' component that is important. In contrast, participants trained to approach healthy food and avoid animals showed no significant change in approach bias for healthy or unhealthy food. It appears that both the 'approach healthy' and 'avoid unhealthy' component is needed to produce an avoidance of unhealthy food. More generally, it is evident that the control comparison condition chosen ('avoid healthy/approach unhealthy' or 'sham' training) is important in evaluating the effectiveness of approach training for healthy food.

Our finding that approach bias modification did not produce a main effect on healthy snack food consumption is at odds with some results for alcohol and chocolate (Wiers et al., 2010; Schumacher et al., 2016). It also does not support the finding of Fishbach and Shah (2006, Study 5), which showed that participants trained to approach healthy food were more likely to choose a healthy snack food over an unhealthy one than those trained to avoid healthy food. However, our finding is consistent with some previous studies that likewise found no benefit for eating behaviour (Becker et al., 2015, Study 1; Dickson et al., 2016).

Our final aim was to examine the role of impulsivity on training effects for both approach bias and food consumption. We found that impulsivity did not moderate the effect of training on approach bias for healthy or unhealthy food. This indicates that the re-training was equivalent in its effectiveness at changing approach biases, regardless of participants' level of trait impulsivity. In contrast, it did moderate the effect of training on healthy food consumption. As expected, highly impulsive participants who were trained to avoid unhealthy and approach healthy food consumed a greater proportion of healthy snacks than highly impulsive participants who were trained to approach unhealthy and avoid healthy food or those in the control (i.e., sham-training) condition. Together, these findings are consistent with the idea that impulsivity affects behaviour, rather than automatic processing. The result also suggests that interventions that target approach bias for food may only be effective at changing subsequent consumption for highly impulsive individuals. Thus, future studies should examine the effect of training on consumption for specific sub-groups of individuals.

Theoretically, our findings support the underlying assumptions of contemporary dual process models of behaviour (Strack & Deutsch, 2004). In particular, our findings support the suggestion that eating behaviour is determined by both automatic and self-regulatory control processes. Specifically, the finding that the training was effective at encouraging a greater proportion of healthy food consumption, but only among the highly impulsive individuals, is

consistent with the proposition that eating behaviour is more likely to be guided by automatic processes when self-regulatory control of impulsive responses is poor (Hofmann et al., 2008).

The current results also have practical implications as they indicate that increasing approach bias for healthy food encourages a greater proportion of healthy food consumption, albeit only for highly impulsive individuals. This suggests that an approach bias modification protocol that trains the simultaneous approach of healthy food and avoidance of unhealthy food would be most beneficial for individuals high on impulsivity, for example, overweight and obese people, who tend to be more impulsive than normal-weight people (Nederkoorn et al., 2006; Nederkoorn et al., 2006; Kakoschke, Kemps, & Tiggemann, 2015b). This is important given that in modern Western society, overweight and obesity rates are increasing, which is partly influenced by consuming too much food high in fat, salt and sugar. Therefore, approach bias modification interventions may be one way to effectively encourage highly impulsive people, such as overweight and obese individuals, to eat more healthy food and less unhealthy food.

In conclusion, the present study demonstrated that approach bias modification can be used in the eating domain to modify approach bias for both healthy and unhealthy food and to encourage healthier eating behaviour. Additionally, our findings support the hypothesis that trait impulsivity moderates the effectiveness of approach bias re-training on consumption of healthy food. Specifically, highly impulsive women benefited the most from receiving the avoid unhealthy and approach healthy food training as they subsequently consumed a greater proportion of healthy snack food. These findings suggest that future research needs to take into account the role of individual differences in impulsivity and could usefully examine the benefits of approach bias modification protocols for increasing healthy food consumption among individuals with problematic eating behaviours driven by impulsivity.

References

- Becker, D., Jostman, N. B., Wiers, R. W., & Holland, R.W. (2015). Approach avoidance training in the eating domain: testing the effectiveness across three single session studies. *Appetite*, 85, 58-65.
- Brignell, C., Griffiths, T., Bradley, B. P., & Mogg, K. (2009). Attentional and approach biases for pictorial food cues. Influence of external eating. *Appetite*, *52*, 299-306.
- Brockmeyer, T., Hahn, C., Reetz, C., Schmidt, U., & Friederich, H.-C. (2015). Approach
 Bias Modification in Food Craving A Proof-of-Concept Study: Approach Bias
 Modification in Food Craving. *European Eating Disorders Review*, 23, 352–360.
- Burton, P., Smit, H. J., & Lightowler, H. J. (2007). The influence of restrained and external eating patterns on overeating. *Appetite*, *49*, 191–197.
- Cohen, J. (1992). A power primer. Psychological bulletin, 112, 155-159.
- Cohen, D. (2008). Obesity and the built environment. Changes in environmental cues cause energy imbalances. *International Journal of Obesity*, *32*, 137–142.
- Cousijn, J., Goudriaan, A. E., & Wiers, R. W. (2011). Reaching out towards cannabis: Approach-bias in heavy cannabis users predicts changes in cannabis use. *Addiction*, 106, 1667-1674.
- Dickson, H., Kavanagh, D. J., & MacLeod, C. (2016). The pulling power of chocolate: Effects of approach–avoidance training on approach bias and consumption. *Appetite*, *99*, 46-51.
- Fishbach, A., & Shah, J. Y. (2006). Self-control in action: Implicit dispositions toward goals and away from temptations. *Journal of Personality and Social Psychology*, 90, 820-832.

- Friese, M., Hofmann, W., & Wanke, M. (2008). When impulses take over: Moderated predictive validity of explicit and implicit attitude measures in predicting food choice and consumption behaviour. *British Journal of Social Psychology*, 47, 397-419
- Guerrieri, R., Nederkoorn, C., & Jansen, A. (2007). How impulsiveness and variety influence food intake in a sample of healthy women. *Appetite*, *48*, 119-122.
- Havermans, R. C. (2013). Pavlovian craving and overeating: a conditioned incentive model. *Current Obesity Reports*, 2, 165-170.
- Havermans, R. C., Giesen, J. C., Houben, K., & Jansen, A. (2011). Weight, gender, and snack appeal. *Eating Behaviors*, *12*, 126-130.
- Hill, J.O., & Peters, J.C. (1998). Environmental contributions to the obesity epidemic. *Science*, 280(5368), 1371-1374.
- Hofmann, W., & Friese, M. (2008). Impulses got the better of me: Alcohol moderates the influence of implicit attitudes toward food cues on eating behavior. *Journal of Abnormal Psychology*, 117, 420.
- Hofmann, W., Friese, M., & Wiers, R. W. (2008). Impulsive versus reflective influences on health behavior: A theoretical framework and empirical review. *Health Psychology Review*, 2(2), 111-137.
- Hofmann, W., Gschwendner, T., Wiers, R., Friese, M., & Schmitt, M. (2008). Working memory capacity and self-regulation. Towards an individual differences perspective on behavior determination by automatic versus controlled processes. *Journal of Personality and Social Psychology*, 95, 962–977.
- Kakoschke, N., Kemps, E., & Tiggemann, M. (2014). Attentional bias modification encourages healthy eating. *Eating Behaviors*, *15*, 120-124.

- Kakoschke, N., Kemps, E., & Tiggemann, M. (2015a). Combined effects of cognitive bias for food cues and poor inhibitory control on unhealthy food intake. *Appetite*, 87, 358-364.
- Kakoschke, N., Kemps, E., & Tiggemann, M. (2015b). External eating mediates the relationship between impulsivity and unhealthy food intake. *Physiology & Behavior*, 147, 117-121.
- Kakoschke, N., Kemps, E., & Tiggemann, M. (2017). Approach bias modification training and consumption: A review of the literature. *Addictive Behaviors*, *64*, 21-28.
- Kemps, E., & Tiggemann, M. (2015). Approach bias for food cues in obese individuals. *Psychology & Health, 30,* 370-380.
- Kemps, E., Tiggemann, M., & Hollitt, S. (2014). Biased attentional processing of food cues and modification in obese individuals. *Health Psychology*, 33, 1391–1401.
- Kemps, E., Tiggemann, M., Martin, R., & Elliott, M. (2013). Implicit approach–avoidance associations for craved food cues. *Journal of Experimental Psychology: Applied*, 19, 30-38.
- Maas, J., Keijsers, G. P., Rinck, M., Tanis, J., & Becker, E. S. (2015). Does a dieting goal affect automatic cognitive processes and their trainability? *Cognitive Therapy and Research*, 39, 378-389.
- Marteau, T. M., Hollands, G. J., & Fletcher, P. C. (2012). Changing human behavior to prevent disease: The importance of targeting automatic processes. Science, 337 1492-1495.
- Masson. M. E., & Loftus, G. R. (2003). Using confidence intervals for graphically based data interpretation. *Canadian Journal of Experimental Psychology*, *57*, 203-220.

- Must, A., Spadano, J., Coakley, E. H., Field, A. E., Colditz, G., & Dietz, W. H. (1999). The disease burden associated with overweight and obesity. *Journal of the American Medical Association*, 282, 1523–1529.
- Nederkoorn, C., Braet, C., Van Eijs, Y., Tanghe, A., & Jansen, A. (2006). Why obese children cannot resist food: The role of impulsivity. *Eating Behaviors*, *7*, 315-322.
- Nederkoorn, C., Houben, K., Hofmann, W., Roefs, A., & Jansen, A. (2010). Control yourself or just eat what you like? Weight gain over a year is predicted by an interactive effect of response inhibition and implicit preference for snack foods. *Health Psychology*, 29, 389–393.
- Neumann, R., & Strack, F. (2000). Approach and avoidance: The influence of proprioceptive and exteroceptive cues on encoding of affective information. *Journal of Personality and Social Psychology*, *79*, 39–48.
- Patton, J.H. & Stanford, M.S. (1995). Factor structure of the Barratt impulsiveness scale. Journal of Clinical Psychology, 51, 768–774.
- Schumacher, S. E., Kemps, E., & Tiggemann, M. (2016). Bias modification training can alter approach bias and chocolate consumption. *Appetite*, *96*, 219-224.
- Seibt, B., Hafner, M., & Deutsch, R. (2007). Prepared to eat: How immediate affective and motivational responses to food cues are influenced by food deprivation. *European Journal of Social Psychology*, 37, 359-379.
- Stanford, M. S., Mathias, C. W., Dougherty, D. M., Lake, S. L., Anderson, N. E., & Patton, J.
 H. (2009). Fifty years of the Barratt Impulsiveness Scale: An update and review. *Personality and Individual Differences*, 47, 385-395.
- Strack, F., & Deutsch, R. (2004). Reflective and impulsive determinants of social behavior. Personality and Social Psychology Review, 8, 220-247.

- Tabachnick, B. G., & Fidell, L. S. (2007). Using multivariate statistics (5th ed.). Boston,MA: Pearson/Allyn and Bacon.
- Thush, C., Wiers, R. W., Ames, S. L., Grenard, J. L., Sussman, S., & Stacy, A. W. (2008).
 Interactions between implicit and explicit cognition and working memory capacity in the prediction of alcohol use in at-risk adolescents. *Drug and Alcohol Dependence*, *94*, 116-124.
- Veenstra, E. M., & de Jong, P. J. (2010). Restrained eaters show enhanced automatic approach tendencies towards food. *Appetite*, *55*, 30-36.
- Wiers, R. W., Gladwin, T. E., Hofmann, W., Salemink, E., & Ridderinkhof, K. R. (2013). Cognitive bias modification and cognitive control training in addiction and related psychopathology mechanisms, clinical perspectives, and ways forward. *Clinical Psychological Science*, 1, 192-212.
- Wiers, C. E., Kühn, S., Javadi, A. H., Korucuoglu, O., Wiers, R. W., Walter, H., ... & Bermpohl, F. (2013). Automatic approach bias towards smoking cues is present in smokers but not in ex-smokers. *Psychopharmacology*, 229, 187-197.
- Wiers, R. W., Rinck, M., Dictus, M., & Van den Wildenberg, E. (2009). Relatively strong automatic appetitive action-tendencies in male carriers of the OPRM1 G-allele. *Genes, Brain and Behavior*, 8, 101-106.
- Wiers, R. W., Rinck, M., Kordts, R., Houben, K., & Strack, F. (2010). Retraining automatic action-tendencies to approach alcohol in hazardous drinkers. *Addiction*, 105, 279-287.
- World Health Organization [WHO]. (2014). Obesity and overweight. Retrieved October 1, 2015 from http://www.who.int/mediacentre/factsheets/fs311/en/.

CHAPTER 8: GENERAL DISCUSSION

Chapter Overview

As outlined in Chapter 1, the thesis had two primary aims. The first was to investigate how automatic processing interacts with self-regulatory control or trait eating style in order to obtain a better understanding of these constructs in the context of unhealthy eating behaviour. The second aim was to assess whether intervention strategies aimed at modifying automatic and/or controlled processing are effective for discouraging unhealthy eating behaviour, and also for encouraging healthier eating. Five empirical studies and one literature review were conducted, each with their own specific aims but drawn from the overarching aims of this thesis. This final chapter aims to provide an integrated discussion of the main findings. First, a brief summary of the findings is provided. Next, theoretical and practical implications are discussed. Finally, methodological issues and directions for future research are considered.

Summary of Findings

Studies 1, 2 and 3 addressed the first aim, namely to develop an understanding of unhealthy eating behaviour by investigating possible underlying mechanisms, in particular, automatic processing in conjunction with self-regulatory control or trait eating style. Study 1 aimed to address one of the key theoretical predictions of recent dual-process models by empirically investigating the combined effects of automatic (attentional and approach bias) and controlled (inhibitory control) processing on consumption of unhealthy food. The study demonstrated attentional and approach biases for unhealthy food cues, but neither predicted consumption. Importantly, approach bias, but not attentional bias, was found to interact with inhibitory control in predicting increased consumption; participants who showed a stronger approach bias for food combined with low inhibitory control consumed the most snack food.

Study 2 turned to address the affective component of automatic processing by testing implicit evaluation of unhealthy food on unhealthy eating behaviour. The findings showed
that there was an interaction between implicit evaluation of unhealthy food and emotional eating style. Specifically, a positive implicit food evaluation (increased liking) predicted increased choice for unhealthy snack food in participants with lower emotional eating.

Study 3 then investigated the combined effect of approach bias for food and eating style on unhealthy food consumption in normal weight and overweight individuals using a pooled sample from Study 1 and the control groups of Studies 4 and 5. Among overweight participants, an external and emotional eating style individually moderated the relationship between approach bias for unhealthy food and subsequent food consumption, such that approach bias was positively related to consumption in high external or emotional eaters, but negatively related in low external or emotional eaters. These interactions were not observed among normal weight participants.

These studies, as a set, contribute to a theoretical understanding of how automatic and controlled processing interact in predicting unhealthy eating behaviour (Study 1), and in what way eating styles moderate the influence of both the affective (Study 2) and cognitive (Study 3) components of automatic processing to determine such behaviour. In general, the findings support the role of automatic processing of unhealthy food cues and help to determine when, and for whom, automatic processing can influence unhealthy eating behaviour.

Studies 4 and 5 addressed the second aim of the thesis, which was to examine whether interventions designed to modify automatic processing and/or self-regulatory control can reduce unhealthy eating behaviour and/or promote healthier eating. These were preceded by a literature review on the effectiveness of approach bias modification as an intervention for reducing the consumption of appetitive substances in general (i.e., alcohol, cigarettes, and unhealthy food). The review concluded that approach bias re-training generally had positive effects on behaviour, including reduced consumption in the laboratory, lower relapse rates, and improvements in self-reported measures (e.g., craving). Importantly, all reviewed studies

(with one exception) that reported a favourable outcome for consumption also showed a successful reduction of approach bias for appetitive cues. Effects were most consistent in the alcohol domain and more varied in the eating domain. In general, the review concluded that it is important for future research to determine how approach bias re-training can be improved.

Study 4 experimentally examined the effectiveness of approach bias re-training for unhealthy food cues and/or strengthening inhibitory control for reducing unhealthy eating behaviour. Although the combined re-training of approach bias along with inhibitory control was more effective than either training task alone for reducing implicit evaluation (liking) of unhealthy food, no significant training effects were found for unhealthy food consumption. Nevertheless, approach bias re-training was effective for reducing unhealthy snack food choice. Study 5 aimed to extend the use of approach bias re-training to the domain of healthy eating by modifying approach biases for both unhealthy and healthy food simultaneously, and examining effects on subsequent food consumption. It was found that the effect of training on consumption was moderated by trait impulsivity, such that only highly impulsive participants consumed a greater proportion of healthy snack food following training.

As a set, these latter studies show that re-training automatic approach biases may be an effective intervention for reducing unhealthy eating behaviour (Study 4) and encouraging healthier eating (Study 5). However, approach bias modification for food may be particularly useful for some individuals, such as those with poor trait self-regulatory control (Study 5).

Theoretical Implications

The present thesis contributes to the relatively small amount of research that has been conducted on automatic processing in the context of unhealthy eating behaviour. The studies focused on examining approach bias for appetitive food cues. In particular, the current studies show that both normal weight (Study 1) and overweight individuals (Study 3) are faster to approach than avoid unhealthy food cues. Thus, appetitive food cues do appear to elicit fast,

implicit responses by way of approaching (moving toward) such cues, which suggests that automatic approach tendencies may be a useful construct to examine in regard to unhealthy eating behaviour. Automatically approaching unhealthy food may be particularly problematic in the current environment, where appetitive, high-calorie food and related cues are abundant due to the ready availability of fast food and the prevalence of advertisements for such food.

Although approach bias toward unhealthy food cues alone was not associated with increased consumption, Studies 1, 2 and 3 also investigated when, and for whom, approach bias is likely to influence unhealthy eating behaviour. The findings from Study 1 provide empirical support for the suggestion that approach bias for unhealthy food cues only influences eating behaviour in combination with low inhibitory control over responses to such cues. A similar finding was demonstrated in one recently published study that found an interaction between an aspect of automatic processing (positive implicit evaluation) and self-regulatory control in a related domain, namely soft drink consumption (Eschenbeck, Heim-Dreger, Steinhilber, & Kohlmann, 2016). Taken together, these findings show that stronger automatic processing and weaker controlled processing are implicated in the consumption of appetitive substances. The results offer strong support for the main premise of dual-process models, which posit that our behaviour is guided by an interplay between these two different types of information processing (Strack & Deutsch, 2004).

An important point highlighted by the findings from Studies 2 and 3 is that automatic processing can also be influenced by other factors, such as trait eating style. Study 2 showed that the affective aspect of automatic processing (implicit evaluation of unhealthy food) was positively related to choosing an unhealthy snack food in low emotional eaters. While this might appear counterintuitive, it is not entirely surprising as implicit evaluation (increased liking) reflects an association between food and *positive* affect, while the emotional eating measure (DEBQ) assesses a tendency to eat in response to *negative* affect. Contrasting results

were found in Study 3, which examined a different aspect of automatic processing, namely approach bias. For normal weight individuals, no combined effects were found. However, for overweight individuals, approach bias for unhealthy food cues was related to consumption of unhealthy snack food in high emotional eaters. The finding that the direction of the effect of an emotional eating style on behaviour varied across the two studies suggests that there are differences between the affective (as measured by implicit evaluation) and cognitive (as measured by approach bias) components of automatic processing of unhealthy food cues.

The results of Study 3 also showed that another eating style was important, namely an external eating style. Among overweight individuals, approach bias for unhealthy food cues interacted with external eating in predicting unhealthy snack food consumption. Although approach bias and external eating style are two distinct variables, it has been suggested that automatic approach bias and individual differences in the tendency to eat in response to external food-related cues might both involve aspects of the same underlying construct, namely a heightened sensitivity to reward (Brignell, Griffiths, Bradley, & Mogg, 2009). Future research could investigate whether approach bias for food and external eating are a function of sensitivity to reward in general to clarify the constructs underlying automatic processes and eating style that can lead to overeating and weight gain.

The above studies have demonstrated that aspects of automatic processing together with self-regulatory control or eating style interact in predicting unhealthy eating behaviour. However, it is important to remember that the scope of the studies presented in the current thesis was limited to investigating particular components of automatic processing and selfregulatory control as well as specific types of trait eating styles. In regards to automatic processing, the studies only examined cognitive bias (attentional and approach) and implicit evaluation. Other potential aspects of automatic processing include a bias in memory. Future studies could investigate the role of memory bias for food cues in eating behaviour. In terms of controlled processing, inhibitory control and trait impulsivity were examined in the current studies. Future research could determine whether other aspects of controlled processing, such as working memory capacity and sensation seeking, also predict unhealthy eating behaviour in combination with automatic processing. Finally, the eating styles considered in the current studies, namely, restrained, emotional and external eating as measured by the DEBQ, are the most commonly researched. Future studies could assess other eating styles (e.g., disinhibited eating) as measured by other scales (e.g., the Three-Factor Eating Questionnaire; Stunkard & Messick, 1985).

The findings of Studies 1, 2, and 3 highlight the importance of examining the combination of automatic and controlled processing on unhealthy eating behaviour. However, as the studies were correlational in nature, the causal direction of relationships cannot be established. The laboratory measure of snack food consumption provides only one behavioural incidence of unhealthy eating behaviour. So it is possible, for example, that approach bias for food and low inhibitory control may be correlates or consequences, rather than predictors (as is posited in the current studies), of unhealthy eating behaviour. Only experimental studies that actually manipulate these constructs can determine the causal direction of the relationship between cognitive processes and unhealthy eating behaviour. Indeed, in Studies 4 and 5, approach bias and inhibitory control were experimentally manipulated to assess subsequent effects on food consumption.

Practical Implications

Studies 4 and 5 were geared toward addressing the second aim of the thesis, which was to determine whether interventions targeting the modification of automatic processing and/or self-regulatory control are effective for reducing unhealthy eating, but also promoting healthier eating. Given the negative consequences of unhealthy eating behaviour, including weight gain, it is important to not only be able to predict, but also to modify such behaviour. The literature review (Chapter 5) showed that the effect of approach bias modification was stronger in clinical samples (with higher levels of substance dependence) than in unselected samples. It seems likely that individuals in unselected samples are less motivated to change their appetitive consumption behaviour. The studies in the eating domain used mostly unselected samples and also showed greater variability than those in the alcohol domain in the effect of training on consumption. If the finding that training was most effective for alcohol dependent clinical samples generalises to the eating domain, then we might expect approach bias modification to be more effective for people with problematic eating, such as overweight or obese individuals.

Dual-process models suggest that one way to improve approach bias modification is to also modify self-regulatory control along with automatic processing. Study 4 tested this prediction experimentally and found that the effectiveness of the combined training depended upon the outcome measure that was considered. Specifically, combining the two training tasks induced a more negative implicit evaluation of unhealthy food, but did not influence food consumption. Approach-avoidance training reduced choice for unhealthy snack food. These findings can partly be explained by the suggestion that eating behaviour varies in terms of the degree of automatic and controlled processing involved. Specifically, single acts of eating behaviour may be more a function of automatic processing than continuous types of eating behaviour (Friese, Hofmann, & Wänke, 2008). In relation to the findings of Study 4, implicit evaluation of food is proposed to be an entirely automatic process. In contrast, choosing an unhealthy snack food represents a single act of eating behaviour under time pressure, and is likely to involve more automatic processing, but might also involve some degree of control. Finally, consumption of snack food during a taste test likely reflects a continuous measure of eating behaviour that involves both automatic and controlled processing (consistent with the results of Study 1).

Although the combined training did not have the predicted effect on consumption, it would be premature to conclude that the results of Study 4 do not support the key theoretical predictions of recent dual-process models. The major conclusion of the review on approach bias modification (Chapter 5) was that the mechanism underlying change in appetitive behaviour is a reduction in approach bias. Similarly, a recent review of attentional bias modification for anxiety emphasised the importance of distinguishing between the underlying process and the procedures aimed at modifying the process (MacLeod & Grafton, 2016). The main conclusion was that if the process has not been successfully modified, then a change in subsequent behaviour is not to be expected. The same principle applies to other types of cognitive modification tasks, such as inhibitory control training. Importantly, the go/no-go task that was used in Study 4 did not appear to effectively modify inhibitory control, as evidenced by the lack of difference in commission errors for food between conditions, and so no effect of the combined training on consumption could really be expected.

Study 5 aimed to address one of the suggestions made in the review on approach bias modification, namely to examine the possibility of not only training avoidance of unhealthy food, but also training approach toward healthy food. The study showed that simultaneously reducing approach bias for unhealthy food cues and increasing approach bias for healthy food resulted in a greater proportion of healthy food consumption, but only for individuals with higher trait impulsivity. Importantly, the finding that impulsivity moderates the effect of approach bias modification on the proportion of healthy food to improve eating behaviour might be most beneficial for individuals with poor trait self-regulatory control. Given the variable success of approach bias modification on consumption in the eating domain, as illustrated in the literature review, future research should examine whether such interventions are more effective when targeted at individuals with higher trait impulsivity.

In addition, Study 5 found that the most effective training protocol for modifying approach bias differed depending on the type of food stimuli used (healthy or unhealthy). Specifically, the effect of training on approach bias for unhealthy food was due to the 'avoid unhealthy' component, rather than the 'approach healthy' component. In contrast, the effect of training on approach bias for healthy food was due to a combination of both. Importantly, the finding that it is easier to train avoidance of unhealthy food than approach toward healthy food suggests that the 'avoid unhealthy' component may be the critical aspect of the training.

As a set, Studies 4 and 5 provide additional insight into the somewhat inconsistent findings for empirical studies using interventions aimed at modifying automatic processing underlying unhealthy consumption. Here approach bias modification was effective for retraining approach bias, but this did not translate to an effect on consumption across the whole sample (Study 4) or did so, but only for a particular sub-sample of the participants (Study 5). These findings are in contrast to the conclusion of the review that avoidance-training was effective at reducing consumption when approach bias for appetitive cues was successfully reduced. However, the findings are consistent with the logic of dual-process models (Strack & Deustch, 2004), which propose that approach bias should only translate to an effect on consumption behaviour when inhibitory control is low. In fact, the participants in the current intervention studies had relatively good inhibitory control. Study 5 showed that training had more positive effects on consumption in highly impulsive individuals, which suggests that approach bias re-training is more likely to be beneficial for individuals with poor trait selfregulatory control. Further investigation is warranted to determine whether individuals with deficits in other aspects of self-regulatory control, such as sensation seeking, are more likely to benefit from approach bias modification in the eating domain.

The studies presented here contribute to emerging evidence for the role of biased automatic processing in unhealthy eating behaviour. In particular, investigating the interplay

180

between approach bias for food with self-regulatory control and trait eating style identified some of the boundary conditions and different types of individuals for which approach bias is likely to predict unhealthy eating behaviour. Importantly, the findings point toward promising interventions for reducing unhealthy food consumption.

Methodological Issues and Future Directions

The series of studies were all conducted in the laboratory using unselected samples of undergraduate students. While these studies provided a better understanding of the processes that contribute to unhealthy eating behaviour in mostly normal weight people, this limits the ability to generalise the findings to clinical samples, including individuals with problematic eating behaviour. While we did not specifically aim to recruit overweight and obese people, participants from three of these studies were pooled so that normal weight and overweight individuals could be compared in Study 3. Importantly, there were differences between the two weight groups in the combination of approach bias for food and trait eating styles on snack food consumption. Future studies could analyse sub-groups in unselected samples to determine whether the findings might differ depending on the weight status of participants.

Future research should aim to generalise the current findings to a clinical sample, such as overweight or obese individuals, given that they have a stronger approach bias for unhealthy food and lower inhibitory control, are more impulsive, and have higher levels of trait external and emotional eating. Elucidating the predictors of unhealthy eating behaviour in overweight and obese people is crucial given the importance of developing interventions to reduce overeating in this population. Another reason for using an overweight or obese sample is that these individuals might be more likely to have motivation or a desire to change their eating behaviour. The literature review on approach bias modification suggests that a change in consumption behaviour occurs more readily in clinical samples, particularly in the alcohol domain. It is likely that approach bias re-training will also be more effective for modifying behaviour in clinical samples in the eating domain. Future studies could recruit individuals who would be motivated to eat more healthily and/or lose weight due to health-related goals.

The studies (4 and 5) in the current thesis that modified automatic processing and/or self-regulatory control used a single-training session, which may not be sufficiently intensive for an immediate effect on consumption. It is likely that more training is necessary to change a complex behaviour such as unhealthy eating that has presumably developed over time. The studies in the alcohol domain that did find positive effects of training on behaviour, such as reduced relapse rates one year later, used multiple training sessions (e.g., Eberl et al., 2013; Wiers et al., 2011). Therefore, it is likely the case that utilising multiple sessions of approach bias re-training would also be more effective for modifying behaviour in the eating domain. Future studies should aim to determine how many training sessions are required to establish an immediate effect on consumption, but research could also usefully explore the longevity of these training effects following multiple sessions conducted over a longer period of time.

Another important issue to consider is the control condition chosen for comparison in training protocols. While some studies compare an 'avoid unhealthy' training condition with an 'approach unhealthy' training condition (an extreme comparison as these two conditions manipulate approach bias in opposite directions), others compare 'avoid unhealthy' training with a 'sham-training' control condition (a less extreme comparison). The latter comparison was used in Study 4, while Study 5 was the first study to compare all three conditions ('avoid unhealthy', 'approach unhealthy', 'sham-training'). Study 5 found that an 'avoid unhealthy' condition was more likely to show a positive effect of training in comparison to an 'approach unhealthy' condition affected the evaluation of training effects. This finding is consistent with one recent study on inhibitory control training demonstrating that a 'no-go unhealthy' training condition was more likely to show a positive effect of training in comparison to 'go

unhealthy' training but not a control condition in which 'go' and 'no-go' responses were not manipulated (Adams, Lawrence, Verbruggen & Chambers, 2017). Taken together, the results suggest that studies using a more extreme control comparison condition might be more likely to demonstrate an effect of the intervention. Indeed, a recent review on a number of different cognitive modification tasks for food suggested that the type of control condition used might explain inconsistent findings in the eating domain (Stice, Lawrence, Kemps & Veling, 2016).

Another methodological issue relates to the timing of the measurement of approach bias in the training studies. Specifically, in Study 5 approach biases for healthy and unhealthy food cues were measured using an assessment version of the Approach-Avoidance task after the training phase. It is possible that this measurement might weaken the effect of training on subsequent behavioural measures, including the taste test. Study 4 addressed this issue by not measuring the bias after training, but this has the problem that approach bias is not measured. Other studies have assessed approach bias after the behavioural measure (Becker, Jostmann, Wiers, & Holland, 2015). However, it is possible that the behavioural measure may influence approach bias. Some other recent studies have included a booster session of training after the measurement of approach bias to reinforce learning of the trained contingencies and to ensure that the effectiveness of training was not weakened by the measurement (Becker et al., 2016; Lindgren et al., 2015; Mass, Jeijsers, Rinck, Tanis, & Becker, 2015), although these studies have produced mixed effects. Future intervention studies could aim to explicitly examine the timing of approach bias measurement.

A more general methodological issue is that all of the empirical studies in the current thesis (correlational and experimental) were conducted in a laboratory setting. Research done in the laboratory does enable control over several variables that are likely to influence eating behaviour. For example, food consumption was assessed with a standarised taste test across all of the studies. The snacks were presented in equal portion sizes and the participants were given the same instructions, questionnaires, and time to complete the taste test. Additionally, using an objective measure of eating behaviour, rather than relying on self-report, is likely to minimise several issues such as demand effects, social desirability, and inaccurate reporting of eating behaviour. However, a number of limitations can arise from conducting research in a standardised situation. For example, the taste test may not provide an accurate portrayal of how people usually eat. Indeed, research suggests that awareness of eating behaviour being monitored in a laboratory setting may reduce the amount of food that people eat (Robinson, Kersbergen, Brunstrom, & Field, 2014). In addition, food consumption in the current studies was assessed at only one specific time point. Measuring eating behaviour over different time points would allow for a more reliable representation of eating behaviour.

Despite the above limitations, the two experiments in the current thesis were able to manipulate automatic and controlled processing so that casual effects on eating behaviour could be established. Future research could usefully aim to determine whether approach bias modification has real-world benefits among individuals for whom unhealthy eating behaviour is a problem, such as overweight or obese people. In addition, future studies should examine the implementation of approach bias re-training tasks outside the laboratory setting through the use of novel delivery methods. Cognitive training tasks may be particularly suitable for development as online or mobile-based interventions given that they are computerised tasks with relatively simple instructions for participants (van Deursen, Salemink, Smit, Kramer & Wiers, 2013; Wiers, Gladwin, Hofmann, Salemink, & Ridderinkhof, 2013). Future research could endeavour to deliver approach bias re-training through an application designed for use with a smartphone so that training can be completed as it is required in real-world situations.

Conclusion

The current thesis addressed an important, but often overlooked, contributor to unhealthy eating behaviour, namely, automatic approach bias for unhealthy food cues. In particular, the empirical studies investigated the role of automatic processing in conjunction with self-regulatory control and trait eating style to foster a better understanding of unhealthy eating behaviour. The work presented here illustrates the complexity of the inter-relationships between these constructs. Theoretically, the findings support recent dual-process models by demonstrating that unhealthy eating behaviour is determined by a combination of automatic and controlled processing. In a practical sense, the findings suggest that interventions aimed at modifying approach bias for food cues may be effective for discouraging unhealthy eating and promoting healthier eating behaviour. Future research should usefully develop approach bias modification as a technique for reducing unhealthy eating behaviour in clinical samples.

References

- Adams, R. C., Lawrence, N. S., Verbruggen, F., & Chambers, C. D. (2017). Training response inhibition to reduce food consumption: Mechanisms, stimulus specificity and appropriate training protocols. *Appetite*, 109, 11-23.
- Becker, E. S., Ferentzi, H., Ferrari, G., Möbius, M., Brugman, S., Custers, J., ... & Rinck, M.
 (2016). Always approach the bright side of life: A general positivity training reduces stress reactions in vulnerable individuals. *Cognitive Therapy and Research*, 40, 57-71.
- Becker, D., Jostmann, N., Wiers, R., & Holland, R. (2015). Approach avoidance training in the eating domain: Testing the effectiveness across three single session studies. *Appetite*, 85, 58–65.
- Brignell, C., Griffiths, T., Bradley, B. P., & Mogg, K. (2009). Attentional and approach biases for pictorial food cues. Influence of external eating. *Appetite*, *52*, 299-306.
- Eberl, C., Wiers, R., Pawelczack, S., Rinck, M., Becker, E., & Lindenmeyer, J. (2013).
 Approach bias modification in alcohol dependence: Do clinical effects replicate and for whom does it work best? *Developmental Cognitive Neuroscience*, *4*, 38–51.
- Eberl, C., Wiers, R. W., Pawelczack, S., Rinck, M., Becker, E. S., & Lindenmeyer, J. (2014). Implementation of approach bias re-training in alcoholism - How many sessions are needed? *Alcoholism: Clinical and Experimental Research*, 38, 587-594.
- Eschenbeck, H., Heim-Dreger, U., Steinhilber, A., & Kohlmann, C. W. (2016). Selfregulation of healthy nutrition: Automatic and controlled processes. *BMC Psychology*, 4(1), 1-8.
- Friese, M., Hofmann, W., & Wänke, M. (2008). When impulses take over: Moderated predictive validity of explicit and implicit attitude measures in predicting food choice and consumption behaviour. *British Journal of Social Psychology*, 47, 397-419.

- Lindgren, K. P., Wiers, R. W., Teachman, B. A., Gasser, M. L., Westgate, E. C., Cousijn, J.,
 ... & Neighbors, C. (2015). Attempted training of alcohol approach and drinking
 identity associations in US undergraduate drinkers: Null results from two
 studies. *PloS one*, *10*(8), e0134642.
- Maas, J., Keijsers, G. P., Rinck, M., Tanis, J., & Becker, E. S. (2015). Does a dieting goal affect automatic cognitive processes and their trainability?. *Cognitive Therapy and Research*, *39*, 378-389.
- MacLeod, C., & Grafton, B. (2016). Anxiety-linked attentional bias and its modification:
 Illustrating the importance of distinguishing processes and procedures in experimental psychopathology research. *Behaviour Research and Therapy*, *86*, 68-86.
- Robinson, E., Kersbergen, I., Brunstrom, J.M., & Field, M. (2014). I'm watching you.Awareness that food consumption is being monitored is a demand characteristic in eating-behaviour experiments. *Appetite*, 83, 19-25
- Stice, E., Lawrence, N. S., Kemps, E., & Veling, H. (2016). Training motor responses to food: A novel treatment for obesity targeting implicit processes. *Clinical Psychology Review*, 49, 16-27.
- Strack, F., & Deutsch, R. (2004). Reflective and impulsive determinants of social behavior. Personality and Social Psychology Review, 8, 220–247.
- Stunkard, A. J., & Messick, S. (1985). The three-factor eating questionnaire to measure dietary restraint, disinhibition and hunger. *Journal of Psychosomatic Research*, 29, 71-83.
- van Deursen, D. S., Salemink, E., Smit, F., Kramer, J., & Wiers, R. W. (2013). Web-based cognitive bias modification for problem drinkers: protocol of a randomised controlled trial with a 2x2x2 factorial design. *BMC Public Health*, *13*, 1-11.

- Veling, H., Aarts, H., & Stroebe, W. (2013a). Using stop signals to reduce impulsive choices for palatable unhealthy foods. *British Journal of Health Psychology*, 18, 354–368.
- Wiers, R., Eberl, C., Rinck, M., Becker, E., & Lindenmeyer, J. (2011). Retraining automatic action tendencies changes alcoholic patients' approach bias for alcohol and improves treatment outcome. *Psychological Science*, 22, 490–497.
- Wiers, R. W., Gladwin, T. E., Hofmann, W., Salemink, E., & Ridderinkhof, K. R. (2013).
 Cognitive bias modification and cognitive control training in addiction and related psychopathology mechanisms, clinical perspectives, and ways forward. *Clinical Psychological Science*, *1*, 192-212.